Recorded

Book Page
3993 34

4:13 p.m.

April 22, 2005

STATE OF NORTH CAROLINA COUNTY OF BUNCOMBE

DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

GRANTOR: THE SETTINGS OF BLACK MOUNTAIN, LLC, A
GEORGIA LIMITED LIABILITY COMPANY

PREPARED BY AND RETURN TO:

SUSAN S. BARBOUR, ESQ. McGuire, Wood & Bissette, PA 48 Patton Avenue Asheville, NC 28801

TABLE OF CONTENTS

ARTICLE 1:	DEFINITIONS	2
1.1	"Act"	2 2 2 2 2 2
1.2	"Additional Property"	2
1.3	"Area of Common Responsibility"	2
1.4	"Articles of Incorporation" or "Articles"	2
1.5	"Association"	
1.6	"Board of Directors" or "Board"	2
1.7		2
1.8	"Builder/Owner" "By-Laws of the Association" or "By-Laws"	2 2 2 2
1.9	"Common Area"	2
1.10	"Common Expenses"	3 3
1.11	"Community	3
1.12	"Community-Wide Standard"	3 3
1.13	"Cost Sharing Agreement"	3
1.14	"County"	3
1.15	"Days"	3
1.16	"Declarant"	3
1.17	"Declaration"	3
1.18	"Design Guidelines"	3
1.19	"Design Review Board	. 3
1.20	"Development Period"	4
1.21	"Exclusive Common Area"	4
1.22	"Foreclosure"	4
1.23	"General Assessments"	4
1.24	"Governing Documents"	4
1.25	"Institutional Mortgage"	4
1.26	"Lot"	4
1.27	"Majority"	4
1.28	"Member"	4
1.29	"Modifications Committee"	4
1.30	"Mortgage"	
1.31	"Mortgagee"	5
1.32	"Owner"	5
1.33	"Parcel"	5
1.34	"Parcel Assessments"	5
1.35	"Person"	5
1.36	"Phase"	5
1.37	"Properties"	5
1.38	"Public Records"	5
1.39	"Residential Unit"	5
1.40	"Right of First Refusal"	6
1.41	"Rules and Regulations"	6
1.42	"Special Assessments"	6

1.43	"Specific Assessments"	6
1.44	"Subdivision Plats"	
1.45	"Supplemental Declaration"	6
ARTICLE 2:	PROPERTY RIGHTS	6
2.1	Development of the Properties	6
2.2	Development of Additional Property	
2.3	Easement of Enjoyment	7
2.4	Exclusive Common Area	8
2.5	No Partition	
2.6	Interest Subject to Plan of Community	9
2.7	Subdivision Plat	
2.8	Property Rights	
ARTICLE 3:	MEMBERSHIP AND VOTING RIGHTS	9
3.1	Membership	
3.2	Multiple Owners	10
3.3	Voting	
3.4	Exercise of Voting Rights	₁₁
3.5	Declarant Control	11
ARTICLE 4:	RIGHTS AND OBLIGATIONS OF THE ASSOCIATION	11
4.1	Function of Association	
4.2	Common Area	11
4.3	Services	
4.4	Personal Property and Real Property for Common Use	12
4.5	Power of Contract	12
4.6	Rules and Regulations	12
4.7	Enforcement	
4.8	Implied Rights	15
4.9	Indemnification	15
4.10	Security	15
4.11	Power to Assess	15
4.12	Board Right to Act	16
ARTICLE 5:	MAINTENANCE	16
5.1	Association's Responsibility	16
5.2	Owner's Responsibility	17
5.3	Standard of Performance	18
5.4	Cost Sharing Agreements	18
ARTICLE 6:	INSURANCE AND CASUALTY LOSSES	19
6.1	Insurance and Casualty or Liability Losses	19
6.2	Disbursement of Proceeds	21
6.3	Damage and Destruction	21
6.4	Renair and Reconstruction	22

6.5	Lot Owner's Responsibility	22
ARTICLE 7:	ANNEXATION AND WITHDRAWAL OF PROPERTY	23
7.1	Annexation Without Approval of Class "A" Membership	23
7.2	Annexation With Approval of Class "A" Membership	
7.3	No Restrictions	
7.4	A mendment	23 24
7.5	Amendment Withdrawal of Property	24
7.5	withdrawar or rioporty	
ARTICLE 8:	ASSESSMENTS	25
8.1	Creation of and Obligation for Assessments	25
8.2	Commencement of Assessments	 26
8.3	Computation of General Assessments	
8.4	Special Assessments	28
8.5	Specific Assessments	
8.6	Foreclosure of the Lien for Assessments	
8.7	Failure to Assess	 29
8.8	Exempt Property	_ _29
8.9	Surplus Funds	
	•	
ARTICLE 9:	ARCHITECTURAL STANDARDS	29
9.1	General	29
9.2	Design Review Board	29
9.3	Submissions to Design Review Board	31
9.4	The Modifications Committee	33
9.5	Commencement of Construction and Occupancy of Residential Units	34
9.6	Approval of Plans	35
9.7	Construction Criteria and Requirement of Compliance With Law	35
9.8	Land Use and Building Type	35
9.9	No Waiver of Future Approvals	38
9.10	Variance	38
9.11	Limitation of Liability	38
9.12	Enforcement	39
ARTICLE 10:	USE RESTRICTIONS	40
10.1	Use Restrictions	40
10.2	Occupants Bound	40
10.3	Leasing	
10.4	Residential Use	40
10.5	Vehicles	41
10.6	Use of Common Area	42
10.7	Animals and Pets	43
10.8	Nuisance	43
10.9	Storage of Materials, Garbage, Dumping, Etc.	44
10.10	Combustible Liquid	44
10.11	Guns	44

10.12	Subdivision of Lot	44
10.13	Drainage and Grading	45
10.14	Irrigation	45
10.15	Streams	45
10.16	Wetlands	45
10.17	Temporary Structures	45
10.18	Garbage Cans, Woodpiles, etc.	46
10.19	Solar Devices	46
10.20	Above-Ground Pools	46
10.21	Oil and Mining Operations	46
ARTICLE 11	: EASEMENTS	46
11.1		46
11.2	Owner's Right to Ingress, Egress, Use and Support	47
11.3	Easement for Additional Property	47
11.4	Easement for Association	48
11.5	Maintenance Easement	40
11.6	Alterations in Boundaries, Additions to Common Area	48
11.7	Easement of Encroachment	
11.8	Use of Common Area	10
11.9	Acknowledgment of Rights and Use	
11.10	Conveyance of Common Area	49
11.11	Construction and Sale Period	49
11.12	Easements for Utilities, etc.	49
	Easement for Law Enforcement and Fire Protection	
	Assignment of Declarant Rights	
	Easement for Walks, Trails, Signs and Perimeter Walls	
11.16	Easements for Lateral Support	51
11.17	Easement for Special Events	51
11.18	Easement for Park and Walking Trail Access	51
11.19	Liability for Use of Easements	51
11.20	Notice of Electric Transmission Line Easement	51
ARTICLE 12	: MORTGAGEE'S RIGHTS	52
12.1		52
12.2	Other Provisions for First Lien Holders	52
12.3	Amendments to Documents	53
12.4	Special FHLMC Provision	54
12.5	·	54
12.6	No Priority	54
12.7	Notice of Default	55
ARTICLE 13	: DECLARANT'S RIGHTS	55
13.1	Transfer or Assignment	55
13.2	Development and Sales	55
13.3	Improvements to Properties	55

13.4	Additional Covenants	56
13.5	Amendments	56
13.6	Additional Covenants and Easements by Declarant	56
13.7	Water System	56
13.8	Annexation by Town of Black Mountain	57
13.9	Reservation of Rights under the Act	57
ARTICLE 14	: RIGHT OF FIRST REFUSAL	57
14.1	Applicability	57
14.2	Right of First Refusal	5/
14.3	Death of an Owner; Gift	58
14.4	Transfers to Declarant	58
14.5	Transfers to Declarant No Further Documentation Required	58
ARTICLE 15	: GENERAL PROVISION	59
15.1	Duration	59
15.2	Amendment	
15.3	Severability	60
15.4	Dispute Resolution	60
15.5	Litigation	60
15.6	Non-Merger	61
15.7	Grants	
15.8	Cumulative Effect; Conflict	61
15.9	Use of Protected Words "The Settings of Black Mountain"	61
15.10	Compliance	61
15.11	Notice of Sale or Transfer of Title	62
15.12	Variances and Waiver of Restrictions	62
15.13	Merger and Subdivision of Lots	62
15.14	Reservation From Lot Conveyance	62
15.15	Incorporation by Reference	62
15.16	Interpretation	63
15.17	Captions	63
	Gender and Grammar	
ARTICLE 16	: CONDEMNATION	63

TABLE OF EXHIBITS

Exhibit	Subject Matter	Page First Appearing
A	Land Initially Submitted	i
В	Consent of Mortgagee	ii
C	By-Laws of The Settings of Black Mountain Association, In	nc. iii

DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

THIS DECLARATION OF COVENANTS, CONDITIONS, AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN (hereinafter referred to as the "Declaration" as defined herein below) is made as of the date set forth on the signature page hereof by The Settings of Black Mountain, LLC, a Georgia limited liability company (hereinafter referred to as the "Declarant" as defined herein below).

WITNESSETH

Declarant is the owner of the real property described in **Exhibit A** attached hereto and incorporated herein by this reference. Declarant intends by this Declaration to impose upon the **Exhibit A** property mutually beneficial restrictions, easements, covenants and conditions under a general plan of improvement and development for the benefit of all owners of property within The Settings of Black Mountain. Declarant desires to provide a flexible and reasonable procedure for the overall development, administration, maintenance, preservation, use and enjoyment of such property as is now or may hereafter be submitted to this Declaration; and

Declarant has retained and reserved the option, right and privilege, but not the obligation, to annex to the Community and subject to this Declaration all or any portion of the Additional Property, as defined hereinbelow;

NOW THEREFORE, Declarant hereby declares that all of the property described in **Exhibit A**, and any Additional Property which may be added by subsequent amendment to this Declaration, shall be held, sold, and conveyed subject to the following easements, restrictions, covenants and conditions which shall run with the real property submitted to this Declaration and which shall be binding on all parties having any right, title or interest in the property described on **Exhibit "A"** and any portion of the Additional Property made subject hereto, their heirs, successors, successors in title, and assigns, and shall inure to the benefit of each owner thereof. By execution hereinbelow, Declarant further declares that this Declaration does not and is not intended to create a unit owners development pursuant to the North Carolina Unit Ownership Act, N.C.G.S.A. § 47A-1, et seq. or a condominium pursuant to the North Carolina Condominium Act N.C.G.S.A. § 47C-1-101, et seq. Pursuant to and subject to the terms and provisions of the North Carolina Planned Community Act as set forth in N.C.G.S. § 47F, et seq., (the "Act") Declarant hereby creates a planned community subdivision initially comprised of the property described in **Exhibit "A"**.

ARTICLE 1: DEFINITIONS

The terms in this Declaration and the attached exhibits shall generally be given their natural, commonly accepted definitions except as otherwise specified. Capitalized terms shall be defined as set forth below.

- 1.1 "Act" shall mean and refer to the North Carolina Planned Community Act as set forth and contained in Chapter 47F of the North Carolina General Statutes.
- 1.2 "Additional Property": Shall mean and refer to any property as may be adjacent to or contiguous with the **Exhibit "A"** Property. Property shall be deemed to be adjacent to or contiguous with the **Exhibit "A"** Property if it physically connects to such property, at any point, or if it is separated only by a road, public or private, or water course, including any river, creek or lake.
- 1.3 "Area of Common Responsibility": The Common Area, together with those areas, if any, for which the Association has or assumes responsibility pursuant to the terms of this Declaration, any Supplemental Declaration, or other applicable covenant, contract or agreement.
- 1.4 "<u>Articles of Incorporation</u>" or "<u>Articles</u>": Shall mean or refer to the Articles of Incorporation of the Association as amended from time to time.
- 1.5 "Association": Shall mean and refer to The Settings of Black Mountain Association, Inc., a North Carolina non-profit corporation, its successors and assigns.
- 1.6 "Board of Directors" or "Board": Shall mean and refer to the board of directors of the Association having its normal meaning under the North Carolina Non-Profit Corporation Act and Law.
- 1.7 "<u>Builder/Owner</u>": Shall mean and refer to the Owner of a Lot, which Owner is designated, in writing, as a Builder/Owner by Declarant who, as a primary vocation, is in the business of construction of Residential Units and who owns such Lot for the purpose of development of a Residential Unit thereon and the sale thereof to a third party.
- 1.8 "By-Laws of the Association" or "By-Laws": Shall mean and refer to the by-laws of the Association which govern the administration and operation of the Association, as the same may be amended from time to time.
- 1.9 "Common Area": Shall mean or refer to all real and personal property now and hereafter owned by the Association for the common use and enjoyment of the Owners or otherwise made available for the use and enjoyment of the Owners. The Common Area shall include, but not be limited to, the roads, streets, entrance ways, recreational areas, street lighting and signage located on property owned by the Association or made available for the use and enjoyment of the Owners. Pursuant to N.C.G.S. § 136-102.6, all future owners acknowledge that the road rights of ways as shown on the subdivision plats are private road rights of ways dedicated to the public use and may not be developed to State of North Carolina Department of Transportation specifications.

- 1.10 "<u>Common Expenses</u>": Shall mean and refer to the actual and estimated expenses of operating the Association and the Community, including any reasonable reserve, all as may be imposed hereunder and found to be necessary and appropriate by the Board pursuant to this Declaration, the By-Laws and the Articles of Incorporation.
- 1.11 "Community": Shall mean and refer to that certain real property described in **Exhibit "A"** attached hereto and such additions thereto as may be made by Declarant by any Supplemental Declaration or amendment hereto which annexes to this Declaration all or any portion of the Additional Property.
- 1.12 "<u>Community-Wide Standard</u>": Shall mean the standard of conduct, maintenance or other activity generally prevailing in the Community as determined by the Declarant for so long as the Development Period continues and thereafter as determined by the Board.
- 1.13 "Cost Sharing Agreement": Any agreement, contract or covenant between the Association and an owner or operator of other property adjacent to, in the vicinity of or within the Community for the allocation of expenses for amenities and/or services that benefit both the Association and the owner or operator of such property.
 - 1.14 "County": Buncombe County, North Carolina.
- 1.15 "<u>Days</u>": Calendar days; provided, however, if the time period by which any action required hereunder must be performed expires on a Saturday, Sunday or legal holiday, then such time period shall be automatically extended to the close of business on the next regular business day.
- 1.16 "Declarant": Shall mean The Settings of Black Mountain, LLC, the owner of the Properties submitted hereto, together with any successor in title who comes to stand in the relation to the Community as his predecessor. Notwithstanding the foregoing, the phrase "Owner" as referred to in this definition shall not include in its capacity as such any Mortgagee except for such Mortgagee who acquires said Declarant's entire interest with respect to the Properties and the Additional Property at the time of such acquisition pursuant to Foreclosure of a Mortgage encumbering said Declarant's interest in the Properties and the Additional Property and who then expressly assumes the position of Declarant.
- 1.17 "<u>Declaration</u>": Shall mean and refer to this Declaration of Covenants, Conditions and Restrictions and all amendments and Supplemental Declarations thereto filed for record in the Public Records of the County.
- 1.18 "<u>Design Guidelines</u>": The architectural and construction guidelines and application and review procedures applicable to all or any portion of the Community promulgated and administered pursuant to Article 9.
- 1.19 "<u>Design Review Board</u>": Shall mean and refer to that certain Board/Committee as empowered in accordance with Article 9 hereof.

- 1.20 "<u>Development Period</u>": The period of time during which the Declarant owns any property which is subject to this Declaration or has the unilateral right to subject Additional Property to this Declaration pursuant to Section 7.1. The Declarant may, but shall not be obligated to, unilaterally relinquish its rights under this Declaration and terminate the Development Period by recording a written instrument in the Public Records.
- 1.21 "Exclusive Common Area": A portion of the Common Area intended for exclusive use or primary benefit of one or more, but less than all Lots, as more particularly described in Article 2.
- 1.22 "<u>Foreclosure</u>": Shall mean and refer to, without limitation, the judicial or non-judicial foreclosure of a Mortgage or the conveyance of a secured party by a deed in lieu of such judicial or non-judicial foreclosure.
- 1.23 "General Assessments": Assessments levied on all Lots subject to assessment under Article 8 to fund Common Expenses for the general benefit of all Lots, as more particularly described in Article 8.
- 1.24 "Governing Documents": The Declaration, By-Laws, Articles of Incorporation, Supplemental Declarations, Preferred Builder Program Agreements, Design Guidelines, Rules and Regulations, Cost Sharing Agreements, and all additional covenants governing any portion of the Properties or any of the above, as each may be amended from time to time.
- 1.25 "<u>Institutional Mortgage</u>": Shall mean and refer to a Mortgage held by a bank, trust company, insurance company, or other recognized lending institution, or by an institutional or governmental purchaser of mortgage loans in the secondary market, such as Federal National Mortgage Association or Federal Home Loan Mortgage Corporation.
- 1.26 "Lot": Shall mean and refer to a platted portion of the Properties other than the Common Area, intended for independent use or ownership. Lots shall be shown on the plats of survey filed with this Declaration in the Public Records of the County, or amendments thereto, or may be further described in any amendment to this Declaration or Supplemental Declaration which may be made applicable to all or any portion of the Properties. The term "Lot" shall include within its meaning, but shall not be limited to, a numbered parcel identified on a plat of the Properties recorded in the Public Records which is intended for independent ownership and shall, for all purposes, include the Residential Unit as may be located thereon.
- 1.27 "Majority": Those votes, Owners, members, or other group, as the context may indicate, totaling more than fifty percent (50%) of the total eligible number.
- 1.28 "Member": Shall mean and refer to a person or entity entitled to membership in the Association as provided herein.
- 1.29 "Modifications Committee": Shall mean and refer to that certain Committee of the Association as empowered in accordance with Article 9 hereof.

- 1.30 "Mortgage": Shall mean and refer to a deed to secure debt, deed of trust, as well as a Mortgage, and a "First Mortgage" is a first priority deed to secure debt, deed of trust or mortgage.
- 1.31 "Mortgagee": Shall mean and refer to the grantor, holder or beneficiary of a Mortgage.
- 1.32 "Owner": Shall mean and refer to the record owner, whether one or more persons or entities, of any Lot which is part of the Properties but excluding in all cases any party holding an interest merely as security for the performance of an obligation. When the term Owner is used it shall include a Builder/Owner, unless the context otherwise so requires.
- 1.33 "Parcel": Shall mean and refer to designated subdivisions of Property subject to this Declaration and comprised of one or more Lots. In the absence of a specific designation of separate Parcel status, all property within a Phase shall be considered as part of the same Parcel; provided, however, the Declarant may designate so long as the Declarant owns a Lot by certification recorded in the Public Records that such Property shall constitute a separate Parcel or Parcels and provided, further, a Parcel may include more than one Phase if so designated by Declarant. A Parcel may be smaller or larger or coterminous with any and all Phases. A Parcel may be established, by way of example and not limitation, to include only interior Lots, waterfront Lots, or other Lots similarly situated one to the other, provided that this shall not be a limitation in the creation of a Parcel but rather an example thereof.
- 1.34 "<u>Parcel Assessments</u>": Shall mean and refer to Assessments for Common Expenses provided for herein or by any Supplemental Declaration or amendments hereto or thereto which are used for the purpose of promoting the recreation, health, safety, welfare, common benefit and enjoyment of the Owners within a specific Parcel, including but not limited to, the maintenance of Property within a given Parcel.
- 1.35 "Person": A natural person, a corporation, a partnership, a limited liability company, a fiduciary acting on behalf of another Person or any other legal entity.
- 1.36 "Phase": Shall mean and refer to the increments of Property subjected to this Declaration including the initial Phase described on **Exhibit "A"** hereto and subsequent Phases described in any amendments or Supplemental Declarations annexing all or any portion of the Properties hereto, each such described Property being a separate Phase.
- 1.37 "<u>Properties</u>": Shall mean and refer to the real property described on <u>Exhibit "A"</u> attached hereto and shall further refer to such Additional Property or part thereof when and if such is annexed by amendment or Supplemental Declaration to this Declaration. The Properties shall include the Common Area.
 - 1.38 "Public Records": The Office of the Register of Deeds of the County.
- 1.39 "Residential Unit": Shall mean and refer to any portion of the Properties intended for use and occupancy as a residence or accommodation by a single household and shall, unless

otherwise specified, include within its meaning by way of illustration and not limitation, single-family detached homes. For purpose of this Declaration, a Residential Unit shall come into existence when substantially complete.

- 1.40 "Rules and Regulations": Shall mean and refer to those Rules and Regulations as promulgated by the Board of Directors of the Association pursuant to this Declaration and By-Laws.
- 1.41 "<u>Right of First Refusal</u>": Shall mean and refer to those rights of repurchase reserved by the Declarant in accordance with Article 14.
 - 1.42 "Special Assessments": Assessments levied in accordance with Section 8.4.
- 1.43 "Specific Assessments": Assessments levied in accordance with Section 6.4 or Section 8.5.
- 1.44 "<u>Subdivision Plats</u>": Shall mean and refer to the Plats recorded by Declarant in the Public Records which depict the Lots within the Community or any Phase of the Properties.
- 1.45 "Supplemental Declaration": Shall mean and refer to a Declaration of Covenants, Conditions and Restrictions which, from and after date of recording of this Declaration, may be executed by Declarant or approved, in writing, by Declarant which, by its terms, provides for restrictions on the use of Property or respective Phase or Parcel in a specified manner, as set forth therein. By way of example and not limitation, a Supplemental Declaration may establish supplemental restrictive covenants, conditions or restrictions applicable to a Phase or Parcel which provide for restrictions regarding design guidelines for Residential Units or provide for supplemental restrictive covenants, conditions or restrictions as may be applicable to any respective portion of the Properties annexed and made subject to the terms of this Declaration.

ARTICLE 2: PROPERTY RIGHTS

Development of the Properties. Except as otherwise set forth in Section 10.4, or as permitted by a Supplemental Declaration as may be applicable to a respective Phase, all Lots within the Properties (i) shall be and are hereby restricted exclusively to single-family residential use (ii) shall be developed and built upon only for detached single family dwelling purposes and (iii) shall be subject to the standards and restrictions set forth in this Declaration including, specifically, Article 9 hereof and Design Guidelines adopted pursuant to this Declaration. Declarant shall have the right, but not the obligation, during the Development Period to make improvements and changes to all Common Area and to all Lots owned by Declarant, including, without limitation, (i) installation and maintenance of any improvements in and to the Common Area, (iii) installation and maintenance of any water, sewer, and other utility systems and facilities, and (iv) installation of refuse facilities.

- 2.2 <u>Development of Additional Property</u>. Declarant hereby reserves the option, right and privilege (but not the obligation), to be exercised in its sole discretion, to submit from time to time the Additional Property or a portion or portions thereof to the provisions of this Declaration and thereby to cause the Additional Property or a portion or portions thereof to become part of the Properties. This option, right or privilege may be exercised only by Declarant in accordance with the terms, conditions, and limitations set forth in Article 7 hereinbelow, which are the only terms, conditions and limitations on such option, right and privilege to add all or any portion of the Additional Property to the Properties.
- 2.3 Easement of Enjoyment. Every Owner, including each Builder/Owner, shall have a right and easement of enjoyment in and to the Common Area subject to any restrictions, limitations, or provisions contained in the Governing Documents. Such right and easement may be exercised by the Owner, including Builder/Owner, as the case may be, and the members of their respective family and their respective tenants, licensees and invitees, subject to such reasonable regulations and procedures as may be adopted by the Board. An Owner, including each Builder/Owner, may assign to any tenant thereof all rights of access to and use of the Common Area so that such tenant, and its respective licensees and invitees shall be entitled to access to and use and enjoyment of the Common Area on the same basis as the assignor and his family and guests. The aforementioned right and easement of enjoyment shall be appurtenant to and shall pass with the title to every Lot subject to the following easements, reservations, rights and provisions, which are expressly reserved hereby:
- (a) the right of the Association to suspend an Owner's voting rights, if any, and right to use the facilities as may be located on the Common Area for any period during which (i) any Association assessment against said Owner's property remains unpaid and (ii) for any infraction of the Rules and Regulations for the duration of the infraction and for an additional period thereafter not to exceed thirty (30) days;
- (b) the right of the Association to borrow money (i) for the purpose of improving the Common Area, or any portion thereof, (ii) for acquiring additional Common Area, (iii) for constructing, repairing, maintaining or improving any facilities located or to be located within the Common Area, or (iv) for providing the services authorized herein, and, subject to the provisions herein, to give as security for the payment of any such loan a Mortgage; provided, however, no portion of the Common Area may be conveyed or subjected to a security interest by the Association unless persons entitled to cast eighty percent (80%) of the votes of the Association agree in writing to such action; provided, further, that the lien and encumbrance of any such Mortgage given by the Association shall be subject and subordinate to any and all rights, interests, options, licenses, easements and privileges herein reserved or established for the benefit of Declarant, any Owner, or the holder of any Mortgage, irrespective of when such Mortgage was executed or given. Notwithstanding the above, or any other provision of this Declaration to the contrary, during the Development Period, the Association shall not, without the consent of Declarant, borrow money or pledge, mortgage or hypothecate all or any portion of the Common Area;
- (c) the right of the Association to grant and accept easements as provided herein and to dedicate or transfer fee simple title to all or any portion of the Common Area to the County or to any other public agency or authority, public service district, public or private utility, or other Person

provided that any such transfer of the fee simple title must be approved by at least eighty percent (80%) of the votes of the Association and by Declarant, during the Development Period; provided nothing herein shall require the County or any public agency or authority of the County to accept any such conveyance until the standards of such proposed grantee have been met and satisfied;

(d) the right of the Declarant, or the Association with the approval of Declarant, to dedicate, transfer, or grant permits, licenses, or easements in and to the Common Area for utilities, roads, and other purposes reasonably necessary or useful for the proper development, maintenance, or operation of the Properties, the Additional Property, and the Association;

2.4 Exclusive Common Area.

- (a) The Declarant may designate certain portions of the Common Area as Exclusive Common Area which thereby shall be reserved for the exclusive use or primary benefit of Owners and occupants of specified Lots. By way of illustration and not limitation, Exclusive Common Areas may include entry features, private streets, landscaped medians and cul-de-sacs, wells, and shared driveways. All costs associated with maintenance, repair, replacement, and insurance of an Exclusive Common Area shall be assessed as a Specific Assessment against the Owners of Lots to which the Exclusive Common Areas are assigned.
- (b) Exclusive Common Area shall be designated as such, and the exclusive use thereof shall be assigned, in the deed by which the Common Area is conveyed to the Association, this Declaration, a Supplemental Declaration and/or on the subdivision plat relating to such Common Area; provided however, any such assignment shall not preclude the Declarant from later assigning use of the same Exclusive Common Area to additional Lots during the Development Period. Thereafter, a portion of the Common Area may be assigned as Exclusive Common Area of a particular Lot and Exclusive Common Area may be reassigned upon approval of the Board and the Majority vote of the total Class "A" Members in the Association, including, if applicable, a Majority vote of the Class "A" Members to which the Exclusive Common Area is assigned, if previously assigned, and to which the Exclusive Common Area is to be assigned or reassigned.
- (c) The Association may, upon approval of a Majority vote of the Class "A" Members to which any Exclusive Common Area is assigned, permit Owners of other Lots to use all or a portion of such Exclusive Common Area upon payment of reasonable user fees, which fees shall be used to offset the Specific Assessments attributable to such Exclusive Common Area.
- 2.5 No Partition. Except as is permitted in this Declaration, there shall be no partition of the Common Area or any part thereof, nor shall any person acquiring any interest in any of the Properties or any part thereof seek any such partition, judicial or otherwise, unless the affected area has been removed from the provisions of this Declaration. This section shall not be construed to prohibit the Board of Directors from acquiring and disposing of tangible personal property nor from acquiring title to real property which may or may not be subject to this Declaration.

2.6 <u>Interest Subject to Plan of Community</u>. Every Owner of a Lot shall take title to such Lot and every Mortgagee and lienholder holding an interest therein shall take title or hold such security interest with respect thereto subject to this Declaration, as may be amended. Any provision of this Declaration to the contrary notwithstanding, the provisions set forth in this Article 2 may not be abrogated, modified, rescinded, supplemented, or amended in whole or in part without the prior written consent of Declarant during the Development Period.

2.7 Subdivision Plat.

- (a) Declarant reserves the right to record, modify, amend, revise and add to, at any time and from time to time, Subdivision Plat(s) setting forth such information as Declarant may deem necessary with regard to the Properties, including, without limitation, the locations and dimensions of the Lots, Common Area, Additional Property, roads, utility systems, drainage systems, utility easements, drainage easements, access easements, and set-back line restrictions.
- (b) Until the time a Lot within a respective Parcel or Phase is transferred by the Declarant to another (other than a Builder/Owner, an affiliate of Declarant, or a holder of a First Mortgage), no Owner of any Lot or Residential Unit shall have any rights whatsoever to the continuation of any covenants, or restrictions on such Parcel or Phase as contained herein or as may be imposed, expressly or impliedly, by recordation of any Subdivision Plat or as might otherwise be implied or expressed. In furtherance thereof, until the time a Lot within a respective Parcel or Phase is transferred by the Declarant as aforementioned, the Declarant may revoke or cancel any Subdivision Plat or other instrument which might be deemed, either expressly or impliedly, to impose any covenants, conditions, or restrictions or may take whatever steps it deems necessary or desirable to avoid the implication of such existing.
- 2.8 Property Rights. Each Lot shall for all purposes constitute real property which may be owned in fee simple and which, subject to the provisions of this Declaration or any Supplemental Declaration, may be conveyed, transferred, and encumbered the same as any other real property. Each Owner shall be entitled to the exclusive ownership and possession of his Lot, subject to the provisions of this Declaration, including without limitation, the provisions of this Declaration. If any chutes, flues, ducts, conduits, wires, pipes, plumbing, or any other apparatus or facilities for the furnishing of utilities or other services to a Lot is conveyed therewith and lies partially within and partially outside of the designated boundaries of the Lot in question, any portions thereof which serve only such Lot shall be deemed to be a part of such Lot and any portions thereof which serve more than one Lot or any portion of the Common Area, shall be deemed to be part of the Common Area. The ownership of each Lot shall include, and there shall pass with the title to each such Lot as an appurtenance thereto, whether or not separately described, all rights of a Member in the Association and all of the right and interest of use in and to the Common Area as set forth herein.

ARTICLE 3: MEMBERSHIP AND VOTING RIGHTS

3.1 <u>Membership</u>. Subject to Section 2 of this Article, every person who is the record owner of a fee or undivided fee interest in any Lot which is subject to this Declaration shall have a membership in the Association. Membership shall be appurtenant to and may not be separated from

ownership of any such Lot, and such ownership of a Lot which is subject to this Declaration shall be the sole qualification for such membership. In the event that fee title to such a Lot is transferred or otherwise conveyed, the membership in the Association which is appurtenant thereto shall automatically pass to such transferee, notwithstanding any failure of the transferror to endorse to his transferee any certificates or other evidences of such membership. The foregoing is not intended to include persons who had an interest merely as security for the performance of an obligation, and the giving of a security interest shall not terminate the Owner's membership.

- 3.2 <u>Multiple Owners</u>. No Owner, whether one or more persons, shall have more than one (1) membership per Lot owned; provided, however, multiple use rights for multiple Owners shall exist subject, however, to the right of the Board to regulate and limit use by multiple Owners. Each Owner, by acceptance of a deed or other conveyance for a Lot, consents and agrees to the dilution of his or her voting interest in the Association by virtue of the submission from time to time of the Additional Property or any portion or portions thereof to the terms of this Declaration as provided herein. The rights and privileges of membership, including the right to vote, may be exercised by a Member or the Member's spouse.
- 3.3 <u>Voting</u>. The Association shall have two (2) classes of membership, Class "A" and Class "B."
- (a) <u>Class "A"</u>. Class "A" members shall be all Owners, including Builder/Owners, with the exception of the Class "B" Member. Class "A" Members shall be entitled on all issues to one (1) vote for each Lot in which they hold the interest required for membership by Section 3.1 hereof.
- (b) <u>Class "B"</u>. The Class "B" member shall be the Declarant. Until termination of the Class "B" membership, as provided for below, the Class "B" member shall be entitled to three (3) times the total number of then existing Class "A" votes and the Class "B" member may appoint the members of the Board of Directors and Officers of the Association. The Class "B" membership shall terminate upon the happening of the earlier of the following:
- (i) unless Declarant has an unexpired option to add Additional Property, one hundred twenty (120) days after one hundred percent (100%) percent of the Lots contemplated to be part of the Community on the master plan thereof, inclusive of Lots as may exist or come to exist with the Additional Property, have been conveyed to Owners other than Builder/Owners or affiliates of Declarant;
 - (ii) when, in its discretion, the Declarant so determines; or
 - (iii) on December 31, 2024.

From and after the happening of these events, whichever occurs earlier, the Class "B" member shall be deemed to be a Class "A" member entitled to one (1) vote for each Lot in which it holds the interest required for membership under Section 3.1 hereof.

- 3.4 Exercise of Voting Rights. In any situation where a Member is entitled to exercise a vote for a Lot and there is more than one (1) Owner of the Lot, the vote for such Lot shall be exercised as the co-Owners determine among themselves. In the event more than one of the co-Owners seek to exercise the vote or if any one of such co-Owners disputes any others right to so vote, the Lot's vote shall be suspended.
- 3.5 Declarant Control. Notwithstanding any other provision to the contrary within this Declaration, the Articles of Incorporation or the By-Laws for the Association, Declarant hereby retains the authority and right to appoint and remove any member or members of the Board of Directors of the Association and any officer or officers of the Association until such time as the Class B membership terminates. Every Grantee of any interest in the Properties, by acceptance of a deed or other conveyance of such interest, agrees that Declarant shall have the authority to appoint and remove directors and officers of the Association in accordance with the foregoing provisions of this section. Upon the expiration of the period in which Declarant maintains the authority and right to appoint and remove members of the Board and officers of the Association, such right shall pass to the Members, including Declarant, if Declarant then owns one or more Lots, and a special meeting of the Association shall be called within one hundred twenty days (120) thereafter. At such special meeting, the Members shall elect a new Board which shall undertake the responsibilities of the Board, and Declarant shall deliver all books, accounts, and records, if any, which Declarant has kept on behalf of the Association and any agreements or contracts, executed by or on behalf of the Association during such period in which Declarant has in its possession.

ARTICLE 4: RIGHTS AND OBLIGATIONS OF THE ASSOCIATION

- 4.1 <u>Function of Association</u>. The Association shall be the entity responsible for management, maintenance, operation and control of the Area of Common Responsibility and all improvements thereon. The Association shall be the primary entity responsible for enforcement of the Governing Documents and such reasonable rules regulating use of the Properties as the Board may adopt pursuant to Article 10. The Association shall also be responsible for administering and enforcing the architectural standards and controls set forth in this Declaration and in the Design Guidelines. The Association shall perform its functions in accordance with the Governing Documents and the laws of the State of North Carolina.
- 4.2 <u>Common Area</u>. The Association, subject to the rights of the Owners set forth in this Declaration, shall manage and control the Common Area and all improvements thereon (including, without limitation, sidewalks, landscaping, recreational facilities, furnishings, equipment, and other personal property of the Association), and shall keep it in attractive condition and good repair, consistent with the Community-Wide Standard. The Board is specifically authorized, but not obligated, to retain or employ professional management to assist in carrying out the Association's responsibilities under this Declaration, the cost of which shall be a Common Expense.
- 4.3 <u>Services</u>. The Association may obtain and pay for the services of any person or entity to manage its affairs or any part thereof and any other personnel as the Association's Board of Directors shall determine to be necessary or desirable for the proper operation of the Properties.

Such personnel may be furnished or employed directly by the Association or by any person or entity with whom or with which it contracts. The Association may obtain and pay for legal, accounting and any other professional services necessary or desirable in connection with the operation of the Properties or the enforcement of this Declaration. The Association may, but shall not be required to, arrange as an Association expense with others to furnish trash collection, cable television, and other common services to each Lot or Residential Unit within the Community. All costs and expenses incident to any of the foregoing shall be a Common Expense payable by the Association.

- 4.4 Personal Property and Real Property for Common Use. The Association through the action of its Board may acquire, hold, and dispose of tangible and intangible personal property and real property. The Declarant and its designees, with the Declarant's prior written consent, may convey to the Association improved or unimproved real estate, or interests in real estate, located within the property described in **Exhibit A**, personal property and leasehold and other property interests. Such property shall be accepted by the Association and thereafter shall be maintained by the Association at its expense for the benefit of its Members. Declarant shall not be required to make any improvements or repairs whatsoever to property to be conveyed and accepted pursuant to this Section 4.4.
- 4.5 <u>Power to Contract</u>. The Association may, acting through its Board, enter and make contracts with any and all Parties as determined by the Board, including contracts with any other residential or commercial associations, Parcels, or neighborhoods within or adjacent to the Community to provide services and/or perform services on behalf of such other association, Parcel, or neighborhood. The Association may, acting through its Board, contract with any governmental division, department or agency for such to provide services on its or its Members' behalf.

4.6 Rules and Regulations.

- (a) The Association, through its Board, may establish reasonable Rules and Regulations concerning the use of the Common Area and improvements located thereon. Rules or Regulations shall not, however, diminish, alter, interfere with, or affect the rights of use, easements, permits, privileges, or licenses provided to Declarant and its respective invitees, licensees, guests, successors and assigns. Furthermore, no Rule or Regulation shall affect or treat Declarant, and its respective invitees, licensees, guests, successors and assigns in a manner more restrictively than the Association's Rules and Regulations may affect or treat its Class "A" members. Copies of such Rules and Regulations and amendments thereto shall be furnished by the Association to all Owners prior to the Rules and Regulations effective date. Such Rules and Regulations shall be binding upon the Owners and users, their families, tenants, guests, invitees, and agents, until and unless such Rules or Regulations requirement is specifically overruled, canceled, or modified by the Board.
- (b) All provisions of the Governing Documents which govern the conduct of Owners and which provide for sanctions against Owners shall also apply to all occupants of Lots and guests and invitees of occupants or Owners. The Owner of each Lot shall be responsible for ensuring that the occupant, and the guests, invitees and licensees of the Owner or the occupants strictly comply with all provisions of the Governing Documents. Fines may be levied against Owners or occupants for violations of the Governing Documents. If a fine is levied first against

an occupant and is not paid within thirty (30) days, the fine may then be imposed against the Owner of the Lot wherein the occupant resides.

4.7 Enforcement.

- (a) The Board may impose sanctions for violation of the Governing Documents after compliance with the notice and hearing procedures set forth in Section 4.7(g) below. In the event that any occupant, guest or invitee of a Lot violates the Governing Documents, the Board or any committee established by the Board, with the Board's approval, may sanction such occupant, guest or invitee and/or the Owner of the Lot that the violator is occupying or visiting. Sanctions may include the actions enumerated below. In every instance in which the Board may act, any committee established and approved by the Board, may act in the Board's stead.
- (i) The Board may impose reasonable monetary fines which shall constitute a lien upon the Lot of the violator. A fine not exceeding One Hundred Fifty Dollars (\$150.00) may be imposed for any one violation. Without further hearing a fine may be imposed for each day of a continuing violation commencing on the day the Board's decision is made to impose the fine until the violation is cured. Collection of fines may be enforced against an Owner as if such charges were a Common Expense owed by the Owner involved. In the event that any occupant, guest or invitee of a Lot violates the Governing Documents and a fine is imposed, the fine shall first be assessed against the occupant; provided however, if the fine is not paid by the occupant within the time period set by the Board, the Owner shall pay the fine upon notice from the Board.
 - (ii) The Board may suspend an Owner's right to vote.
- (iii) The Board may suspend any Person's right to use any recreational facilities within the Common Area; provided however, nothing herein shall authorize the Board to limit ingress or egress to or from a Lot.
- (iv) The Board may suspend any services provided by the Association to an Owner or the Owner's Lot if the Owner is more than thirty (30) Days delinquent in paying any assessment or other charge owed to the Association.
- (v) The Board may levy Specific Assessments to cover costs incurred in bringing a Lot into compliance in accordance with Section 8.5(b) and 9.12.
- (b) The Association may also elect to enforce the provisions of the Governing Documents by filing suit at law to recover monetary damages or in equity to enjoin any violation, or both.
- (c) In addition, the Board, or the covenants committee if established, may elect to enforce any provision of the Governing Documents by self-help as more particularly described in Section 5.2(c) and 9.10 (specifically including, but not limited to the filing of liens for non-payment of assessments and/or notices of violations in the Public Records, the towing of

vehicles that are in violation of parking rules and the removal of pets that are in violation of pet rules). Entry onto a Lot pursuant to this Section 4.7 shall not be deemed a trespass.

- (d) In any action to enforce the provisions of this Declaration, the By-Laws, any Supplemental Declaration, or any rule or regulation, if the Association prevails, it shall be entitled to recover all costs, including, without limitation, reasonable attorneys' fees and court costs, incurred in such action.
- (e) The Association shall not be obligated to take action to enforce any covenant, restriction, or rule which the Board in the exercise of its business judgment determines is, or is likely to be construed as, inconsistent with applicable law, or in any case in which the Board reasonably determines that the Association's position is not strong enough to justify taking enforcement action. Any such determination shall not be construed a waiver of the right of the Association to enforce such provision under any circumstances or estop the Association from enforcing any other provision of the Governing Documents.
- (f) The Association, by contract or other agreement, may enforce County, City, State or Federal ordinances, laws, or rules if applicable, and permit local governments to enforce ordinances on the Properties for the benefit of the Association and its Members.
- (g) The Board shall not impose a fine, suspend the right to vote or suspend the right to use the Common Area (provided, however, a late charge for failure to pay any assessment pursuant to Article 8 of this Declaration shall not be considered a fine for purposes of this paragraph), unless and until the Association has sent or delivered written notice to the violator as provided in subsection (i) below. Any such fine or fines may be effective or commence upon the sending of such notice or such later date as may be set forth in such notice, notwithstanding the violator's right to request a hearing before the Board to challenge such fine under subsection (ii) below.
- (i) <u>Notice</u>. If any provision of the Declaration or Bylaws or any rule or regulation of the Association is violated, the Board shall send the violator written notice identifying the violation and fine(s) being imposed and advising the violator of the right to request a hearing before the Board to contest the violation or fine(s) or to request reconsideration of the fine(s). Fine(s) may be effective or commence upon the sending of such notice or such later date specified in such notice, notwithstanding the violator's right to request a hearing before the Board to challenge the fine. In the event of a continuing violation, each day the violation continues or occurs again constitutes a separate offense, and fines may be imposed on a per diem basis without further notice to the violator.
- (ii) <u>Hearing</u>. If a written request for hearing is received from the violator within ten (10) days of the date of the violation notice provided above, then the Board shall schedule and hold in executive session a hearing affording the violator a reasonable opportunity to be heard. The minutes of the meeting shall contain a written statement of the results of the hearing. The Board may establish rules of conduct for such hearing, which may include limits on time and on the number of participants who may be present at one time.

- 4.8 <u>Implied Rights.</u> The Association may exercise any right or privilege given to it expressly by the Governing Documents, or reasonably implied from or reasonably necessary to effectuate any such right or privilege.
- Indemnification. The Association shall indemnify every officer, director, and 4.9 Design Review Board or committee member against all damages, liabilities, and expenses, including reasonable attorneys fees, incurred in connection with any action, suit, or other proceeding (including settlement of any suit or proceeding, if approved by the then Board of Directors) to which he or she may be a party by reason of being or having been an officer, director, Design Review Board or committee member, except that such obligation to indemnify shall be limited to those actions for which liability is limited under this Section 4.9, the Articles of Incorporation and North Carolina law. The officers, directors, and Design Review Board or committee members shall not be liable for any mistake of judgment, negligent or otherwise, except for their own individual willful misfeasance, malfeasance, misconduct, or bad faith. The officers, directors, and Design Review Board or committee members shall have no personal liability with respect to any contract or other commitment made or action taken in good faith on behalf of the Association (except to the extent that such officers, directors, or Design Review Board or committee members may also be Members of the Association). The Association shall indemnify and forever hold each such officer, director, and Design Review Board or committee member harmless from any and all liability to others on account of any such contract, commitment or action. This right to indemnification shall not be exclusive of any other rights to which any present or former officer, director, or Design Review Board or committee member may be entitled. The Association shall, as a Common Expense, maintain adequate general liability and officers' and directors' liability insurance to fund this obligation, if such insurance is reasonably available.
- 4.10 Security. The Association may, but shall not be obligated to, maintain or support certain activities within the Properties designed to make the Properties safer than they otherwise might be. Neither the Association, the original Declarant, nor any successor Declarant shall in any way be considered insurers or guarantors of security within the Properties, nor shall any of them be held liable for any loss or damage by reason of failure to provide adequate security or the ineffectiveness of security measures undertaken. No representation or warranty is made that any system or measure, including any monitoring system or any mechanism or system for limiting access to the Properties, cannot be compromised or circumvented, nor that any such system or security measure undertaken will in all cases prevent loss or provide the detection or protection for which the system is designed or intended. Each Owner acknowledges, understands and covenants to inform its tenants and all occupants of its Lot that the Association, its Board of Directors and committees, Declarant, and any successor Declarant are not insurers and that each Person using the Properties assumes all risks of personal injury and loss or damage to property, including Lots and the contents of Lots, resulting from acts of third parties.
- 4.11 <u>Power to Assess</u>. The Association shall have the right and power, as more particularly set forth in this Declaration, to fix, levy, collect, and enforce payment by any lawful means, all charges and assessments pursuant to the terms of this Declaration, to pay all expenses in

connection therewith, and all office and other expenses incident to the conduct and affairs of the business of the Association.

4.12 <u>Board Right to Act</u>. To the extent not otherwise required by the provisions of the North Carolina Non-Profit Corporation Act, this Declaration, the By-Laws, or the Articles of Incorporation, the powers granted to the Association by this Declaration or the Articles of Incorporation or By-Laws of the Association shall be exercised by the Board of Directors, acting through the Officers of the Association, without any further consent or action on the part of the Members.

ARTICLE 5: MAINTENANCE

5.1 Association's Responsibility.

- (a) The Association shall maintain and keep in good condition, order and repair the Area of Common Responsibility, which shall include, but need not be limited to:
 - (i) Common Area and the facilities located therein;
- (ii) all landscaping and other flora and all structures and improvements, including but not limited to any gates, gate house, entry features, fencing, private streets, parking areas, sidewalks, streetlights, parks, trails, pools, cabana, and tennis courts situated upon the Common Area;
- (iii) all furnishings, equipment and other personal property of the Association;
- (iv) any landscaping and other flora, buffers, entry features, fencing, streetlights, structures and improvements within public rights-of-way within or abutting the Properties or upon such other public land adjacent to the Properties as deemed necessary in the discretion of the Board; and
- (v) such additional portions of any property included within the Area of Common Responsibility as may be dictated by the Governing Document, or any contract or agreement for maintenance thereof entered into by the Association.
- (b) The Association may, as a Common Expense, maintain other property and improvements which it does not own, including without limitation property dedicated to the public, or provide maintenance or services related to such property over and above the level being provided by the property owner, if the Board of Directors determines that such maintenance is necessary or desirable to maintain the Community-Wide Standard.
- (c) The Association shall maintain the facilities and equipment within the Area of Common Responsibility in continuous operation, except for any periods necessary, as determined in the sole discretion of the Board, to perform required maintenance or repairs, unless

Members holding sixty-seven percent (67%) of the Class "A" votes in the Association, and the Declarant during the Development Period, agree in writing to discontinue such operation.

- (d) The Association may be relieved of all or any portion of its maintenance responsibilities herein to the extent that (i) such maintenance responsibility is otherwise assumed by or assigned to one (1) or more Owners or (ii) such property is dedicated to any local, state, or federal government or quasi-governmental entity; provided, however, that in connection with such assumption, assignment or dedication, the Association may reserve or assume the right or obligation to continue to perform all or any portion of its maintenance responsibilities, if the Board determines that such maintenance is necessary or desirable to maintain the Community-Wide Standard. Except as provided above, the Area of Common Responsibility shall not be reduced by amendment of this Declaration or any other means during the Development Period except with the written consent of the Declarant.
- (e) Except as otherwise specifically provided herein, all costs associated with maintenance, repair and replacement of the Area of Common Responsibility shall be a Common Expense to be allocated among all Lots as part of the General Assessment, without prejudice to the right of the Association to seek reimbursement from the Owner(s) of, or other Persons responsible for, certain portions of the Area of Common Responsibility pursuant to the Governing Documents, any recorded covenants, or any agreements with the owner(s) thereof. All costs associated with maintenance, repair and replacement of Exclusive Common Area may, at the election of the Board, be assessed as a Specific Assessment against the Lot or Lots to which the Exclusive Common Area is assigned.
- (f) The Association shall not be liable for any injury or damage to any personal property (a) caused by the elements; (b) resulting from any rain or other surface water which may leak or flow from any portion of the Common Area; or (c) caused by any pipe, plumbing, drain, conduit, appliance, equipment, security system or utility line or facility, the Association is obligated to maintain. Nor shall the Association be liable to any Owner for any loss or damage, by theft or otherwise, of any property of such Owner or its respective guests, invitees, successors or assigns which may be stored in or upon any portion of the Common Area or any portion of the Properties. No diminution or abatement of assessments shall be claimed or allowed by reason of any alleged failure of the Association to take some action or to perform such function required to be taken or performed by the Association under this Declaration, or for inconvenience or discomfort arising from the making of improvements or repairs which are the responsibility of the Association, or from any action taken by the Association to comply with any law, ordinance, or with any order or directive of any municipal or governmental authority, the obligation to pay each such assessment being a separate and independent covenant on the part of each Owner.

5.2 Owner's Responsibility.

(a) Each Owner shall maintain his or her Lot, including the Residential Unit thereon, and all structures, parking areas, sprinkler and irrigation systems, landscaping and other flora, and other improvements comprising the Lot, including the Residential Unit thereon, in a manner consistent with the Community-Wide Standard and the Governing Documents, unless such maintenance responsibility is otherwise assumed by or assigned to the Association. Each

Owner shall also maintain the driveway and any mailbox or mail facility serving his or her Lot. Any sidewalks located on the Lot or within the right-of-way immediately adjacent to the Owner's Lot shall be maintained by the Association and shall not be the Owner's responsibility.

- In the event the Board determines that (i) any Owner has failed or refused to discharge properly his or her obligations with regard to the maintenance, repair, or replacement of items for which he or she is responsible hereunder; or (ii) that the need for maintenance, repair, or replacement which is the responsibility of the Association hereunder is caused through the willful or negligent act of an Owner, or his or her family, guests, lessees, or invitees, the Association, except in the event of an emergency situation, shall give the Owner written notice of the Association's intent to provide such necessary maintenance, repair, or replacement, at their sole cost and expense; the notice shall set forth with reasonable particularity the maintenance, repair, or replacement deemed necessary and the cost thereof. The noticed party shall have ten (10) days within which to pay such amount claimed; or, in the event such maintenance or repair is to the Owner's Residential Unit or Lot, complete said maintenance, repair, or replacement; or, in the event that such maintenance, repair, or replacement is not capable of completion within said ten (10) day period, to commence such work within said ten (10) day period which shall be completed within a reasonable time. If any Owner does not comply with the provisions hereof, the Association may provide any such maintenance, repair, or replacement at such person's sole cost and expense, and the cost shall be added to and become a part of the assessment to which such party is subject and shall become a lien against the Lot of such party collectible as an assessment pursuant to Article 8 hereof. By way of example and not limitation, in the event of an Owner's failure to comply with the provisions hereof, the Association may cause any weeds, grass, trees or landscaping to be cut, pruned or removed, as the case may necessitate, and may remove or cause to be removed such garbage, trash or rubbish as has accumulated thereon. Such shall expressly include removal of dead or diseased trees or landscaping.
- (c) In addition to the foregoing, the Association may do anything necessary to secure compliance with the Governing Documents so as to place said Lot in a neat, attractive, healthful, and sanitary condition and the charges incurred for securing such compliance, including the cutting, trimming, pruning, or removal of weeds, grass, landscaping, trees or such garbage, trash or rubbish as may be removed, may be charged to the Owner or occupant of such Lot for the cost of such work. The Owner or occupant, as the case may be, agrees by the purchase or occupancy of the property to pay such statement immediately upon receipt thereof, and such payments made by the Association shall be secured by a lien in the same manner as General Assessments paid the Association under this Declaration.
- 5.3 <u>Standard of Performance</u>. All maintenance shall be performed in a manner consistent with the Community-Wide Standard and the Governing Documents. Maintenance responsibilities shall include the responsibility for repair and replacement as necessary. Neither the Association nor any Owner shall be liable for any damage or injury occurring on, or arising out of the condition of, property which it does not own except to the extent that it has been negligent in the performance of its maintenance responsibilities.
- 5.4 <u>Cost Sharing Agreements</u>. Adjacent to or in the vicinity of the Properties, there may be neighboring property which is not subject to this Declaration (hereinafter "Adjacent

Properties") at the time of initial recordation of this Declaration in the Public Records. The owners of the Adjacent Properties may not be Members of the Association and if they are not they shall not be entitled to vote, nor shall they be subject to assessment under Article 8 of this The Association may enter into agreements with the owners or operators of portions of the Adjacent Properties which (a) obligate the owners or operators of the Adjacent Properties to share in certain costs associated with the maintenance, repair, replacement and insuring of portions of the Area of Common Responsibility, if any, which are used by or benefit jointly the owners of the Adjacent Properties and the Owners within the Properties; (b) permit use of any recreational and other facilities located on the Adjacent Properties by the Owners within the Properties, and/or; (c) obligate the Association to share in certain costs associated with the maintenance, repair, replacement and insuring of portions of the Adjacent Properties, if any, which are used by or benefit jointly the owners of the Adjacent Properties and the Owners within the Properties. The owners of the Adjacent Properties shall be subject to assessment by the Association only in accordance with the provisions of such agreement(s) unless all or any portion of such property is annexed to this Declaration by the Declarant. The owners of the Adjacent Properties shall not be subject to the restrictions contained in this Declaration except as otherwise specifically provided herein, or in a Cost Sharing Agreement or declaration of easements.

ARTICLE 6: INSURANCE AND CASUALTY LOSSES

6.1 Insurance and Casualty or Liability Losses. Commencing not later than the time of the first conveyance of a Lot to a person other than the Declarant, the Association's Board of Directors or its duly authorized agent shall have the authority to and shall obtain insurance for all insurable improvements on the Common Area against loss or damage by fire or other hazards, including extended coverage, vandalism, and malicious mischief. This total amount of insurance after application of deductibles shall not be less than eighty percent (80%) of the replacement cost of the insured property at the time the insurance is purchased and at each renewal date, exclusive of land, excavations, foundations and other items normally excluded from insurance The Board shall also obtain a public liability policy applicable to the Common Area covering the Association, its officers, directors, members, and agents. The public liability policy shall have at least a One Million (\$1,000,000.00) Dollar each occurrence limit (combined single limit (C.S.L.) for death, bodily injury and property damage), Two Million (\$2,000,000.00) Dollar general aggregate, C.S.L., and Two Million (\$2,000,000.00) Dollar products/completed operations aggregate, C.S.L. Unless otherwise provided by the Board of Directors, the cost of all such insurance coverage shall be paid from the Common Expense. Each insurance policy may contain a deductible, and the amount thereof shall be added to the face amount of the policy in determining whether the insurance at least equals the full replacement cost. The Association may allocate equitably the payment of a reasonable insurance deductible between the Association and the Owners affected by a casualty against which the Association is required to insure, provided, however, that the amount of deductible which can be allocated to any one Owner shall not exceed \$1,000.00 per casualty loss.

All such insurance coverage obtained by the Board of Directors shall be written in the name of the Association, as Trustee, for the respective benefited parties, as further identified in (b) below. Such insurance shall be governed by the provisions hereinafter set forth:

- (a) All policies shall be written with a company licensed to do business in North Carolina and holding a rating of A-X1 or better in the Financial Category as established by A.M. Best Company, Inc., if available, or, if not available, the most nearly equivalent rating.
- (b) All policies on the Common Area shall be for the benefit of the Association and the Owners and their Mortgagees, as their interests may appear.
- (c) Exclusive authority to adjust losses under policies in force on any portion of the Properties, including the Common Area, obtained by the Association shall be vested in the Association's Board of Directors; provided, however, that no Mortgagee having an interest in such losses may be prohibited from participating in the settlement negotiations, if any, related thereto.
- (d) All casualty insurance policies shall have an inflation guard endorsement and an agreed amount endorsement, if reasonably available, with an annual review by one or more qualified persons, at least one of whom must be in the insurance or real estate industry and familiar with insurance for improvements in the County.
 - (e) Insurance policies carried pursuant to this Section 6.1 shall provide that:
- (i) Each Lot Owner is an insured person under the policy to the extent of the Lot Owner's insurable interest;
- (ii) The insurer waives its right to subrogation under the policy against any Lot Owner or member of the Lot Owner's household;
- (iii) No act or omission by any Lot Owner, unless acting within the scope of the Owner's authority on behalf of the Association, will preclude recovery under the policy; and
- (iv) If, at the time of a loss under the policy, there is other insurance in the name of a Lot Owner covering the same risk covered by the policy, the Association's policy provides primary insurance.
- (f) The Association's Board of Directors shall be required to make every reasonable effort to secure insurance policies that will provide for the following:
- (i) a waiver of subrogation by the insurer as to any claims against the Association's Board of Directors and its Manager;
- (ii) a waiver by the insurer of its rights to repair and reconstruct instead of paying cash;

- (iii) that no policy may be canceled, invalidated or suspended on account of any defect or of the conduct of any director, officer, or employee of the Association or its duly authorized manager without prior demand in writing delivered to the Association to cure the defect or to cease the conduct and the allowance of a reasonable time thereafter within which the defect may be cured by the Association, its manager, any Owner or Mortgagee;
- (iv) a waiver of the insurer's right to cancel without first giving thirty (30) days prior written notice of such cancellation to the Association and to any Mortgagee to which a Mortgagee endorsement has been issued; and
- (v) all liability insurance policies shall contain cross-liability endorsements to cover liability of the Association to an individual Owner and shall also name the Declarant as an additional insured.
- (g) It shall be the responsibility of each Owner at his own expense, to provide public liability, property damage, title, and other insurance with respect to such Owner's Lot. The Association's Board of Directors may require all Owners to carry public liability and property damage insurance with respect to their respective Lots and to furnish copies of certificates thereof to the Association.

In addition to the other insurance required by this Section 6.1, the Board shall obtain, as a Common Expense, worker's compensation insurance, if and to the extent required by law, and a fidelity bond or bonds (provided that the Community includes Lots upon which financing exists that is held or serviced by any financial agency, corporation, or secondary mortgage market enterprise which requires the maintenance of such fidelity bond) on directors, officers, employees, and other persons handling or responsible for the Association's funds. The amount of fidelity coverage shall be determined in the directors' best business judgment, but may not be less than three (3) months' assessments, plus reserves on hand if such bond is obtained. Bonds shall contain a waiver of all defenses based upon the exclusion of persons serving without compensation and may not be canceled or substantially modified without at least ten (10) days' prior written notice to the Association.

Section 6.2. <u>Disbursement of Proceeds</u>. Any loss covered by the property hazard policy described under Section 6.1 above, shall be adjusted with the Association, but the insurance proceeds for that loss are payable to any insurance trustee designated for that purpose, or otherwise to the Association, and not to any mortgagee or beneficiary under the deed of trust. The insurance trustee or the Association shall hold any insurance proceeds in trust for Lot Owners and lienholders as their interests may appear. Subject to the provisions of Section 6.3, the proceeds shall be disbursed first for the repair or restoration of the damaged property, and Lot Owners and lienholders are not entitled to receive payment of any portion of the proceeds unless there is a surplus of proceeds after the property has been completely repaired or restored, or the planned community is terminated.

Section 6.3. <u>Damage and Destruction</u>. Any portion of the Community for which insurance is required under Section 6.1 of this Article which is damaged or destroyed shall be

repaired or replaced promptly by the Association unless (i) the planned community is terminated; (ii) repair or replacement would be illegal under any state or local health or safety statute or ordinance; or (iii) the Lot Owners decide not to rebuild by an eighty percent (80%) vote, including one hundred percent (100%) approval of Lot Owners assigned to Limited Common Area not to be rebuilt. The cost of repair or replacement in excess of insurance proceeds and reserves is a Common Expense. If any portion of the Community is not repaired or replaced, (i) the insurance proceeds attributable to the damaged Common Area shall be used to restore the damaged area to a condition compatible with the remainder of the Community; and (ii); the remainder of the proceeds shall be distributed to all the Lot Owners or lienholders, as their interests may appear, in proportion to the Common Expense liabilities of all the Lots. Notwithstanding the above, Section 47F-2-118 of the North Carolina Planned Community Act (termination of the planned community) governs the distribution of insurance proceeds if the Community is terminated.

Section 6.4. Repair and Reconstruction. If the damage or destruction for which the insurance proceeds are paid is to be repaired or reconstructed and such proceeds are not sufficient to defray the cost thereof, the Board of Directors shall, without the necessity of a vote of the Association's members, levy Specific Assessment against all Owners ultimately responsible for the payment of the policy premium in the same proportion as an Owner's assessment bears to the Association's budget. Additional assessments may be made in like manner at any time during or following the completion of any repair or reconstruction. If the funds available from insurance exceed the cost of repair, such excess shall be deposited to the benefit of the Association.

Section 6.5. Lot Owner's Responsibility. By virtue of taking title to a Lot each Owner of a Lot covenants and agrees with all other Owners and with the Association to carry all risk casualty insurance in the event the Association does not carry blanket all-risk casualty insurance on improvements on Lots. Each Owner, if any, further covenants and agrees that in the event of a partial loss or damage and destruction resulting in less than total destruction of any structure located on a Lot, he or she shall proceed promptly to repair or reconstruct the damaged structure in a manner consistent with the aesthetic appearance and quality of the original construction and the Community-Wide Standards. In the event that any Residential Unit is totally destroyed or rendered uninhabitable or unusable and the Owner or Owners thereof determines not to rebuild or reconstruct, then that Owner or Owners shall clear that Lot of all debris and return it to substantially the natural state in which it existed prior to the beginning of construction. The obligation of an Owner hereunder specified shall not be applicable to any Owner whose Lot is insured under a casualty insurance policy obtained by the Association such Owner's behalf. In the event that any Residential Unit is totally destroyed or rendered uninhabitable or unusable and the Owner or Owners thereof determines to rebuild or reconstruct, then such Owner or Owners shall repair or rebuild such Residential Unit to substantially the same condition as existed prior to such fire or other casualty and in accordance with all applicable standards, restrictions, and provisions of this Declaration (including, without limitation, Article 9 hereof) and all applicable zoning, subdivision, building, and other governmental regulations. All such work of repair and construction, as identified herein, shall be commenced promptly following such damage or destruction and shall be carried through diligently to conclusion within a reasonable time.

ARTICLE 7: ANNEXATION AND WITHDRAWAL OF PROPERTY

7.1 Annexation Without Approval of Class "A" Membership. As the owner thereof, or if not the owner, with the consent of the owner thereof, Declarant shall have the unilateral right, privilege, and option (but not the obligation), from time to time at any time until twenty (20) years from the date this Declaration is recorded in the Public Records, subject to the provisions of this Declaration and the jurisdiction of the Association, pursuant to the terms and conditions contained within this Declaration, all or any portion of the Additional Property, whether in fee simple or leasehold, by filing in the Public Records, an amendment or Supplemental Declaration annexing all or any portion of such property. Such amendments to this Declaration or Supplemental Declaration shall not require the vote of Members. Any such annexation shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein. Such amendment or Supplemental Declaration may specify such specific use restrictions and other covenants, conditions, and restrictions to be applicable to the annexed property as Declarant may, in its own discretion, determine.

Declarant shall have the unilateral right to transfer to one or more other persons the right, privilege, and option to annex the Additional Property which is herein reserved to Declarant, provided that such transferee or assignee shall be the developer of at least that part of said Additional Property to which such right, privilege, and option is assigned; and provided, further, such assignment shall not remove or alter Declarant's further right, option and privilege to annex.

- Annexation With Approval of Class "A" Membership. Subject to the written consent of the owner thereof, upon the written consent or affirmative vote of a majority of the Class "A" members present or represented by proxy at a meeting duly called for such purpose, and of the Declarant during the Development Period, the Association may annex real property other than the Additional Property, and following the expiration of the right in this Article 7, Section 7.1, the Additional Property, to the provisions of this Declaration and the jurisdiction of the Association by filing for record in the Public Records, an amendment, or Supplemental Declaration in respect to the property being annexed. Any such amendment or Supplemental Declaration shall be signed by the President and the Secretary of the Association, and any such annexation shall be effective upon recordation in the Public Records, unless otherwise provided therein. The time within which and the manner in which notice of any such meeting of the Class "A" members of the Association, called for the purpose of determining whether Additional Property pursuant to this section shall be annexed, and the quorum required for the transaction of business at any such meeting, shall be as specified in the By-Laws of the Association for regular or special meetings, as the case may be.
- 7.3 No Restrictions. The Additional Property, by initial recordation of this Declaration, is not restricted by the terms of such Declaration, in any manner other than for the express benefits as might be granted to the Additional Property under the terms of this Declaration. Following the date of initial recordation of this Declaration, the Declarant may, but shall not be required to, annex any portion or portions of such Additional Property. This Declaration shall only be applicable, except for the benefits as might be expressly bestowed upon the Additional Property under the terms of this Declaration as initially recorded, to those portions of the Additional Property as expressly made subject hereto by the Declarant or its successor and assigns in interest to the Additional Property. Until such time as a portion or portions of the Additional Property are made

subject by express amendment to the terms of this Declaration or by Supplemental Declaration, no other portion of the Additional Property shall be, by implication, innuendo, or otherwise, subject to the terms of this Declaration other than for the benefits as expressly might be bestowed upon the Additional Property under the terms of this Declaration.

- (a) Portions of the Additional Property (together with additions thereto) may be added, as provided herein, to the Properties at different times, and there are no limitations fixing the boundaries of those portions or regulating the order, sequence, or location in which any of such portions may be added to the Properties. The exercise of the option to submit a portion or portions of the Additional Property to this Declaration shall not bar the further exercise of this option as to other portions or the balance of the Additional Property.
- (b) If the Additional Property or any portion or portions thereof are added to the Properties, Declarant reserves the right to designate the boundaries of the Lots, as well as the Common Area, if any, to be added to the Properties in connection therewith.
- (c) Should the options to add the Additional Property, or any portion or portions thereof, not be exercised within the term specified herein or be terminated by Declarant, such option shall in all respects expire and be of no further force and effect. In the event that such option expires or is terminated, as aforesaid, Declarant shall not be obligated to impose on the Additional Property or any portion thereof any covenants, conditions, or restrictions whatsoever.
- (d) The option hereby reserved by Declarant to cause all or any portion or portions of the Additional Property to become part of the Properties shall in no way be construed to impose upon Declarant any commitment or obligation to add all or any portion or portions of the Additional Property to the Properties or to construct thereon any improvements of any nature whatsoever.
- (e) If the Additional Property or any portion or portions thereof is added to the Properties, then from and after the addition to the Properties of the Additional Property or such portion or portions by such amendment to this Declaration or by Supplemental Declaration, the number of votes in the Association shall be increased by the number of Lots to be located on the Additional Property or such portion or portions thereof as are added so that there shall continue to be, subject to the provisions hereof, one vote in the Association per Lot in the Community.
- 7.4 <u>Amendment</u>. This Article 7 shall not be amended during the Development Period without the prior written consent of Declarant.
- 7.5 Withdrawal of Property. During the Development Period, Declarant reserves the right to amend this Declaration unilaterally at any time, without prior notice and without the consent or joinder of any Person, for the purpose of removing certain portions of the Property then owned by the Declarant or its affiliates or the Association from the provisions of this Declaration to the extent originally included in error or as a result of any changes whatsoever in the plans for the Community desired to be effected by the Declarant. No Owner or Builder/Owner shall withdraw any property from the Community without the prior written consent of Declarant.

ARTICLE 8: ASSESSMENTS

8.1 Creation of and Obligation for Assessments.

- (a) The Board may authorize the creation of assessments for Common Expenses of the Association from time to time, as follows: (i) General Assessments; (ii) Special Assessments; (iii) Specific Assessments; and (iv) Parcel Assessments. Each Owner, by accepting a deed or entering into a contract of sale for any portion of the Properties, is deemed to have notice of liability for these assessments and to covenant and agree to pay these assessments.
- (b) There are hereby created General Assessments for Common Expenses as may from time to time be authorized by the Board of Directors. General Assessments shall, from and after the respective Commencement Date (as defined hereinbelow) relating to a respective Lot, be levied against such respective Lot and shall be used to pay expenses determined by the Board to be for the benefit of the Association, its Members, and the Properties as a whole, including, but not limited to, maintenance and insurance of the Area of Common Responsibility, including the Common Area, and expenses otherwise incurred by the Association in accordance with its rights, powers, and privileges for the purposes of promoting the recreation, health, safety, welfare, common benefit and enjoyment of the Owners and occupants of the Properties, and maintaining the Properties and improvements therein. The General Assessment levied against and payable by a Lot shall be equal to the General Assessment which is levied against and payable by each such other Lot. Despite anything contained herein to the contrary, the assessments against a Lot shall not commence until the Commencement Date for such Lot as set forth in Section 8.2.
- (c) There are hereby created Parcel Assessments for Common Expenses as may from time to time be authorized by the Board of Directors. Parcel Assessments shall, from and after the Commencement Date relating to a specific Lot, be levied against such Lots within particular Parcels of the Properties for whose benefit expenses are incurred, such as maintaining and operating facilities and amenities within a Parcel reserved for use of the residents within that Parcel, expenses of enforcing all assessments, covenants, and conditions relating to a specific Parcel, and expenses determined by the Board to be for the benefit of a specific Parcel. Each Lot within a Parcel shall pay a Parcel Assessment equal to each other such Lot within such Parcel. Parcel Assessments established in one Parcel do not need to be equal to Parcel Assessments established in another Parcel. Despite anything contained herein to the contrary, the assessments against a Lot shall not commence until the Commencement Date for such Lot as set forth in Section 8.2.
- (d) All such assessments, together with late charges, simple interest at the rate of eighteen percent (18%) per annum, costs and reasonable attorney's fees actually incurred to enforce or collect such assessments, shall be a charge on the Lot and shall, after remaining unpaid for a period of thirty (30) days or longer and upon the filing of a claim of lien in the office of the Buncombe County, North Carolina Superior Court Clerk, be a continuing lien upon the Lot against which each assessment is made regardless of conveyance thereof. The Association's lien shall be prior and superior to all other liens except the lien for real estate taxes and other governmental assessments and charges against the Lot and liens and encumbrances (specifically including, but not limited to, a mortgage or deed of trust on the Lot) recorded before the docketing of the claim of lien

in the office of the Buncombe County, North Carolina Superior Court Clerk. Each such assessment, together with late fees, interest, costs, and reasonable attorney's fees actually incurred, shall also be the personal obligation of the person who was the owner of such Lot at the time the assessment came due, and his or her grantee shall be jointly and severally liable for such portion thereof as may be due and payable at the time of conveyance; provided, however, any First Mortgagee or holder of a secondary purchase money Mortgage covering the Lot, provided that neither the grantee nor any successor grantee on the secondary purchase money Mortgage is the seller of the Lot who obtains title to a Lot subject to this Declaration pursuant to the remedies provided in such Mortgage or Foreclosure of the Mortgage, will not be liable for such Lot's unpaid assessments which accrued prior to the acquisition of title to such Lot by the Mortgagee. Assessments shall be paid in such manner and on such dates as may be fixed by the Board of Directors, which may include, without limitation, acceleration of the annual assessment for Owners who are delinquent in the payment of the their assessments. The assessments shall be paid annually in advance, unless otherwise provided by the Board. The Association shall, upon request, furnish to any Owner liable for any type of Assessment a written statement signed by the Association officer setting forth whether the Assessment has been paid. This statement shall be conclusive evidence of payment. Association may require the advance payment of a reasonable processing fee for the issuance of such a statement.

- (e) No Owner is exempt from liability for assessments by non-use of Common Area, including Exclusive Common Area reserved for such Owner's use, abandonment of his or her Lot, or any other means. The obligation to pay assessments is a separate and independent covenant on the part of each Owner which runs with title to the Lot. No diminution or abatement of assessments or set-off shall be claimed or allowed for any alleged failure of the Association or Board to take some action or perform some function required of it, or for inconvenience or discomfort arising from the making of repairs or improvements, or from any other action it takes.
- 8.2 Commencement of Assessments. Notwithstanding anything herein to the contrary, any and all assessments provided for in or otherwise assessed pursuant to this Declaration shall commence against a Lot as provided in this Section 8.2. Any and all assessments shall commence in respect to each respective Lot at the time of conveyance of the respective Lot by the Declarant to an Owner or Builder/Owner other than the Declarant; provided, however, that the commencement of the Assessment against a Lot conveyed to a Builder/Owner by Declarant may be delayed as determined by the Declarant from time to time for a term of twelve (12) months from and after the date of conveyance to such Builder/Owner or until substantial completion of the Residential Unit on such Lot, whichever first occurs. Declarant shall not be responsible or liable for the payment of assessments (whether General, Parcel, Special or Specific) in respect to Lots for which Declarant holds record title and which do not contain occupied Residential Units (except as hereinafter provided); provided that Declarant covenants and agrees to pay assessments in the same manner as Lots conveyed to Owners for each Lot owned by Declarant containing an occupied Residential Unit. The date of commencement of the assessment as to any particular Lot, as provided in this Section 8.2, is the "Commencement Date." The first annual assessment for a Lot payable to the Association with respect to such Lot shall be adjusted according to the number of months remaining in the calendar year as of the Commencement Date. Such prorated assessment shall be paid on the Commencement Date or such later date as provided by the Board.

8.3 <u>Computation of General Assessments.</u>

- (a) General Assessments shall be levied equally against all Lots subject to assessment and shall be set at a level which is reasonably expected to produce total income for the Association equal to the total budgeted Common Expenses, including reserves. In determining the total funds to be generated through the levy of General Assessments, the Board, in its discretion, may consider other sources of funds available to the Association, including any surplus from prior years, any assessment income expected to be generated from any additional Lots reasonably anticipated to become subject to assessment during the fiscal year, and any income expected to be generated from any Cost Sharing Agreements.
- It shall be the duty of the Board at least forty-five (45) days prior to the commencement of a fiscal year to prepare and adopt a budget covering the estimated costs of operating the Association and the Properties during the coming year. The budget shall separately list General and Parcel Assessments, if any. Within thirty (30) days after adoption of the proposed budget, the Board shall provide to all Lot Owners a summary of the budget, the General Assessment to be levied therefrom and a notice of a meeting to consider ratification of the budget, including a statement that the budget may be ratified without a quorum. Such meeting shall be held not less than ten (10) nor more than sixty (60) days after mailing of the summary and notice. There shall be no requirement that a quorum be present at the meeting. The budget and assessment established by the Board shall be ratified at the meeting and become and be effective unless Lot Owners representing at least a sixty seven percent (67%) of the votes in the Association vote to reject the budget at the meeting. Notwithstanding the foregoing, in the event that the membership rejects the proposed budget or the Board fails for any reason so to determine the budget for the succeeding year, then the budget and assessments last ratified by the Lot Owners shall be continued until such time as the Lot Owners ratify a subsequent budget proposed by the Board.
- (b) During the Development Period, the Declarant may, but shall not be obligated to, reduce the General Assessment for any fiscal year by payment of a subsidy and/or contributions of services and materials, which may be treated as either a contribution or a loan, in the Declarant's discretion. Any such anticipated payment or contribution by the Declarant shall be disclosed as a line item in the Common Expense budget. Payments by the Declarant in any year shall under no circumstances obligate the Declarant to continue such payments in future years unless otherwise provided in a written agreement between the Association and the Declarant. To the extent that Declarant pays any amount or provides any in-kind services to the Association, any such payment or provision of services shall be a reduction and credit against any amount as may otherwise be claimed owed by Declarant to the Association with subparagraph 8.3(b) hereinabove.

8.4 <u>Special Assessments</u>. In addition to other authorized assessments, the Association may levy Special Assessments from time to time to cover unbudgeted expenses or expenses in excess of those budgeted. Special Assessments shall be allocated equally among all Lots. Any Special Assessment shall become effective unless disapproved at a meeting by Members holding a Majority of the total Class "A" votes and, during the Development Period, by the Declarant. There shall be no obligation to call a meeting for the purpose of considering any Special Assessment except on petition of the Members as provided for in the By-Laws, which petition must be presented to the Board within twenty (20) Days after delivery of the notice of such Special Assessment. Special Assessments shall be payable in such manner and at such times as determined by the Board, and may be payable in installments extending beyond the fiscal year in which the Special Assessment is approved.

8.5 Specific Assessments.

- (a) The Association shall have the power to levy Specific Assessments against a particular Lot or Lots to cover the costs, including overhead and administrative costs, of providing benefits, items, or services to the Lot(s) or occupants thereof upon request of the Owner pursuant to a menu of special services which the Board may from time to time authorize to be offered to Owners and occupants (which may include, without limitation, landscape maintenance, pest control, etc.), which assessments may be levied in advance of the provision of the requested benefit, item or service as a deposit against charges to be incurred by the Owner.
- (b) The Association shall have the power to levy Specific Assessments against a particular Lot or Lots as a fine levied pursuant to Section 4.7 or to cover costs incurred in bringing the Lot(s) into compliance with the terms of the Governing Documents, or costs incurred as a consequence of the conduct of the Owner or occupants of the Lot, their agents, contractors, employees, licensees, invitees, or guests.
- 8.6 Foreclosure of the Lien for Assessments. The lien, when delinquent, may be enforced by suit, judgment, and judicial or nonjudicial Foreclosure, as permitted under North Carolina law. The Association may bid for the Lot at the Foreclosure sale and acquire, hold, lease, mortgage, and convey the Lot. While a Lot is owned by the Association following Foreclosure: (i) no right to vote shall be exercised on its behalf; (ii) no assessment shall be levied on it; and (iii) the other Lots shall be charged, in addition to the usual assessment, a pro rata share of the assessment allocated to the Lot owned by the Association. The Association may sue for unpaid assessments and other charges authorized hereunder without Foreclosing or waiving the lien securing the same. The sale or transfer of any Lot shall not affect the assessment lien or relieve a Lot from the lien for any subsequent assessments. However, the sale or transfer of any Lot pursuant to Foreclosure of the First Priority Mortgage shall extinguish the lien as to any installments of assessments due prior to Foreclosure. A Mortgagee or other purchaser of a Lot who obtains title pursuant to Foreclosure of the Mortgage shall not be personally liable for assessments on such Lot due prior to such acquisition of title. Such unpaid assessments shall be deemed to be Common Expenses collectible from Owners of all Lots subject to assessment under Section 8.2, including such acquirer, its successors and assigns. All other Persons acquiring liens or encumbrances on any Lot after this Declaration has been recorded shall be deemed to consent that the liens or encumbrances shall be inferior to future liens for assessments, as provided

herein, whether or not prior consent is specifically set forth in the instruments creating the liens or encumbrances.

- 8.7 <u>Failure to Assess</u>. Failure of the Board to establish assessment amounts or rates or to deliver or mail each Owner an assessment notice shall not be deemed a waiver, modification, or a release of any Owner from the obligation to pay assessments. In such event, each Owner shall continue to pay General Assessments on the same basis as during the last year for which an assessment was made, if any, until a new assessment is levied, at which time the Association may retroactively assess any shortfalls in collections.
- 8.8 Exempt Property. The following property shall be exempt from payment of assessments: (a) all Common Area and such portions of the property owned by the Declarant as are included in the Area of Common Responsibility pursuant to Section 5.1; and (b) any property dedicated to and accepted by any governmental or quasi-governmental authority or public utility.
- 8.9 <u>Surplus Funds</u>. Any surplus funds of the Association remaining after payment of or provision of Common Expenses, the funding of a reasonable operating expense surplus and any prepayment of reserves, shall be deposited in the reserves of the Association unless otherwise determined by resolution of the Board.

ARTICLE 9: ARCHITECTURAL STANDARDS

- 9.1 <u>General</u>. In order to preserve the natural setting and beauty of the Properties, to establish and preserve a harmonious and aesthetically pleasing design for the Properties, and to protect and promote the character of the Properties, the Lots, the Residential Units and the Common Area and all improvements, structures, landscaping and items located thereon, all Lots, Residential Units and all improvements, structures, landscaping and items located thereon shall be subject to the restrictions set forth in this Article 9. Every grantee of an interest in the Properties, by acceptance of a deed or other conveyance of such interest, agrees to be bound by the provisions of this Article 9.
- 9.2 Design Review Board. No construction, improvements, landscaping, buildings, structures or development of any kind whatsoever, shall be commenced, carried out on, constructed, altered, added to or maintained upon any portion of the Properties, including Lots or Common Area, unless; (a) approved in writing by the Design Review Board; (b) developed, constructed or altered by Declarant; (c) developed, constructed or altered by the Association in respect to the Common Area; or (d) if a Modifications Committee is established pursuant to the terms hereof, approved by the Design Review Board after recommendation from the Modifications Committee. Approval of the Design Review Board shall be subject to such regulations, architectural standards and application procedures as may be promulgated by the Design Review Board and as set forth herein. The Design Review Board will charge a reasonable fee as determined by the Design Review Board, from time to time, to cover the administrative expense of its review and comment, such fee to be payable to the Design Review Board. The Design Review Board may require a deposit or the placement of a bond by an applicant to assure compliance with this Declaration and to cover any expenses of damages caused by construction or improvement activities required to be approved by it.

- (a) During the Development Period, the Declarant, in its sole discretion, shall appoint the members of the Design Review Board. The members of the Design Review Board during the Development Period need not be residents of the Community or own property in the Community. All members of the Design Review board appointed by the Declarant shall serve at the pleasure of the Declarant. The Design Review Board shall act on behalf of the Declarant and the Association until such time as the Declarant no longer has the right to annex property to the Community pursuant to Article 7 hereof or the Declarant no longer owns any Lot upon which no Residential Unit has been constructed, whichever is later, unless sooner waived in writing by the Declarant. From and after the later of these events, the Design Review Board shall be appointed by the Board and function in the same manner as committees of the Association under the authority of the Board.
- (b) Following the Development Period, members of the Design Review Board shall be appointed by the Board of Directors and the Design Review Board members shall be required to be Owners or their spouses. Members of the Design Review Board as are appointed by the Board shall serve at the pleasure of the Board. During the time that the Design Review Board members are appointed by the Board, the Design Review Board shall be comprised of not less than three (3) nor more than five (5) members.
- (c) The Design Review Board shall elect a chairperson and the chairperson or in his absence, the vice-chairperson, shall be the presiding officer at its meetings. The Design Review Board shall meet as often as they so determine and shall be required to meet upon call of the chairperson, and all meetings shall be held at such places as may be designated by the chairperson. A majority of the Design Review Board members serving shall constitute a quorum for the transaction of business, and the affirmative vote of a majority of those present in person at a meeting of the Design Review Board shall constitute the action of the Design Review Board on any matter before it. The Design Review Board may, from its members, appoint one (1) such member to execute approval of plans as might be proposed by an applicant. The Design Review Board shall not be required to maintain minutes of its meetings and any approval of the Design Review Board may be evidenced by the members of the Design Review Board, or the designate appointed by such members, certifying to the approval of such plans by entry of an approval on the face thereof. Approvals of the Design Review Board may occur at meetings thereof or based upon communications by and between the members thereof, without call of a meeting, but with polling of each member thereof by the chairperson or his or her designate.
- (d) The Design Review Board is authorized to retain services of consulting architects, landscape architects, urban designers, engineers, inspectors, and/or attorneys in order to advise and assist the Design Review Board in performing its functions set forth herein and such costs shall be either the sole obligation of the applicants or a Common Expense of the Association as determined in the discretion of the Board.
- (e) Any member of the Design Review Board appointed by the Declarant may be removed with or without cause by the Declarant at any time by written notice to such appointee, and a successor or successors appointed to fill such vacancy. Any member of the Design Review Board appointed by the Board may be removed with or without cause by the Board at any time by written

notice to such appointee and its successor or successors appointed to fill such vacancies shall serve at the pleasure of the Board.

(f) The Design Review Board is hereby authorized to promulgate from time to time written architectural standards, regulations, policies, procedures and guidelines (hereinafter referred to as the "Design Guidelines") governing the construction, location, landscaping, material and design of improvements, structures, the contents of submission of plans and specifications, and other information as may be required in order to evidence compliance with and obtain approval pursuant to this Article 9. The Design Review Board shall make its standards, regulations, policies, procedures and guidelines available to Owners, Builder/Owners, and developers who seek to engage in development or improvement of construction upon all or any portion of the Properties and shall conduct its operations in accordance therewith. Decisions of the Design Review Board shall take into account and be founded upon the nature, kind, shape, color, size, material and location of any construction, improvements, buildings, structures or development and the quality of workmanship planned, the design and harmony of external design and relation to surrounding structures, topography and elevation of such construction, improvements, buildings, structures and development and the Community-Wide Standard. The Design Guidelines shall be binding upon and enforceable against all Owners. The provisions hereof shall not be applicable to any of the Additional Property; provided such shall be applicable to those portions of the Additional Property annexed to this Declaration. From and after the time that the Board appoints the members of the Design Review Board, any Design Guidelines proposed by the Design Review Board and any approvals or denials by the Design Review Board shall first be approved by the Board before their being effective.

9.3 <u>Submissions to Design Review Board.</u>

- (a) No construction, improvements, buildings, structures or development of any kind whatsoever, shall be commenced, carried out on, constructed, altered, added to or maintained on any Lot or Residential Unit, other than as developed, constructed or altered by Declarant unless and until three (3) copies of the plans and specifications and related data (including, if required by the Design Guidelines, shall have been submitted to and approved in writing by the Design Review Board or if a Modifications Committee is established pursuant to the terms hereof, approved by the Design Review Board after recommendation from the Modifications Committee.
- (b) For purposes of this Article 9 and specifically Section 9.3(a), above, "construction, improvements, buildings, structures or development" shall include by way of example and not limitation the construction, installation or alteration of Residential Units, sidewalks, driveways, parking areas, mailboxes, decks, patios, courtyards, swimming pools, greenhouses, playhouses, play equipment, awnings, walls, steps, stoops, yard equipment, fences, exterior lights, garages, landscaping, hardscaping, lawns, guests or servants quarters, or other outbuildings. Plans, specifications and related data as submitted for any such construction, improvements, buildings, structures or development requested for approval shall show the nature, color, type, shape, height, materials and location of the same. One (1) copy of such plans, specifications and related data so submitted shall be retained in the records of the Design Review Board," or "approved as noted by the Design Review Board," or "disapproved by the Design

Review Board." Until the members of the Design Review Board are appointed by the Board, the Design Review Board shall have the sole discretion to determine whether the plans and specification submitted for approval are acceptable to the Design Review Board in connection with the approval rights. Any disapproval by the Design Review Board may be based upon any ground whatsoever so long as such disapproval is consistent with the objectives and purposes of this Declaration, including, but not limited to, purely aesthetic considerations, provided that such disapproval is not arbitrary or capricious.

- (c) Any and all plans submitted for the construction of a Residential Unit on a Lot, shall depict thereon the proposed Residential Unit in such detail as requested by the Design Review Board, including all driveways, parking areas, mailboxes, basketball goals, decks, patios, courtyards, swimming pools, tennis courts, greenhouses, playhouses or equipment, walls or fences, awnings, steps, stoops, yard equipment, lighting, garages, out buildings, landscaping, hardscaping, lawns, guest or servants quarters and such other structures, out buildings and items as determined by the Design Review Board or as contemplated by the applicant to be a part of the improvements on the Lot. Any and all requests for alterations or additions to a Lot or Residential Unit, including alterations or additions to existing structures or improvements or the addition of additional structures, items or improvements shall depict thereon the proposed addition or alteration in its proposed location with all height, material, location and other specifics as may be requested by the Design Review Board.
- The Design Review Board shall have the right to establish a maximum (d) percentage of a Lot which can be cleared or graded and a maximum percentage of a Lot which may be covered by Residential Units, buildings, structures or other improvements, which standard shall be promulgated on the basis of topography, percolation data, soil types and conditions, vegetation cover and other environmental factors taken into account by the Design Review Board. Following approval of any plans and specifications by the Design Review Board, representatives of the Design Review Board shall have the right, without notice, during reasonable hours to enter upon and inspect any Lot, Residential Unit or other improvements or structures with respect to which construction is underway to determine whether or not the plans and specifications thereof have been approved and are complied with. In the event that Design Review Board shall determine that such plans and specifications have not been approved or are not being complied with, the Design Review Board, acting in the name and at the expense of the Association, shall be entitled to enjoin further construction and to require the removal or correction of any work, improvement or structure in place which does not comply with the approved plans and specifications. The Design Review Board, acting in the name and at the expense of the Association, shall be entitled to set applicable fines for non-compliance of Architectural and Design Guidelines requirements and criteria.
- (e) In the event that the Design Review Board fails to approve or disapprove in writing any proposed plans and specifications within sixty (60) days after such plans and specifications have been submitted, such plans and specifications will be deemed to have been expressly approved, provided the proposed improvements are generally in harmony with the Community Wide Standard and comply with the provisions of the Design Guidelines. Upon the approval of plans and specifications by the Design Review Board, no further approval under this Article 9 shall be required with respect thereto, unless construction has not substantially commenced within one hundred eighty (180) days of approval of such plans and specifications or

unless such plans and specifications are materially altered or changed. For purposes of this section, "substantially commenced" shall mean any clearing, grading, pouring of footing or any other type of affirmative action to commence with the construction of a Residential Unit. After one hundred eighty (180) days without substantial commencement of construction, all plans and specifications must be re-submitted and additional applicable fees paid.

- (f) No landscaping, grading, excavation or filling of any nature whatsoever shall be implemented and installed on a Lot, other than by Declarant, unless and until the plans therefor have been submitted to and improved in writing by the Design Review Board. The provisions of this Article 9 regarding the time for approval of plans, the right to inspect, the right to enjoin and/or require removal, etc. shall also be applicable to any proposed landscaping, clearing, grading, excavation or filling. The Design Review Board shall be permitted to promulgate Design Guidelines with regard to any such landscaping, grading, excavation or filling as set forth in the Design Guidelines.
- (g) Before construction of any Residential Unit on any Lot begins and as part of the approved plan regarding such Residential Unit as submitted to the Design Review Board, the Design Review Board shall be provided a general landscape design which shall be reviewed and approved by it before commencement of the installation of landscape with respect to such Residential Unit. The landscaping as shown on such landscape design plan shall be installed and in place as part of the construction of the Residential Unit.

9.4 The Modifications Committee.

- (a) By way of example and not limitation of the terms of this Article 9, the Design Review Board shall have the right to approve upon submission of a request pursuant to the terms of this Article any and all modifications, additions or alterations made on or to a Lot. Notwithstanding this authority of the Design Review Board, the Design Review Board, may authorize, continue and discontinue a Modifications Committee to review any and all proposed modifications, additions or alterations to a Lot on which a Residential Unit has been completed, or affecting the exterior portion of the Residential Unit thereon.
- If such Modifications Committee is established, such Committee's authority shall be limited to review of proposed modifications, additions or alterations made on or to such Lot or affecting the exterior portion of the Residential Unit thereon, following (a) completion of the construction of the initially proposed Residential Unit and the improvements thereon as approved by the Design Review Board (b) occupancy of such Residential Unit and (c) issuance of a certificate of occupancy if such is issued by the local jurisdiction. The Modifications Committee shall not have authority to review modifications, additions or alterations made by the Declarant or by Builder/Owners, which modifications, additions or alterations shall be subject, at all times, only to review and approval of the Design Review Board.
- (ii) The results of any and all review by the Modifications Committee shall be submitted to the Design Review Board with a recommendation, in writing, by the Modifications Committee for approval, approval as noted with comments, or disapproval. The recommendation of the Modifications Committee shall not be binding until adopted by the Design Review Board, who

33

may accept, accept with comment or reject such recommendation. If any recommendation of the Modifications Committee is rejected by the Design Review Board, the Design Review Board may substitute its own response to any such proposed modification, addition or alteration, which shall then be binding.

- (iii) Review and recommendation by the Modifications Committee shall be based upon the terms of this Declaration and the Design Guidelines.
- (b) The Modifications Committee shall, if established, consist of at least three (3) and no more than five (5) members, all of whom shall be appointed by the Declarant until such time as the members of the Design Review Board are appointed by the Board of Directors. At such time that the members of the Design Review Board are appointed by the Board of Directors, the Board of Directors shall also appoint the members of the Modifications Committee, if existing. If the Design Review Board determines that it has jurisdiction over and in respect to any submission to the Modifications Committee, the Design Review Board shall be authorized to review and approve such submission to the exclusion of Modifications Committee. The Modifications Committee shall, as soon as practical after receipt by it of any submission hereunder review the same and deliver its recommendation to the Design Review Board. If the Design Review Board fails to approve or disapprove, in writing, any proposed modification, addition or alteration within sixty (60) days after such have been submitted, such plans and specifications will be deemed expressly approved, provided the proposed modification, addition or alteration is in general harmony with the Community Wide Standard. The provisions hereof shall not be applicable to the Additional Property provided such shall be applicable to those portions of the Additional Property as have been annexed to this Declaration.
- (c) Nothing contained herein shall be construed to limit the right of an Owner to remodel the interior of his or her Residential Unit or to paint the interior of his or her Residential Unit any color desired, so long as the same is not visible from outside the Residential Unit.
- 9.5 Commencement of Construction and Occupancy of Residential Units. Residential Units may not be temporarily or permanently occupied until the exteriors thereof and the landscaping of the Lot on which the Residential Unit is located have been completed and a certificate of occupancy for such Residential Unit, if issued by the local jurisdiction, has been issued. Once commenced, the construction of a Residential Unit and original improvements contemplated therewith on a Lot shall diligently be continued and shall be completed within twelve (12) months from the date of commencement. For the purposes of this Section 9.5, commencement of construction shall mean that (a) all plans for such construction have been approved by the Design Review Board; (b) a building permit has been issued for the Lot by the appropriate jurisdiction; and (c) construction of a structure has physically commenced beyond site preparation. Completion of a structure shall mean that a certificate of occupancy has been issued by the appropriate jurisdiction During the continuation of construction of any Residential Unit, structure or improvements respecting a Lot or any modifications, additions or alterations thereto, any and all contractors in respect to the construction thereof shall maintain the Lot, the Residential Unit, and the surrounding Common Area in a reasonably clean and uncluttered condition and, to the extent possible, all construction trash and debris shall be kept within refuse containers. Upon completion of construction, the Owner of the Lot shall cause such contractors to immediately remove all

equipment, tools and construction material and debris from the Lot and Residential Unit on which such construction has been completed.

- 9.6 Approval of Plans. No approval of plans and specifications and no publication of Design Guidelines pursuant to the terms of this Declaration by the Design Review Board shall be construed as representing or implying that such plans, specifications or standards will, if followed, result in properly designed improvements. Such approvals and standards shall in no event be construed as representing or guaranteeing that any Residential Unit or other improvement built in accordance therewith will be built in a good workmanlike manner. The Declarant, the Association, the Design Review Board and the Modifications Committee, if acting, shall not be responsible or liable for (i) any defects in any plans or specifications submitted, revised, or approved pursuant to the terms of this Declaration, (ii) any loss or damages to any person rising out of the approval or disapproval of any plans or specifications, (iii) any loss or damage arising from the non-compliance of such plans and specifications as with any governmental ordinances and regulations, nor (iv) any defects in construction undertaken pursuant to such plans and specifications.
- 9.7 Construction Criteria and Requirement of Compliance With Law. All Residential Units and other structures and improvements shall be constructed, modified, altered or added to in compliance with any and all applicable state, county and municipal zoning and building restrictions and any applicable regulations and restrictions as might apply to the real estate. All grading, clearing, construction of impervious surfaces, building and other construction activity performed on Lots or Residential Units that are subject to the rules, regulations, guidelines and restrictions of any regulatory authority shall be performed in accordance with such rules, regulations, guidelines and restrictions.
- 9.8 <u>Land Use and Building Type</u>. Unless otherwise set forth by Declarant in a Supplemental Declaration applicable to a Phase, the following shall be applicable to all Lots.
- (a) <u>Structures and Improvements</u>. For purposes of lending assistance to define structures and improvements as required to be approved by the Design Review Board, such structures and improvements shall include, but shall not be limited to, staking, clearing, excavation, grading and other site work; initial construction of any Residential Unit or accessory building; exterior alteration of existing improvements; installation or replacement of mailboxes; swing sets and similar sports and play equipment, garbage cans, wood piles; swimming pools; gazebos or playhouses; hot tubs; wells; satellite dishes or any other apparatus for the transmission or reception of television, radio, satellite, or other signals of any kind; hedges, walls, dog runs or fences of any kind, including invisible fences; artificial vegetation or sculpture; and planting or removal of landscaping materials. Notwithstanding the foregoing, the Declarant and the Association shall regulate size and placement of antennas, satellite dishes, or any other apparatus for the transmission or reception of television, radio, satellite or other signals of any kind only in strict compliance with all federal laws and regulations.
- (b) In addition to the foregoing activities requiring prior approval, the following items are strictly regulated, and the Design Review Board shall have the right, in its sole discretion, to prohibit or restrict these items within the Properties. Each owner must strictly comply with the terms of this Section 9.8 unless approval or waiver in writing is obtained from

35

the Design Review Board. The Design Review Board may, but is not required to, adopt specific guidelines as part of the Design Guidelines or rules and regulations which address the following items.

- (i) <u>Signs</u>. No sign of any kind shall be erected by an Owner without the prior written consent of the Design Review Board, except (1) such signs as may be required by legal proceedings; and (2) not more than one (1) professional security sign of such size deemed reasonable by the Design Review Board in its sole discretion. No signs, banners, or marketing identifications items, including but not limited to: For Sale, For Rent, For Lease, are permitted on any home site, in windows of homes or structures, on any deck or any road right of way unless placed at the direction of the Declarant. The Declarant and the Design Review Board reserve the right to adopt additional restrictions with respect to the size, content, color, lettering, design and placement of any approved signs. All signs must be professionally prepared. This provision shall not apply to entry, directional, or other signs installed by the Declarant or its duly authorized agent as may be necessary or convenient for the marketing and development of the Properties.
- (ii) Tree Removal. No trees that are more than six (6) inches in diameter at a point three (3) feet above the ground shall be removed without the prior written consent of the Design Review Board, provided, however, any trees, regardless of their diameter, that are located within ten (10) feet of a drainage area, a sidewalk, a residence, or a driveway, or any diseased or dead trees needing to be removed to promote the growth of other trees or for safety reasons may be removed without the written consent of the Design Review Board. The Design Review Board may adopt or impose requirements for, or condition approval of, tree removal upon the replacement of any tree removed. Additional clearing or pruning of tree canopies may be required on some home sites as per fire ordinance recommendations.
- (iii) <u>Lighting</u>. Exterior lighting visible from the street shall not be permitted except for: (1) approved lighting as originally installed on a Lot; (2) one (1) approved decorative post light; (3) pathway lighting; (4) street lights in conformity with an established street lighting program for the Properties; (5) seasonal decorative lights during the usual and common season; or (6) front house illumination of model homes pursuant to criteria set forth in the Design Guidelines.
- Board, detached accessory structures may be placed on a Lot to be used for a playhouse, swimming pool, tool shed, dog house, garage or other approved use. A garage may also be an attached accessory structure. Such accessory structures shall conform in exterior design and quality to the dwelling on the Lot. With the exception of a garage that is attached to a dwelling and except as may be provided otherwise by the Design Review Board, an accessory structure placed on a Lot shall be located only behind the dwelling as such dwelling fronts on the street abutting such Lot or in a location approved by the Design Review Board. All accessory structures shall be located within side and rear setback lines as may be required by the Design Review Board or by applicable zoning law.

- (v) <u>Garages</u>. Garages having courtyard, side or rear entry based upon lot type and grade are preferred. See Design Guidelines for front entry garage parameters. All garages must have doors and each garage door must be coordinated in design and color with the Residential Unit to which it is appurtenant pursuant to the Design Guidelines. Each Owner shall provide in respect to each of such Owner's Residential Units parking of at least two (2) automobiles within garages. The garage shall be constructed at the same time that the Residential Unit is constructed and occupancy of the Residential Unit shall not be authorized until the garage is complete without Design Review Board approval for a variance.
- (vi) <u>Utility Lines</u>. Overhead utility lines, including lines for cable television, are not permitted except for temporary lines as required during construction and lines installed by or at the request of Declarant.
- (vii) Minimum Dwelling Size. Each Residential Unit located on any Lot shall have established in the Design Guidelines a minimum square footage of enclosed, heated and cooled living space. Upon written request of an Owner, the Design Review Board may waive the square footage requirement if, in the Design Review Board's sole discretion, the resulting appearance of such Residential Unit will preserve and conform to the overall appearance, scheme, design, value and quality within the Properties.
- (viii) <u>Sight Distance at Intersections</u>. All property located at street intersections or driveways shall be landscaped and improved so as to permit safe sight across such areas. No fence, wall, hedge or shrub shall be placed or permitted to remain where it would cause a traffic or sight problem. The Design Guidelines may include additional sight line limitations.
- (ix) <u>Air Conditioning Facilities.</u> No window or wall type air conditioners shall be permitted to be used, erected, placed, or maintained on or in any Residential Unit, building, improvement or structure within the Properties.
- (x) <u>Utility Location</u>. The Owners shall ascertain the location of utility service lines and keep the area over the route, if underground, of said service drops free of excavations and clear of structures, trees and other obstructions; it being understood that the lighting and power company may install, maintain, repair, replace and remove said underground service drops, if any, and open the ground for any such purpose or purposes; and no payment will be due or made by any utility for such use or activity.
- (xi) <u>Walls, Fences, and Hedges</u>. No walls or fences shall be erected or maintained on any Lot nearer to the street front than the front building line as set forth and as may be shown on the Subdivision Plat unless approved, in writing, by the Design Review Board. All fences and walls shall be approved by the Design Review Board pursuant to standards adopted by it prior to installation. All walls and fences on any Lot must be no higher than as might be approved and must be of a material the same as the house (i.e. brick, stone or stucco) or as otherwise approved by the Design Review Board. No fence may be installed which will impede the natural flow of water across the Lot. All fences, walls and hedges must be approved by the Design Review Board. Ownership of any wall, fence, or hedge erected as a protective screening on a Lot shall pass with

title to the Lot. Each Owner shall be responsible for maintaining any wall, fence or hedge as may exist on his or her Lot and must not block site views for safe vehicular ingress and egress to the Lot.

- (xii) <u>Parking</u>. Each Lot shall have provided thereon adequate off street parking as determined by the Design Review Board.
- (xiii) <u>Mailboxes</u>. Only one (1) mailbox may be located on each Lot, which mailbox shall be uniform as to style and materials consistent with the Design Guidelines. Each mailbox shall be located and thereafter maintained in a location approved by the Design Review Board consistent with the requirements, if any, of the United States Postal Service.
- (xiv) <u>Driveway Construction</u>. No driveway shall be placed on any Lot nor be connected to any street or road within the Properties until the location and materials on the Lot and street access of such driveway are approved by the Design Review Board. In the event there are any concrete curbs in the Properties and such curves are chipped, cracked and/or broken on the street front side as a result of driveway installation or otherwise such shall be repaired or replaced at the expense of the Owner of the Residential Unit prior to occupancy of the Residential Unit on said Lot.
- (c) <u>Building Setback Requirements</u>. Setback requirements which govern specific home site sections as referenced in the Design Guidelines shall apply.
- 9.9 No Waiver of Future Approvals. Approval of proposals, plans and specifications, or drawings for any work done or proposed, or in connection with any other matter requiring approval, shall not be deemed to constitute a waiver of the right to withhold approval as to any similar proposals, plans and specifications, drawings, or other matters subsequently or additionally submitted for approval. Non-compliance will result in the levy of fines by the Design Review Board as set forth in the Design Guidelines.
- 9.10 <u>Variance</u>. The Design Review Board may authorize variances from compliance with any of its guidelines and procedures when circumstances such as topography, natural obstructions, hardship, or aesthetic or environmental considerations require, but only in accordance with duly adopted rules and regulations. Such variances may only be granted, however, when unique circumstances dictate and no variance shall (a) be effective unless in writing; (b) be contrary to this Declaration; or (c) prevent the Design Review Board from denying a variance in other circumstances. For purposes of this Section 9.10, the inability to obtain approval of any governmental agency, the issuance of any permit, or the terms of any financing shall not be considered a hardship warranting a variance.
- 9.11 <u>Limitation of Liability</u>. The standards and procedures established pursuant to this Article 9 are intended to provide a mechanism for maintaining and enhancing the overall aesthetics of the Properties only, and shall not create any duty on the part of the Declarant, the Association, the Board or the Design Review Board to any Person. Review and approval of any application pursuant to this Article 9 is made on the basis of aesthetic considerations only and neither the Declarant, the Association, nor the Design Review Board shall bear any responsibility for ensuring the structural integrity or soundness of approved construction or modifications, the

adequacy of soils or drainage, nor for ensuring compliance with building codes and other governmental requirements. Neither the Declarant, the Association, the Board, the Design Review Board or any committee, or member of any of the foregoing shall be held liable for any injury, damages, or loss arising out of the manner or quality of approved construction on or modifications to any Lot. In all matters, the committees and their members shall be defended and indemnified by the Association as provided in Section 4.9.

9.12 Enforcement.

- The Declarant, any member of the Design Review Board or the Board, or (a) the representatives of each shall have the right, during reasonable hours and after reasonable notice, to enter upon any Lot to inspect for the purpose of ascertaining whether any structure or improvement is in violation of this Article 9. Any structure, improvement or landscaping placed or made in violation of this Article 9 shall be deemed to be nonconforming. Upon written request from the Design Review Board, Owners shall, at their own cost and expense, remove such structure or improvement and restore the property to substantially the same condition as existed prior to the nonconforming work. Should an Owner fail to remove and restore the property as required, any authorized agent of Declarant, the Design Review Board or the Board shall have the right to enter the property pursuant to Section 11.4, remove the violation, and restore the property to substantially the same condition as previously existed. Entry for such purposes and in compliance with this Section 9.12 shall not constitute a trespass. In addition, the board may enforce the decisions of the Declarant, and the Design Review Board by any means of enforcement described in Section 4.4. All costs, together with the interest at the maximum rate then allowed by law, may be assessed against the benefited Lot and collected as a Specific Assessment.
- (b) Unless otherwise specified in writing by the Design Review Board, all approvals granted hereunder shall be deemed conditioned upon completion of all elements of the approved work and all work previously approved with respect to the same Lot, unless approval to modify any application has been obtained. If, after commencement, any Person fails to diligently pursue to completion all approved work, the Association shall be authorized, after notice to the Owner of the Lot and an opportunity to be heard in accordance with the By-Laws, to enter upon the Lot and remove or complete any incomplete work and to assess all costs incurred against the Lot and the Owner thereof as a Specific Assessment.
- (c) Neither the Design Review Board, the Association, the Declarant, nor their respective members, officers or directors shall be held liable to any Person for exercising the rights granted by this Article 9. Any contractor, subcontractor, agent, employee, or other invitee of an Owner who fails to comply with the terms and provisions of this Article 9 or the Design Guidelines may be excluded by the Design Review Board from the Properties, subject to the notice and hearing procedures contained in the By-Laws.
 - (d) In addition to the foregoing, the Association shall have the authority and standing to pursue all legal and equitable remedies available to enforce the provisions of this Article 9 and the decisions of the Design Review Board.

ARTICLE 10: USE RESTRICTIONS

- 10.1 <u>Use Restrictions</u>. This Article 10 sets out certain use restrictions which must be complied with by all Owners and occupants of any Lot. The Properties shall be used only for residential, recreational, and related purposes (which may include, without limitation, model homes, sales offices, an information center and/or a sales office for any real estate broker retained by the Declarant to assist in the sale of property described on **Exhibit A**, offices for any property manager retained by the Association, business offices for the Declarant or the Association or related parking facilities) consistent with this Declaration and any Supplemental Declaration.
- 10.2 Occupants Bound. All provisions of the Declaration, By-Laws, and of any rules and regulations, use restrictions or Design Guidelines governing the conduct of Owners and establishing sanctions against Owners shall also apply to all occupants even though occupants are not specifically mentioned.
- 10.3 <u>Leasing</u>. Lots with a completed home may be leased for residential purposes only and must be for a lease period of not less than three (3) months and all leases must be in writing. All leases shall require, without limitation, that the tenant acknowledge receipt of a copy of the Declaration, By-Laws, use restrictions, and rules and regulations of the Association. The lease shall also obligate the tenant to comply with the foregoing. The Board may require notice of any lease together with such additional information deemed necessary by the Board. The Board may also adopt rules regarding the leasing of homes. Such rules may include a limitation on the maximum number of times a home may be leased in any calendar year as well as minimum lease terms (e.g., three (3) months).

10.4 Residential Use.

- (a) Subject to the rights of Declarant as set out herein, Lots may be used only for residential purposes of a single family, and for ancillary business or home office uses so long as: (i) the existence or operation of the activity is not apparent or detectable by sight, sound, or smell from outside the Lot; (ii) the activity conforms to all zoning requirements for the Properties; (ii) the activity does not involve door-to-door solicitation of residents of the Properties; (iv) the activity does not increase traffic or include frequent deliveries within the Properties; and (v) the activity is consistent with the residential character of the Properties and does not constitute a nuisance, or a hazardous or offensive use, or threaten the security or safety of other residents of the Properties, as may be determined in the sole discretion of the Board.
- (b) Subject to the rights of Declarant set out herein, no other business, trade, or similar activity shall be conducted upon a Lot without the prior written consent of the Board. The terms "business" and "trade", as used in this provision, shall be construed to have their ordinary, generally accepted meanings and shall include, without limitation, any occupation,

work, or activity undertaken on an ongoing basis which involves the provision of goods or services to persons other than the provider's family and for which the provider receives a fee, compensation, or other form of consideration, regardless of whether: (i) such activity is engaged in full or part-time, (ii) such activity is intended to or does generate a profit, or (iii) a license is required.

- (c) The leasing of a Lot shall not be considered a business or trade within the meaning of this Section 10.4. This Section 10.4 shall not apply to any activity conducted by the Declarant with respect to its development and sale of the Properties or its use of any Lots which it owns within the Properties, including the operation of a timeshare or similar program.
- (d) No garage sale, moving sale, rummage sale, or similar activity shall be conducted upon a Lot without the prior written consent of the Board.
- 10.5 <u>Vehicles</u>. All vehicles shall be subject to such reasonable rules and regulation as the Board of Directors may adopt. In addition, the following shall apply.
- (a) Automobiles and non-commercial trucks and vans shall be parked only in the garages or in the driveways, if any, serving the Lots unless otherwise approved by the Design Review Board; provided however, the Declarant and/or the Association may designate certain on-street parking areas for visitors or guests subject to reasonable rules. No automobile or non-commercial truck or van may be left upon any portion of the Properties, except in a garage, if it is unlicensed or if it is in a condition such that it is incapable of being operated upon the public highways. Such vehicle shall be considered a nuisance and may be removed from the Properties. No motorized vehicles shall be permitted on trails, paths, or unpaved Common Area except for public safety vehicles and as specifically authorized by the Board. For purpose of the foregoing, a non-commercial truck shall be deemed to be a truck upon which there is contained no commercial lettering, phone numbers, or business advertisement on the exterior thereof and which does not contain commercial license tags.
- (b) Recreational vehicles belonging to Owners or occupants of the Lots shall be parked only in the garage serving the Lot. Guests of an Owner or occupant may park a motor home on the driveway serving the Owner's or occupant's Lot for a period not to exceed seven (7) Days each calendar year. The term "recreational vehicles", as used herein, shall include, without limitation, motor homes, boats, "jet skis" or other watercraft, trailers, other towed vehicles, motorcycles, minibikes, scooters, go-carts, golf carts, campers, buses, and camper trucks and vans. Any recreational vehicle parked or stored in violation of this provision in excess of seven (7) Days shall be considered a nuisance and may be removed from the Properties. Fees and costs for removal and storage under this provision shall be assessed against the Lot as a Specific Assessment.
- (c) Service and delivery vehicles may be parked in the Properties during daylight hours for such periods of time as are reasonably necessary to provide service or to make a delivery within the Properties.
- (d) Without prior written approval and authorization of the Board of Directors, no boats, boat trailers, campers, canoes, motorcycles, mopeds, all terrain vehicles, vehicles used

primarily for recreational purposes, vehicles primarily used for commercial purposes, abandoned vehicles, vehicles which are either dismantled, partially dismantled, inoperative, discarded or one which does not have a valid license plate attached thereto, shall be stored, allowed to remain, or repeatedly parked on the Properties subject to this Declaration, except in any area, if any, designated by the Board of Directors. No vehicles shall be parked on the Common Area other than in authorized parking areas. No vehicles shall be parked or stored on blocks or other such devices on the Common Area or other portion of the Properties visible from the Common Area. No vehicles shall be parked so as to obstruct the fire lanes or roadways within the Properties. If permitted, boat trailers, boats, campers, motorcycles, mopeds, all terrain vehicles, vehicles primarily used for recreational or commercial purposes, travel trailers or inoperative automobiles are to be stored out of view from the streets and shall not be stored in any street right-of-way or on driveways. The Association is expressly authorized to remove, by towing or other methods, at the Owner's expense, any unlawful or prohibited vehicle in violation hereof.

All vehicular traffic on the private streets and roads within the Properties (e) shall be subject to the provisions of the laws of the State of North Carolina and County concerning operation of motor vehicles on public streets. The Board is hereby authorized to promulgate, administer, and enforce reasonable Rules and Regulations governing vehicular and pedestrian traffic, including reasonable safety measures and speed limits including modifications of those in force on public streets, within the Properties. The Association shall be entitled to enforce same by establishing such enforcement procedures as it deems appropriate, including levying fines for the violation thereof. In the event of a conflict between such provisions of the laws of the State of North Carolina and County and such Rules and Regulations promulgated by the Association, the Rules and Regulations of the Association shall control unless the laws of the State of North Carolina or County are determined by the Board to be more restrictive. Only drivers licensed to operate motor vehicles by the State of North Carolina or by any other state in the United States may operate any type of motor vehicle, including golf carts, mopeds, all terrain vehicles, motorcycles, motor driven bicycles within the Properties. All vehicles of any kind and nature which are operated on the streets of the Properties shall be operated in a careful, prudent, safe, and quiet manner with due consideration for the rights of all Owners within the Properties.

10.6 <u>Use of Common Area.</u>

(a) There shall be no obstruction of the Common Area, including without limitation any recreational vehicle and boat storage area, nor shall anything be kept, parked or stored on any part of the Common Area without the prior written consent of the Association, except as specifically provided herein. With the prior written approval of the Board of Directors, and subject to any restrictions imposed by the Board, an Owner or Owners may reserve portions of the Common Area for use for a period of time as set by the Board. Any such Owner or Owners who reserve a portion of the Common Areas as provided herein shall assume all risks associated with the use of the Common Area and all liability for any damage or injury to any person or thing as a result of such use. The Association shall not be liable for any damage or injury resulting from such use unless such damage or injury is caused solely by the willful acts or gross negligence of the Association, its agents or employees.

- (b) Owners of Lots, as well as their families, tenants, guests, invitees, and pets shall refrain from any actions which deter from the enjoyment by other Owners of areas within the Properties designated as Common Area. Prohibited activities shall include without limitation, maintenance of dogs or other pets under conditions which interfere with the use of the specified areas by other Owners, playing of loud radios or musical instruments, holding of large gatherings without advance approval of the Board, and use of portable outdoor grills, cooking facilities, tents or other temporary structures, stages, vending machines or facilities, except for events approved in advance by the Board. The Board may promulgate other rules and restrictions for the use of these areas.
- Animals and Pets. No animals, livestock or poultry of any kind shall be raised, bred, or kept on any Lot nor within any Residential Unit, except that dogs, cats or other usual household pets may be kept by the respective Owners on their respective Lots and within their respective Residential Unit, provided that they are not kept, bred or maintained for any commercial purpose and do not endanger the health of or unreasonably disturb the Owner of any Lots within the Community; provided that the Board of Directors may, by adoption of Rules and Regulations, (i) prohibit from the Properties animals which are determined by the Board to be dangerous or detrimental to the health, safety or welfare of the Owners and (ii) prohibit any respective pet from travel upon or use of the Common Area unless the owner of such pet pays a user fee to the Association in an amount determined by the Board. In addition, Rules and Regulations may include but not be limited to the prohibition of animals as to size, weight or type. No more than three (3) dogs or cats may be kept on any Lot at any time. No pet enclosure shall be erected, placed or permitted to remain on any Lot subjected to this Declaration except as approved pursuant to Article 9. No dog runs shall be constructed and maintained on any Lot. In the event a pet or pets become a nuisance in the opinion of the Declarant or the Board, they shall be removed from the Properties. All animals, as are permitted herein, shall be kept and maintained in accordance with the Rules and Regulations established by the Board.
- 10.8 Nuisance. It shall be the responsibility of each Owner and occupant to prevent the development of any unclean, unhealthy, unsightly, or unkempt condition on his or her property. The Properties shall not be used, in whole or in part, for the storage of any property or thing that will cause such Lot to appear to be in an unclean or untidy condition or that will be obnoxious to the eye; nor shall any substance, thing, or material be kept that will emit foul or noxious odors or that will cause any noise or other condition that will or might disturb the peace, quiet, safety, comfort, or serenity of the occupants of surrounding property. No noxious or offensive activity shall be carried on within the Properties, nor shall anything be done tending to cause embarrassment, discomfort, annoyance, or nuisance to any Person within the Properties. There shall not be maintained any plants or animals or device or thing of any sort whose activities or existence in any way is noxious, dangerous, unsightly, unpleasant, or of a nature as may diminish or destroy the enjoyment of the Properties. Without limiting the generality of the foregoing, no speaker, horn, whistle, siren, bell, amplifier or other sound device, except such devices as may be used exclusively for security purposes or as approved by the Design Review Board, shall be located, installed or maintained upon the exterior of any Lot unless required by law. Any siren or device for security purposes shall contain a device or system which causes it to shut off automatically within a reasonable period of time. The reasonable and normal

development, construction and sales activities conducted or permitted by the Declarant shall not be considered a nuisance or a disturbance of the quiet enjoyment of any Owner or occupant.

10.9 Storage of Materials, Garbage, Dumping, Etc.

- (a) All garbage cans shall be located or screened so as to be concealed from view of neighboring streets and property. All rubbish, trash, and garbage shall be regularly removed and shall not be allowed to accumulate. There shall be no dumping of grass clippings, leaves or other debris; rubbish, trash or garbage; petroleum products, fertilizers, or other potentially hazardous or toxic substances on wetlands or in any drainage ditch or stream within the Properties, except that fertilizers may be applied to landscaping on Lots provided care is taken to minimize runoff. No lumber, metals, bulk materials, refuse, trash or other similar materials shall be kept, stored, or allowed to accumulate outside the buildings on any Lot, except as may be permitted during any period of construction of improvements to a Lot.
- (b) Each Owner shall maintain its Lot in a neat and orderly condition throughout construction of a residential dwelling or accessory structure and not allow trash and debris from its activities to be carried by the wind or otherwise scattered within the Properties. Storage of construction materials on the Lot shall be subject to such conditions, rules, and regulations as may be set forth in the Design Guidelines. Each owner shall keep roadways, easements, swales, and other portions of the properties clear of silt, construction materials and trash from its activities at all times. Trash and debris during construction shall be contained in standard size dumpsters or other appropriate receptacles and removed regularly from Lots and shall not be buried or covered on the Lot. All Lots on which construction is in progress must be inspected by the Owners or their builders prior to each weekend, and during the weekend all materials shall be neatly stacked or placed and any trash or waste materials shall be removed. In addition, Owners shall remove trash and debris from the Lot upon reasonable notice by Declarant in preparation for special events.
- 10.10 <u>Combustible Liquid</u>. Storage of gasoline, propane, heating or other fuels is prohibited, except for a reasonable amount of fuel that may be stored in containers appropriate for such purpose on each Lot for emergency purposes and operation of lawn mowers and similar tools or equipment and except as may be approved in writing by the Design Review Board. The Association shall be permitted to store fuel for operation of maintenance vehicles, generators and similar equipment.
- 10.11 <u>Guns</u>. The discharge of firearms on the Properties is prohibited. The term "firearms" includes without limitation "B-B" guns, pellet guns, and firearms of all types. The Board may impose fines, and exercise other means of enforcement as set forth in this Declaration, but shall have no obligation to exercise self-help to prevent or stop such discharge.
- 10.12 <u>Subdivision of Lot</u>. Without the Declarant's prior written consent, no Lot shall be subdivided or its boundary lines changed after a subdivision plat depicting the Lot has been approved and filed in the Public Records. Any such division, boundary line change, or replatting shall not be in violation of the applicable subdivision and zoning regulations, if any.

10.13 Drainage and Grading.

- (a) Catch basins and drainage areas are for the purpose of natural flow of water only. No improvements, obstructions or debris shall be placed in these areas. No Owner or occupant may obstruct or rechannel the drainage flows after location and installation of drainage swales, storm sewers, or storm drains.
- (b) Each Owner shall be responsible for maintaining all drainage areas located on its Lot, and for controlling the natural and man-made water flow from its Lot. Required maintenance shall include, but not be limited to, maintaining ground cover in drainage areas and removing any accumulated debris from catch basins and drainage areas. No Owner shall be entitled to overburden the drainage areas or drainage system within any portion of the Properties with excessive water flow from its Lot. Owners shall be responsible for all remedial acts necessary to cure any unreasonable drainage flows from Lots. Neither the Association nor the Declarant bears any responsibility for remedial actions to any Lot.
- (c) No Person shall alter the grading of any Lot without prior approval pursuant to Article 9 of this Declaration. The Declarant hereby reserves for itself and the Association a perpetual easement across the Properties for the purpose of altering drainage and water flow. The exercise of such an easement shall not materially diminish the value of or unreasonably interfere with the use of any Lot without the Owner's consent.
- 10.14 <u>Irrigation</u>. Wells will be allowed for irrigation purposes only. Locations must be in the rear of the lot and not within street view and must be buried or located within the main structure so as not to be visible. Landscape screening is required if a well is located outside the main structure on the lot. Owners shall not install irrigation systems which draw upon surface waters nor from any stream or other bodies of water within the Properties. However, the Declarant and the Association shall have the right to draw water from such sources for the purpose of irrigating the Area of Common Responsibility.
- 10.15 <u>Streams</u>. With the exception of any activities undertaken by Declarant, no streams which run across any Lot may be dammed, or the water therefrom impounded, diverted, or used for any purpose without the prior written consent of the Board.
- 10.16 Wetlands. All areas designated on a recorded plat as "wetlands" shall be generally left in a natural state, and any proposed alteration of the wetlands must be in accordance with any restrictions or covenants recorded against such property and be approved by all appropriate regulatory bodies. Prior to any alteration of a Lot, the Owner shall determine if any portion thereof meets the requirements for designation as a regulatory wetland. Notwithstanding anything contained in this Section 10.16, the Declarant, the Association, and the successors, assigns, affiliates and designees of each may conduct such activities as have been or may be permitted by the U.S. Army Corps of Engineers or any successor thereof responsible for the regulation of wetlands.
- 10.17 <u>Temporary Structures</u>. Subject to the Declarant's and Builder/Owner's rights reserved herein, and other than for temporary facilities as might be installed by Declarant or the

Association for purposes of administration of the Properties, no structure of a temporary character, whether trailer, basement, tent, shack, garage, barn, or other outbuilding shall be permitted, maintained or used on any Lot at any time as a residence or for any other purpose, either temporarily or permanently. No garage, servants' quarters, or other permitted accessory structure shall be erected, placed, or maintained on any Lot until construction of a main Residential Unit has commenced. Subject to Article 9, any structure on which construction has commenced must be completed within a reasonable length of time.

- 10.18 Garbage Cans, Woodpiles, etc. All garbage cans, woodpiles, and other similar items shall be located or screened so as to be concealed from view of neighboring Lots, Residential Units, streets, Common Area and the Additional. All rubbish, trash, and garbage shall be regularly removed from the Lot and shall not be allowed to accumulate thereon. Subject to Declarant's reserved rights, no Lot shall be used or maintained as a dumping ground for rubbish, trash or garbage. Trash, garbage, or other rubbish shall not be kept on any Lot, except in sanitary containers constructed of metal, plastic, or masonry materials with sanitary covers or lids. All such sanitary containers shall be kept in a clean and sanitary condition. No Lot shall be used for the open storage of any materials whatsoever, which storage is visible from the street, except that new building materials used in the construction of improvements erected upon any Lot may be placed upon such Lot at the time construction is commenced and may be maintained thereon for a reasonable time, so long as the construction progresses without undue delay, until the completion of the improvements, after which these materials shall either be removed from the Lot or stored in a suitable enclosure of the Lot. Exterior clotheslines are expressly prohibited on any Lot.
- 10.19 <u>Solar Devices</u>. No artificial or man made device which is designated or used for collection of or heating by solar energy or other similar purposes shall be placed, allowed, or maintained upon any portion of the Properties, including upon any Lot or Residential Unit unless approved in accordance with Article 9.
- 10.20 <u>Above-Ground Pools</u>. Above ground swimming pools shall be strictly prohibited from the Properties.
- 10.21 Oil and Mining Operations. No oil drilling or development operation, soil refining, quarrying, or mining operations of any kind shall be permitted upon or in any Lot, nor shall oil wells, tanks, tunnels, mineral excavations, or shafts be permitted upon or in any Lot. No derrick or other structure designed for use in boring for oil, natural gas, or minerals shall be erected, maintained, or permitted upon any Lot.

ARTICLE 11: EASEMENTS

- 11.1 <u>Reserved Easements</u>. Declarant hereby reserves the following easements and rights in respect to the Properties which reservation is in addition to the other easements reserved in this Declaration.
- (a) the perpetual, alienable and transferable easement and right reserved to the benefit of Declarant and its successors and assigns and, subject to regulation by the Declarant, to the

benefit of Builder/Owners, to enter and travel upon, over, and across the Common Area for the purpose of completion and repair of improvements within the Properties or Additional Property, including Residential Units and for all reasonable purposes to further assist and enhance the marketing and sale of the Properties, Lots or Residential Units together with the easement in and to the Common Area and Lots not conveyed to an Owner for the maintenance of signs, sales offices, construction offices, business offices, and model Residential Units, together with such other facilities as in the sole opinion of the Declarant may be reasonable required, convenient or incidental to the completion, improvement and/or marketing and sale of Lots, Residential Units or the Additional Property, so long as Declarant or any Builder/Owner owns any Lot or Residential Unit primarily for the purpose of sale or has the unexpired option to add the Additional Property or any portion thereof to the Property;

Notwithstanding the foregoing reserved easements and rights, the beneficiaries of any rights or easements hereinabove set forth for purposes of travel, entry, access, ingress or egress waive all rights of uncontrolled and unlimited travel, entry, access, ingress or egress across the Common Area and acknowledge and agree that such travel, entry, access, ingress or egress shall be limited to roads, streets, sidewalks, pathways, parking lots, walkways and trails located within or upon the Common Area.

11.2 Owner's Right to Ingress, Egress, Use and Support.

- (a) Every Owner (and their business invitees and licensees) shall have and is hereby granted the right and easement of ingress and egress over, upon, and across the Common Area necessary for access to his or her Lot and shall furthermore have and is hereby granted the right and easement to lateral support. Such rights shall be appurtenant to and pass with the title to each Lot and shall be subject to the terms of this Declaration. However, all Owners, by accepting title to Lots, waive all rights of uncontrolled and unlimited access, ingress and egress to and from each such Lot and acknowledge and agree that such access, ingress and egress shall be limited to roads, streets, sidewalks, walkways, parking lots and trails located within the Properties from time to time, provided that pedestrian and vehicular access to and from all Lots shall be provided at all times.
- (b) There is reserved unto (i) the Declarant and (ii) the Association, the right and privilege, but not the obligation to maintain gates controlling vehicular access to and from the Properties provided, however, that no such gates shall prohibit or unreasonably interfere with the use and enjoyment of the reserved rights and easements set forth in this Article 11 and, provided further, that no such gates controlling vehicular access to the Properties shall impede or otherwise interfere with (i) Declarant to market, sell and develop the Community inclusive of the Additional Property or any portions thereof or; (ii) to the extent determined by Declarant the right of Builder/Owners to market, sell, and develop Lots within the Properties, including the Additional Property.
- 11.3 <u>Easement for Additional Property</u>. There is hereby reserved to Declarant, its successors, assigns and successors-in-title to the Additional Property and all parts thereof for the benefit of and as an appurtenance to the Additional Property and as a burden upon the Properties, a perpetual, alienable and transferable right and easement for (i) pedestrian and vehicular ingress, egress and parking, across, within, and on all roads, sidewalks, trails and parking facilities, from

time to time located within the Common Area (ii) the installation, maintenance, repair, replacement, connection and use of security systems and utility facilities and distribution lines, including, without limitation, drainage systems, storm sewers, and electrical, gas, telephone, water, sewer and master television antennas and/or cable system lines, and (iii) drainage and discharge of surface water onto and across the Properties, provided that such drainage and discharge shall not unreasonably interfere with the use and enjoyment of the Lots, the Properties or any improvements from time to time located thereon.

- 11.4 <u>Easement for Association</u>. There is hereby reserved for the benefit of the Association, its officers, agents and employees, including, but not limited to, any manager employed by the Association and any employees of such manager, a general right and easement to enter upon any Lot or portion thereof in the performance of its respective duties. Except in the event of emergencies, this easement is to be exercised only during normal business hours and then, whenever practical, only upon advance notice and with permission of the Owner or occupant of the Lot directly affected thereby.
- 11.5 <u>Maintenance Easement</u>. Subject to the provisions regarding maintenance set forth herein there is hereby reserved for the benefit of the Declarant, the Association, and their respective agents, employees, successors and assigns, an alienable, transferable and perpetual right and easement to enter upon any Lot for the purpose of mowing, removing, clearing, cutting or pruning landscape covering, grass, underbrush, weeds, stumps, or other unsightly growth and removing trash, so as to maintain the Community Wide Standard and reasonable standards of health, fire safety, and aesthetic appearance within the Properties; provided that such easement shall not impose any duty or obligation upon the Declarant or the Association to perform any such action. Exercise of this easement shall not constitute a trespass.
- 11.6 Alterations in Boundaries, Additions to Common Area. There is hereby reserved in Declarant the right to alter, modify and realign the boundaries of the Common Area and any Lots owned by Declarant, including the realignment of boundaries between adjacent Lots owned by Declarant and the alteration, modification and realignment of any and all rights of ways for ingress, egress and regress. Any such alteration may be shown on an amendment to the Subdivision Plat adopted and recorded by Declarant.
- 11.7 <u>Easement of Encroachment</u>. If any portion of the improvements constructed on the Common Area encroaches upon a Lot or any improvement constructed on a Lot unintentionally encroaches upon the Common Area as a result of construction, reconstruction, repair, shifting, settlement, or movement of any portion of the improvements, a valid easement for the encroachment and for the maintenance of the same shall exist so long as the encroachment exists; provided, however, if any improvement on a Lot or Common Area is knowingly and willfully constructed, reconstructed, or repaired so as to encroach, respectively, on the Common Area or a Lot, no such easement shall exist.
- 11.8 <u>Use of Common Area</u>. Other than for the right of ingress and egress and the normal intended use as interpreted by the Declarant until expiration of the Development Period and thereafter as interpreted by the Association, the Owners are hereby prohibited and restricted from using any of the Common Area except as may be allowed by the Board or as may be expressly

48

permitted in this Declaration or any amendment or Supplemental Declaration applicable to all or a portion of the Properties. By way of explanation and not limitation, no planting or gardening shall be done upon the Common Area without prior approval of the Board, and no fences, hedges, walls or structures shall be erected or maintained upon the Common Area, except as are installed or modified by Declarant or as approved by the Board or their designated representatives. It is expressly acknowledged and agreed by all parties concerned that this Section 11.8 is for the mutual benefit of all Owners and is necessary for the protection of said Owners.

- 11.9 <u>Acknowledgment of Rights of Use</u>. Each Owner, and each Member of the Association, by acceptance of a deed or contract for deed to any Lot or Residential Unit is deemed to accept the reservations, rights of use, licenses, easements, and permits existing in, through, and over the Common Area.
- 11.10 <u>Conveyance of Common Area</u>. The Association covenants to accept title to all or portions of the Common Area when conveyed by the Declarant.
- 11.11 <u>Construction and Sale Period</u>. Despite any provisions contained in this Declaration to the contrary, including the Use Restrictions of Article 10, herein, it shall be expressly permissible for Declarant and its assigns (which may include Builder/Owners as approved by Declarant) to maintain and carry on upon such portion of the Properties as the Declarant may deem necessary, including, but not limited to, the Common Area, such facilities and activities as in the sole opinion of Declarant may be reasonably required, convenient, or incidental to construction upon or sale and marketing of any of the Properties or Additional Property, including, without limitation, business offices, signs, model homes, and sales offices, so long as construction on or offering for sale by Declarant or a Builder/Owner of all or any portion of the Properties or Additional Property, including Lots continues. The right to maintain and carry on such facilities and activities shall include specifically the right to use Lots and Residential Units owned by Declarant or Builder/Owners as models and sales offices and to authorize sales and construction personnel to travel upon and enter the Common Area. Furthermore, Declarant reserves the right, during installation of streets or other facilities, as shown on any Subdivision Plat or plat of any Phase, to enter onto any Lot or Lots for the purpose of disposing of excavation, if necessary, provided that such Lot or Lots have not been conveyed to and/or contracted for or by any other Owner or Owners.
- 11.12 Easements for Utilities, etc. Unless otherwise noted on any recorded plat, easements five (5) feet in width are reserved on either side of all front, side and rear lot lines for the installation, maintenance and repair of any utility services or drainage facilities, including but not limited to, water, sewer, telephone, gas, cable television, electricity and drainage ditches or swales. In addition to the above referenced reserved easements, there is hereby reserved to the Declarant for as long as the Development Period continues and thereafter to the Association, a blanket easement and the power to grant blanket easements, upon, across, over, and under all of the Properties, including Lots, for ingress, egress, installation, replacing, repairing, and maintaining master television antenna or cable systems, security, and similar systems, walkways, and all utilities, including, but not limited to, water, sewers, telephones, gas, cable television, and electricity; provided this easement shall not authorize entry into or cause physical damage to any structure as might exist unless owned by Declarant if the Declarant is so acting or the Association if the Association is so acting. In furtherance of this easement, from and after such time as the

Association is empowered as referred to herein, the Board shall, upon written request of Declarant, grant such easements as may be reasonably necessary for the development of any property described in Exhibit "A" or the Additional Property. In addition, Declarant reserves the easements and rights-of-way as shown on any Subdivision Plat or the plat of any Phase or any of the Properties, including Lots, for the purpose of constructing, maintaining, and repairing a system or systems of electric lighting, electric power, telegraph and telephone line or lines, gas, sewers, or any other utility Declarant determines to install in, across, and/or under the Properties; provided, however, Declarant reserves the right to relocate, make changes in and additions to the above easements for the purpose of most efficiently and economically installing the improvements. To the extent possible, all utility lines and facilities serving the Properties and located therein shall be located underground. By virtue of any such easement and facilities, it shall be expressly permissible for the providing utility company or other supplier or servicer, with respect to the portions of the Properties so encumbered, (i) to erect and maintain pipes, lines, manholes, pumps and other necessary equipment and facilities, (ii) to cut and remove any trees, bushes or shrubbery, (iii) to grade, excavate, or fill, or (iv) to take any other similar action reasonably necessary to provide economical and safe installation, maintenance, repair, and use of such utilities and systems; provided, however, that such utility company or other supplier or servicer shall take reasonable actions to repair any damages caused by such utility company or other supplier or servicer during the exercise of any rights conveyed under any easement granted hereunder.

- 11.13 Easement for Law Enforcement and Fire Protection. Declarant hereby grants to the County or such other governmental authority or agency as shall from time to time have jurisdiction over the Properties with respect to law enforcement and fire protection, the perpetual, alienable and transferable right and easement upon, over and across all of the Common Area for purposes of performing such duties and activities related to law enforcement and fire protection in these Properties as shall be required or appropriate from time to time by such governmental authorities under applicable law.
- 11.14 <u>Assignment of Declarant Rights</u>. Declarant may assign its rights as Declarant and by such assignment create the status of each such assignee as Declarant to all or any portion of the Properties or Additional Property to any party or parties who take title to all or any portion of the Properties or Additional Property for the purpose of development and sale. Declarant, however, unless otherwise specifically assigned, shall, so long as it owns any interest in the Community, retain its Declarant status to such portion of the Properties retained by it and retain all Class "B" votes despite any such transfer or assignment.
- 11.15 Easement for Walks, Trails, Signs and Perimeter Walls. It is hereby reserved for the benefit of Declarant, the Association, and their respective successors and assigns, the alienable, transferable and perpetual right and easement upon, over, and across (i) those strips of land ten (10) feet in width located along and adjacent to the exterior boundaries of all Lots, such strips to be bounded by the exterior boundaries adjacent to streets and roads and by lines in the interior of such Lots which are ten (10) feet from and parallel to such exterior boundaries, for the installation, maintenance, and use of sidewalks, traffic directional signs, and related improvements, provided that Declarant shall have no obligation to construct any such improvements. There is further reserved for the benefit of the Declarant, the Association, and their respective successor and assigns, the alienable, transferable and perpetual right and easement upon, over, and across those strips of

- land fifteen (15) feet in width located along those boundaries of all Lots that constitute part of the perimeter boundary of the Properties, such easement to be for the purpose of constructing, installing, replacing, repairing and maintaining a greenway walking trail, perimeter wall or fence around the perimeter boundary of the Properties, provided that Declarant nor the Association shall have any obligation to construct any such greenway walking trail, perimeter wall or fence.
- 11.16 <u>Easements for Lateral Support</u>. Declarant reserves, creates, establishes, promulgates and declares non-exclusive, perpetual, reciprocal, appurtenant easements over every portion of the Common Area, every Lot, and any improvement which contributes to the lateral support of another portion of the Common Area or of another Lot shall be burdened with an easement for lateral support, and each shall also have the right to lateral support which shall be appurtenant to and pass with title to such property.
- 11.17 <u>Easement for Special Events.</u> Declarant reserves, creates, establishes, promulgates and declares for itself, its successors, assigns and designees a perpetual, non-exclusive reciprocal, appurtenant easement over the Common Area for the purpose of conducting educational, cultural, entertainment, or sporting events, and other activities of general community interest at such locations and times as Declarant, in its sole discretion, deems appropriate. Each Owner, by accepting a deed or other instrument conveying any interest in a Lot, acknowledges and agrees that the exercise of this easement may result in a temporary increase in traffic, noise, gathering of crowds, and related inconveniences, and each Owner agrees on behalf of itself and the occupants of its Lot to take no action, legal or otherwise, which would interfere with the exercise of such easement or to recover damages for or as the result of any such activities.
- 11.18 <u>Easement for Park and Walking Trail Access</u>. Declarant hereby grants to the Owners a perpetual, non-exclusive easement over and across any areas designated as parks, recreation areas, walking trails or paths on any recorded Subdivision Plat of the Properties. Use of such facilities shall be governed by reasonable rules and regulation promulgated by the Association.
- 11.19 <u>Liability for Use of Easements</u>. No Owner shall have a claim or cause of action against the Declarant, its successors or assigns, arising out of the exercise of any easement reserved hereunder or shown on any Subdivision Plat for the Properties, except in cases of willful or wanton misconduct.
- 11.20 Notice of Electric Transmission Line Easement. Each Owner acknowledges the existence of an electric power transmission line easement having a width of one hundred (100) feet which traverses the Community from Old Lakey Gap Road in a southerly direction as shown on various recorded plats of the properties. There are Lots within the Community upon which a portion of said Lots are encumbered by and subject to this power line easement. No structure(s) may be constructed and maintained upon that portion of a Lot which is subject to said power line easement.

ARTICLE 12: MORTGAGEE'S RIGHTS

The following provisions are for the benefit of holders, insurers, or guarantors of First Mortgages in the Community. To the extent applicable, necessary, or proper, the provisions of this Article apply to both this Declaration and to the By-Laws of the Association. Where indicated, these provisions apply only to "eligible holders," as hereinafter defined.

- 12.1 <u>Notice of Action</u>. So long as required by the Federal National Mortgage Association but only provided that the Community is a planned development approved by or seeking approval by such Association (and such is approved by the Declarant, in writing) an institutional holder, insurer, or guarantor or a First Mortgagee who provides a written request to the Association (such request to state the name and address of such holder, insurer, or guarantor and the unit number), (therefore becoming an "eligible holder"); will be entitled to timely written notice of:
 - (a) any proposed termination of the development;
- (b) any condemnation loss or any casualty loss which affects a material portion of the Properties or which affects any Lot or Residential Unit on which there is a first mortgage held, insured, or guaranteed by such eligible holder;
- (c) any delinquency in the payment of assessments or charges owed by an Owner of a Lot or Residential Unit subject to the mortgage of such eligible holder, insurer, or guarantor, where such delinquency has continued for a period of sixty (60) days;
- (d) any lapse, cancellation, or material modification of any insurance policy maintained by the Association; or
- (e) any proposed action which would require the consent of eligible holders, as required in Sections 2 and 3 of this Article.

12.2 Other Provisions For First Lien Holders.

- (a) Any restoration or repair of the Common Area after a partial condemnation or damage due to an insurable hazard shall be substantially in accordance with this Declaration and the original plans and specifications unless approval is obtained of the eligible holders of First Mortgages to which at least fifty-one (51%) percent of the votes of Lots and Residential Units, subject to mortgages held by such eligible holders are allocated.
- (b) Any election to terminate the development after substantial destruction or a substantial taking in condemnation must require the approval of the eligible holders of First Mortgages to which at least fifty-one (51%) percent of the votes of Lots and Residential Units, subject to Mortgages held by such eligible holders, are allocated.

- 12.3 <u>Amendments to Documents</u>. The following provisions do not apply to amendments to the constituent documents or termination of the development made as a result of destruction, damage, or condemnation pursuant to (a) and (b) above, or to the addition of land in accordance with Article 7 which might occur pursuant to any plan of expansion or phased development previously approved by the agencies and corporations, to the extent such approval was required under the applicable programs of the agencies and corporations.
- (a) The consent of at least sixty-seven (67%) percent of the Class "A" votes and of the Declarant, so long as it owns any land subject to this Declaration, and the approval of the eligible holders of First Mortgages to which at least sixty-seven (67%) percent of the votes of such subject to a mortgage appertain, shall be required to terminate the development.
- (b) So long as required by the Federal National Mortgage Association but only provided that the Community is a planned development approved or seeking approval by the Federal National Mortgage Association (and such is approved by the Declarant, in writing) the consent of at least sixty-seven (67%) percent of the Class "A" votes and of the Declarant, so long as it owns any land subject to this Declaration, and the approval of eligible holders of First Mortgages to which at least fifty-one (51%) of the votes of Residential Units and Lots subject to a mortgage appertain, shall be required materially to amend any provisions of the Declaration, By-Laws, or Articles of Incorporation of the Association, or to add any material provisions thereto, which establish, provide for, govern, or regulate any of the following:
 - (i) voting;
 - (ii) assessments, assessment liens, or subordination of such liens;
 - (iii) reserves for maintenance, repair, and replacement of the Common Area;
 - (iv) insurance or Fidelity Bonds;
 - (v) rights to use of the Common Area, subject to the allowances herein contemplated;
 - (vi) responsibility for maintenance and repair of the Properties;
 - (vii) other than as in this Declaration provided, expansion or contraction of the Properties, or the addition, annexation, or withdrawal of property to or from the regime;
 - (viii) other than as in this Declaration provided or contemplated, boundaries of any Lot;
 - (ix) leasing of Residential Units;

- (x) imposition of any right of first refusal or similar restriction on the right of any Owner to sell, transfer, or otherwise convey his or her Lot;
- (xi) establishment of self-management by the Association where professional management has been required by any of the agencies or corporations; or
- (xii) any provision included in the Declaration, By-Laws, or Articles of Incorporation which are for the express benefit of holders, guarantors, or insurers of first mortgages on Lots and Residential Units.
- 12.4 <u>Special FHLMC Provision</u>. So long as required by the Federal Home Loan Mortgage Corporation (The Mortgage Corporation), but only provided that the Community is a planned development approved by or seeking approval by The Mortgage Corporation (and such is approved by the Declarant, in writing), the following provisions apply in addition to and not in lieu of the foregoing. Unless two-thirds (2/3) of the First Mortgagees or Owners give their consent, the Association shall not:
- (a) by act or omission seek to abandon, partition, subdivide, encumber, sell, or transfer the Common Area which the Association owns, directly or indirectly (the granting of easements for public utilities or for other public purposes consistent with the intended use of the Common Area shall not be deemed a transfer within the meaning of this clause);
- (b) change the method of determining the obligations, assessments, dues or other charges which may be levied against an Owner;
- (c) by act or omission change, waive, or abandon any scheme of regulations or enforcement thereof pertaining to the architectural design or the exterior appearance and maintenance of Lots or Residential Units and of the Common Area;
- (d) fail to maintain fire and extended coverage insurance, as required by this Declaration; or
- (e) use hazard insurance proceeds for any Common Area losses for other than the repair, replacement, or reconstruction of such property.
- 12.5 <u>Payment of Taxes</u>. First Mortgagees may, jointly or individually, pay taxes or other charges which are in default and which may or have become a charge against any Common Area and may pay overdue premiums on hazard insurance policies or secure new hazard insurance coverage on the lapse of a policy for such Common Area. First Mortgagees making such payments shall be owed immediate reimbursement therefor from the Association.
- 12.6 <u>No Priority</u>. No provision of this Declaration or the By-Laws gives or shall be construed as giving any Owner or any other party priority over any rights of the First Mortgagee in

the case of a distribution to such Owner of insurance proceeds or condemnation awards for losses to or a taking of Common Area.

12.7 <u>Notice of Default</u>. Notwithstanding anything contained herein which might otherwise be construed to the contrary, a First Mortgagee, upon request, will be entitled to written notification from the Association of any default in the performance by an Owner of a Lot or Residential Unit in which such Mortgagee has an interest of any obligation under this Declaration, the By-Laws, or the Articles of Incorporation which is not cured within sixty (60) days.

ARTICLE 13: DECLARANT'S RIGHTS

- 13.1 <u>Transfer or Assignment</u>. Any or all of the special rights and obligations of the Declarant set forth in the Governing Documents may be transferred or assigned in whole or in part to the Association or to other Persons, provided that the transfer shall not reduce an obligation nor enlarge a right beyond that which the Declarant has under this Declaration or the By-Laws. Upon any such transfer, the Declarant shall be automatically released from any and all liability arising with respect to such transferred rights and obligations. No such transfer or assignment shall be effective unless it is in a written instrument signed by the Declarant and duly recorded in the Public Records.
- Development and Sales. The Declarant may maintain and carry on upon the Properties such facilities and activities as, in the sole opinion of the Declarant, may be reasonably required, convenient, or incidental to the development of the Properties and/or the construction or sale of Lots, such as sales activities, tournaments, charitable events, and promotional events, and restrict Members from using the Common Area during such activities. Such activities shall be conducted in a manner to minimize (to the extent reasonably possible) any substantial interference with the Members' use and enjoyment of the Common Area. In the event that any such activity necessitates exclusion of Owners from Common Areas, such activities shall not exceed seven (7) consecutive Days. The Declarant has easements over the Properties for access, ingress and conducting such activities. In addition, the Declarant may establish within the Properties, including any clubhouse, such facilities as, in the sole opinion of the Declarant, may be reasonably required, convenient, or incidental to the development of the Properties and/or the construction or sale of the Lots, including, but not limited to, business offices, signs, model Lots, tents, sales offices, sales centers and related parking facilities. During the Development Period, Owners may be excluded from use of all or a portion of such facilities in the Declarant's sole discretion. The Declarant shall have easements over the Properties for access, ingress, and egress and use of such facilities. During the Development Period, Declarant may permit the use of any facilities situated on the Common Area by Persons other than Owners without the payment of any use fees.
- 13.3 <u>Improvements to Properties</u>. The Declarant and its employees, agents and designees shall have a right and easement over and upon all of the Common Area for the purpose of making, constructing and installing such improvements to the Common Area as it deems appropriate in its sole discretion. The Declarant and its employees, agents and designees shall

also have a right and easement over and upon each and every Lot, the boundary line or lines of which form a portion of the perimeter of the Properties, if deemed appropriate by the Declarant, in its sole discretion.

- 13.4 <u>Additional Covenants</u>. No Person shall record any declaration of covenants, conditions and restrictions, declaration of condominium, easements, or similar instrument affecting any portion of the Properties without Declarant's review and written consent. Any attempted recordation without such consent shall result in such instrument being void and of no force and effect unless subsequently approved by written consent signed by the Declarant and recorded in the Public Records. No such instrument recorded by any Person, other than the Declarant pursuant to Section 13.6, may conflict with the Declaration, By-Laws or Articles.
- 13.5 <u>Amendments</u>. Notwithstanding any contrary provision of this Declaration, no amendment to or modification of any use restrictions and rules or Design Guidelines made after termination of the Class "B" membership shall be effective during the Development Period without prior notice to and the written consent of the Declarant. This Article 13 may not be amended without the written consent of the Declarant. With the exception of the rights described in Section 13.7, the rights contained in this Article 13 shall terminate upon the earlier of (a) twenty (20) years from the date this Declaration is recorded, or (b) upon recording by Declarant of a written statement that all sales activity has ceased.
- 13.6 Additional Covenants and Easements by Declarant. The Declarant may unilaterally subject any portion of the Properties to additional covenants and easements, including covenants obligating the Association to maintain and insure such property on behalf of the Owners and obligating such Owners to pay the costs incurred by the Association through General Assessments, Parcel Assessments or Specific Assessments. Such additional covenants and easements shall be set forth in a Supplemental Declaration filed either concurrently with or after the annexation of the subject property, and shall require the written consent of the owner(s) of such property, if other than the Declarant. Any such Supplemental Declaration may supplement, create exceptions to, or otherwise modify the terms of this Declaration as it applies to the subject property in order to reflect the different character and intended use of such property.
- 13.7 <u>Water System</u>. It is anticipated that all Lots will be serviced by public water services ("Water System"). Declarant has petitioned the Town of Black Mountain for annexation into the Town corporate boundaries and upon acceptance, water service shall be provided by The Town of Black Mountain. Owners shall be obligated to connect to the water system and shall be obligated to pay any impact fee, connection fee or other service fee assessed by The Town of Black Mountain to connect to the Water System. Declarant shall have the right to charge Owners and the Association for any water system connection fees, impact fees or other fees related to the Water System charged to the Declarant by The Town of Black Mountain. The Declarant and the Association shall have a blanket easement over the Community, including Lots, for the purpose of installing, operating, maintaining and repairing the Water System. The Declarant shall have the right to sell, lease or otherwise convey or transfer the Water System to The Town of Black Mountain or any other third party, including an affiliated company. Under no circumstances shall the exercise of any of the rights pursuant to this Section 13.7 require the consent or

approval of the Association or the Owners. No individual water well shall be permitted on a Lot unless said well is approved by Declarant and is approved as installed by Declarant and/or any other required governmental authority.

- 13.8 <u>Annexation by Town of Black Mountain</u>. Declarant reserves the right, without joinder of any Lot Owner or the Association, to petition the Town of Black Mountain requesting the annexation of the subdivision or any part thereof to the Town of Black Mountain and the right to petition on behalf of any lot Owner or the Association in any such voluntary annexation proceeding.
- 13.9 <u>Reservation of Rights under the Act</u>. Declarant specifically reserves those Special Declarant Rights as defined in Chapter 47F-1-103 (28) of the North Carolina Planned Community Act.

ARTICLE 14: RIGHT OF FIRST REFUSAL

- 14.1 Applicability. Except for sales and conveyances by the Declarant, no unimproved lot may be sold by any Owner except in compliance with the provisions of this Article 14. For the purposes of this Article 14, a Lot shall be considered unimproved unless and until any proposed improvements to such Lot have been approved by the Design Review Board pursuant to the provisions of Article 9 herein and the good faith commencement of the construction of such approved improvements (i.e. at a minimum, completion of footings and foundation of the approved residence and bona fide evidence of total expenditures for improvements to the lot of at least \$50,000.00) shall have occurred.
- Right of First Refusal. Before any unimproved lot (or any ownership interest therein) may be sold to any Person other than Declarant or its successors, the Owner of such lot shall first offer in writing to sell the Lot to the Declarant or its successors at a price equal to that agreed upon pursuant to a bona fide written offer to purchase the lot. Upon receipt by an Owner of a bona fide written offer to purchase an unimproved lot, such Owner shall send to Declarant a copy of such bona fide offer along with written notification that such Owner is offering the Lot for sale to Declarant pursuant to this right of first refusal at the same price as set forth in said offer. If the Declarant or its successor does not accept or reject in writing said offer of sale within thirty (30) days from the date of receipt of the same, then the Owner of such Lot shall have the right to sell the Lot to the third party making such bona fide offer pursuant to such bona fide offer, without any further additional obligation to offer the Lot to Declarant. Declarant shall have this right of first refusal with regard to each bona fide offer which an Owner receives for the purchase of an unimproved Lot. Any Owner who buys an unimproved lot from another Owner shall be governed by the provisions of this Article 14 and the waiver of the right of first refusal with respect to any sale shall not limit Declarant's rights of first refusal with respect to any subsequent sale of any unimproved Lot. Provided, however, that the right of first refusal reserved to the Declarant pursuant to this Article 14 shall be valid and enforceable with respect to any unimproved Lot only for a period of ten (10) years from the date of the first conveyance of such Lot from Declarant to an Owner other than the Declarant, and upon the expiration of said ten

- (10) year period, the Owner of such Lot shall have the right to sell the unimproved Lot to any third party without the obligation to offer the Lot to Declarant. Further provided that this Article 14 shall not be applicable with respect to any foreclosure sale of a deed of trust on an unimproved lot or deed in lieu of foreclosure which is made and delivered in good faith. In each instance where an offer to purchase an unimproved lot is presented to Declarant by an Owner pursuant to the right of first refusal granted herein, Declarant shall determine in its sole discretion and on a case by case basis whether to exercise its right of first refusal, and such determination may be made on such basis and for such reason as Declarant in its sole discretion shall choose. Should an Owner fail to comply with the provisions of this Article 14 and sell an unimproved Lot without first offering said Lot to Declarant in accordance with the terms hereof, then the purchaser of such unimproved Lot shall purchase such Lot subject to the right of first refusal herein granted, and Declarant shall thereafter at any time have the right to purchase such Lot, whether or not it is subsequently improved, from the purchaser thereof at the price as set forth in the bona fide offer to purchase upon which said sale was based and shall also be entitled to any other rights and remedies available at law or in equity for the violation of this Article 14.
- 14.3 <u>Death of an Owner; Gift.</u> The personal representative, heirs, successors and assigns of any Owner who dies while owning any unimproved Lot, or the donee of a gift of a Lot from an Owner, shall become an Owner subject to the terms and conditions of this Declaration and any subsequent sale, transfer and conveyance of such Lot shall be governed by the provisions of this Article 14.
- 14.4 <u>Transfers to Declarant.</u> In the event that Declarant exercises its right of first refusal pursuant to this Article 14, the closing of the conveyance of such Lot shall occur within sixty (60) days after receipt by the Owner of written notice from Declarant or its successors that it elects to exercise its right of first refusal with respect to such Lot. At the closing, Declarant shall make payment to such Owner of the Purchase price as described in the bona fide written offer, in cash or cash equivalent. The Owner shall deliverto Declarant a general warranty deed conveying fee simple marketable title to the Lot free and clear of all exceptions except those that existed at the time of the acquisition of the Lot by such Owner, the lien of ad valorem taxes for the current year and any other exceptions that may be approved by Declarant. In the event the closing occurs after the death of an Owner, Declarant may, in its discretion, require the personal representative of the Onwer to post such bonds or other assurances as the Declarant may deem reasonable in order to protect Declarant from any loss which might be caused by the failure to pay any federal or state inheritance tax or the failure to pay the claims of any creditors who may have alien on the Lot superior to Declarant's rights as a purchaser of said Lot.
- 14.5 No Further Documentation Required. The right of first refusal reserved by the Declarant in this Article 14 shall run with the title to each Lot in the Community which is subject to (or hereinafter subjected to) this Declaration, and be binding upon each purchaser of a Lot from Declarant or its successors and upon any subsequent Owner, whether such Owner purchased such Lot from Declarant or from a third party. The provisions of this Article 14 shall constitute record notice to all purchasers of Lots in the Subdivision subject to (or hereinafter subjected to) these Restrictions of the right of first refusal herein reserved, and no additional language in any deed of conveyance of a Lot and no recording of additional instruments shall be

58

required to make all Owners of such Lots in the Community subject to the provisions of this Article 14.

ARTICLE 15: GENERAL PROVISION

15.1 Duration.

- (a) Unless terminated as provided in this Section 14.1, or unless otherwise limited by North Carolina law, this Declaration shall have perpetual duration. If North Carolina law limits the period during which covenants may run with the land, then to the extent consistent with such law, this Declaration shall automatically be extended at the expiration of such period for successive periods of twenty (20) years each.
- (b) Pursuant to §47F-2-118 of the North Carolina Planned Community Act the planned community may be terminated by agreement of Lot Owners to which at least eighty percent (80%) of the votes in the Association are allocated. The agreement shall be drafted, executed or ratified, and recorded pursuant to the terms of §47F-2-118 of the North Carolina Planned Community Act.
- (c) Nothing in this Section 15.1 shall be construed to permit termination of any easement created in this Declaration without the consent of the holder of such easement.

15.2 Amendment.

- (a) <u>By Declarant</u>. Until termination of the Class "B" membership, Declarant may unilaterally amend this Declaration for any purpose. Thereafter, the Declarant, during the Development Period, may unilaterally amend this Declaration at any time and from time to time if such amendment is necessary (i) to bring any provision into compliance with any applicable governmental statute, rule, regulation, or judicial determination; (ii) to enable any reputable title insurance company to issue title insurance coverage on the Lots; (iii) to enable any institutional or governmental lender, purchaser, insurer or guarantor of Mortgage loans, including, for example, the Department of Veteran Affairs, the Federal Housing Administration, the Federal National Mortgage Association or Federal Home Loan Mortgage Corporation, to make, purchase, insure or guarantee Mortgage loans on the Lots; or (iv) to satisfy the requirements of any local, state or federal governmental agency. However, any such amendment shall not adversely affect the title to any Lot unless the Owner shall consent in writing. In addition, during the Development Period, Declarant may unilaterally amend this Declaration for any other purpose, provided the amendment has no material adverse effect upon any right of any Owner.
- (b) By the Board. The Board shall be authorized to amend this Declaration without the consent of the Members to correct scriveners' errors and other mistakes of fact, provided that amendments under this provision have no material adverse effect on the rights of the Owners. During the Development Period, any such amendment shall require the written consent of the Declarant.

- (c) <u>By Members</u>. Except as otherwise specifically provided above and elsewhere in this Declaration, this Declaration may be amended only by the affirmative vote or written consent, or any combination thereof, of Members holding sixty-seven percent (67%) of the total Class "A" votes in the Association, including sixty-seven percent (67%) of the Class "A" votes held by Members other than the Declarant, and, during the Development Period, the written consent of the Declarant. Notwithstanding the above, the percentage of votes necessary to amend a specific clause shall not be less than the prescribed percentage of affirmative votes required for action to be taken under that clause.
- (d) <u>Validity and Effective Date</u>. Any amendment to the Declaration shall become effective upon recording in the Public Records, unless a later effective date is specified in the amendment. No challenge to the validity of an amendment adopted pursuant to this Section 15.2 may be brought more than one year after recordation such amendment. In no event shall a change of conditions or circumstances operate to amend any provisions of this Declaration. No amendment may remove, revoke, or modify any right or privilege of the Declarant or the Class "B" Member without the written consent of the Declarant, the Class "B" Member, or the assignee of such right or privilege.

If an Owner consents to any amendment to this Declaration or the By-Laws, it will be conclusively presumed that such Owner has the authority to consent, and no contrary provision in any Mortgage or contract between the Owner and a third party will affect the validity of such amendment.

- 15.3 <u>Severability</u>. Invalidation of any provision of this Declaration, in whole or in part, or any application of a provision of this Declaration by judgment or court order shall in no way affect other provisions or applications.
- 15.4 <u>Dispute Resolution</u>. It is the intent of the Association and the Declarant to encourage the amicable resolution of disputes involving the Properties and to avoid the emotional and financial costs of litigation if at all possible. Accordingly, the Association, the Declarant and each Owner covenants and agrees that it shall attempt to resolve all claims, grievances or disputes involving the Properties, including, without limitation, claims, grievances or disputes arising out of or relating to the interpretation, application or enforcement of the Governing Documents through alternative dispute resolution methods, such as mediation and arbitration. To foster the amicable resolution of disputes, the Board may adopt alternative dispute resolution procedures. Participation in alternative dispute resolution procedures shall be voluntary and confidential. Should either party conclude that such discussions have become unproductive or unwarranted, then the parties may proceed with litigation.
- 15.5 <u>Litigation</u>. Except as provided below, no judicial or administrative proceeding shall be commenced or prosecuted by the Association unless approved by a vote of members holding seventy-five percent (75%) of the total Association vote. This Section 14.5 shall not apply, however, to (a) actions brought by the Association to enforce the provisions of the Governing Documents (including, without limitation, the restrictive covenants, Design Guidelines and Foreclosure of liens); (b) the imposition and collection of assessments as provided in Article 8; (c) proceedings involving challenges to <u>ad valorem</u> taxation; (d) counter-

claims brought by the Association in proceedings instituted against it or (e) actions brought by the Association against any contractor, vendor, or supplier of goods or services arising out of a contract for services or supplies. This Section 14.5 shall not be amended unless such amendment is approved by the percentage of votes, and pursuant to the same procedures, necessary to institute proceedings as provided above.

- 15.6 <u>Non-Merger</u>. Notwithstanding the fact that Declarant is the current owner of the Properties, it is the express intention of Declarant that the easements established in the Declaration for the benefit of the Properties and Owners shall not merge into the fee simple estate of individual lots conveyed by Declarant or its successors, but that the estates of the Declarant and individual Lot Owners shall remain as separate and distinct estates. Any conveyance of all or a portion of the Properties shall be subject to the terms and provisions of this Declaration, regardless of whether the instrument of conveyance refers to this Declaration.
- 15.7 <u>Grants</u>. The parties hereby declare that this Declaration, and the easements created herein shall be and constitute covenants running with the fee simple estate of the Properties. The grants of easements in this Declaration are independent of any covenants and contractual agreements undertaken by the parties in this Declaration and a breach by either party of any such covenants or contractual agreements shall not cause or result in a forfeiture or reversion of the easements granted in this Declaration.
- 15.8 <u>Cumulative Effect; Conflict.</u> The provisions of this Declaration shall be cumulative with any additional covenants, restrictions, and declarations, and the Association may but shall not be required to, enforce such additional covenants, conditions, and provisions; provided however, in the event of a conflict between or among this Declaration and such covenants or restrictions, and/or the provisions of any articles of incorporation, by-laws, rules and regulations, policies, or practices adopted or carried out pursuant thereto, this Declaration, the By-Laws, Articles, and use restrictions and rules of the Association shall prevail. The foregoing priorities shall apply, but not be limited to, the lien for assessments created in favor of the Association. Nothing in this Section 14.8 shall preclude any Supplemental Declaration or other recorded declaration, covenants and restrictions applicable to any portion of the Properties from containing additional restrictions or provisions which are more restrictive than the provisions of this Declaration, and the Association shall have the standing and authority to enforce the same.
- 15.9 <u>Use of Protected Words "The Settings of Black Mountain"</u>. No Person shall use the words "The Settings of Black Mountain" or any derivative in any printed or promotional material without the Declarant's prior written consent. However, Owners may use the words "The Settings of Black Mountain" in printed or promotional matter where such terms are used solely to specify that particular property is located within The Settings of Black Mountain and the Association shall be entitled to use the word "The Settings of Black Mountain" in its name.
- 15.10 <u>Compliance</u>. Every Owner and occupant of any Lot shall comply with the Governing Documents. Failure to comply shall be grounds for an action by the Association or by any aggrieved Owner(s) to recover sums due, for damages or injunctive relief, or for any

other remedy available at law or in equity, in addition to those enforcement powers granted to the Association in Section 4.7.

- 15.11 Notice of Sale or Transfer of Title. Any Owner desiring to sell or otherwise transfer title to a Lot shall give the Board at least seven (7) Days' prior written notice of the name and address of the purchaser or transferee, the date of such transfer of title. The transferor shall continue to be jointly and severally responsible with the transferee and for all obligations of the Owner of the Lot, including assessment obligations, until the date upon which such notice is received by the Board, notwithstanding the transfer of title.
- 15.12 <u>Variances and Waiver of Restrictions</u>. So long as permitted by North Carolina law and so long as Declarant owns any Lot or interest in property subjected to this Declaration, Declarant may waive or otherwise allow and authorize variances from the terms and restrictions hereof or the terms and restrictions of any Supplemental Declaration as might hereafter be relevant to the Property or any part thereof.
- 15.13 Merger and Subdivision of Lots. Upon application in writing by an Owner of adjoining Lots, the Declarant and, upon assignment of such right, the Board of Directors, may authorize the merger of adjoining Lots or the subdivision of a Lot, subject to the consent of such Mortgagees as may have an interest in the affected Lot. Such merger or subdivision shall be in conformance within the provisions of the Governing Documents and any Supplemental Declaration that may be applicable to such Lots, including provisions which may further regulate merger or subdivision and use provisions regulating use of Lots. Such plats and plans as may be necessary to show the merged or subdivided Lots shall be thereafter prepared at the expense of the requesting Owner, who shall additionally be responsible for all costs, including legal fees, associated with the merger or subdivision of such Lots. The Declarant (or Board of Directors, as the case may be) may impose conditions for use of the merged or subdivided Lot as a condition precedent to granting approval for such a merger or subdivision. From and after the time a merger or subdivision of Lots is approved, such resulting Lots shall, for all purposes, be considered Lots in accordance with their new boundaries.
- 15.14 Reservation From Lot Conveyance. It is expressly agreed and understood that the title conveyed by Declarant to any Lot, Common Area or other parcel of land within the Properties by contract, deed, or other conveyance shall be subject to any easement affecting same for roadways or drainage, access, ingress, egress, water, gas, sewer, storm sewer, electric light, electric power, telegraph, telephone, or cable television purposes and shall convey no interest in any pipes, lines, poles, or conduits, or in any utility facility or appurtenances thereto, constructed by or under authority of Declarant or any easement owner, or their agents through, along, or upon the premises affected thereby to serve said land or any other portion of the Properties, and where not affected, the right to maintain, repair, sell, or lease such appurtenances to any municipality or other governmental agency or to any public service corporation or to any other party is hereby expressly reserved.
- 15.15 <u>Incorporation by Reference</u>. All dedications, limitations, restrictions, and reservations shown on any recorded Subdivision Plat of the Community as recorded by the Declarant or any recorded plat of a Phase as recorded by Declarant are incorporated herein and made a part hereof as if fully set forth herein and shall be construed as being adopted in each and

every contract, deed, or conveyance executed or to be executed by or on behalf of Declarant and, thereafter, each successive Owner, conveying any of the Properties, whether specifically referred to therein or not.

- 15.16 <u>Interpretation</u>. In all cases, the provisions set forth and provided for in this Declaration shall be construed together and given that interpretation or construction which, in the opinion of the Class B member, if such membership exists, or directors of the Association if the Class B member is terminated, will best affect the intent of the general plan of the Properties. The provisions hereof shall be liberally interpreted and, if necessary, they shall be so extended or enlarged by implication as to make them fully effective. The effective date of this Declaration shall be the date of its filing for record on the records of the Superior Court of Buncombe County, North Carolina.
- 15.17 <u>Captions</u>. The captions of each article and section hereof as to the contents of each article and section are inserted only for convenience and are in no way to be constructed as defining, limiting, extending or otherwise modifying or adding to the particular article or section to which they refer.
- 15.18 <u>Gender and Grammar</u>. The singular wherever used herein shall be construed to mean the plural when applicable, and the use of the masculine pronoun shall include the neuter and feminine.

ARTICLE 16: CONDEMNATION

Whenever all or any part of the Common Area shall be taken (or conveyed in lieu of and under threat of condemnation) by any authority having the power of condemnation or eminent domain, the award made for such taking shall be payable to the Association, as Trustee for all Owners, to be disbursed as follows:

If the taking involves a portion of the Common Area on which improvements have been constructed, then, unless within sixty (60) days after such taking the Declarant, for so long as Declarant owns a Lot for the primary purpose of sale or has the unexpired option to add the Additional Property or portion or portions thereof to the Properties, and at least seventy-five (75%) percent of the Class "A" members of the Association shall otherwise agree, the Association shall restore or replace such improvements so taken on the remaining land included in the Common Area, to the extent lands are available therefore, in accordance with plans approved by the Board of Directors of the Association. If such improvements are to be repaired or restored, the above provisions in Article 6 hereof regarding the disbursement of funds in respect to casualty damage or destruction which is to be repaired shall apply. If the taking does not involve any improvements on the Common Area, or if there is a decision made not to repair or restore, or if there are net funds remaining after any such restoration or replacement is completed, then such award of funds or remaining net funds shall be disbursed to the Association and used for such purposes as the Board of Directors of the Association shall determine.

If the taking or conveyance in lieu thereof includes all or any portion of a Lot also includes any part of the Common Area, then a Court of competent jurisdiction shall apportion such

award or proceeds and such award or proceeds shall be disbursed to the Association and the Owner so affected so as to give just compensation to the Owners of any Lot; providing, however, such apportionment may instead be resolved by the agreement of (i) the Board of Directors, (ii) the Owners of all Lots, wholly or partially taken or sold, together with the Mortgagees for each such Lot, and (iii) Declarant, for so long as Declarant owns a Lot for the primary purpose of sale or has the unexpired option to add the Additional Property or any portion or portions thereof to the Development.

In the event that all or any part of a Lot is taken (or conveyed in lieu of and under threat of condemnation) by any authority having the power of condemnation or eminent domain, and in further event that the Owner of such Lot responsible for the maintenance and repair of such Lot as the case may be, elects not to restore the remainder then such Owner making such election shall promptly clear away any remaining improvements damaged or destroyed by such taking or conveyance and shall leave such Lot and any remaining undamaged improvements thereon in a clean, orderly, safe, and sightly condition. In addition, if the size or configuration of such Lot remaining after such taking or conveyance is insufficient to permit the restoration of the remaining improvements thereon or therein to their condition prior to such taking or conveyance in compliance with all applicable standards, restrictions and provisions of this Declaration and all applicable zoning, subdivision, building, and other governmental regulations, then such Owner shall have the option, after clearing away all remaining improvements or portions thereof and placing the remainder in a clean, orderly, safe, and sightly condition referred to above, of deeding their remaining portion of the Lot as part of the Common Area, and thereafter any such Owner shall not have any further voting rights or membership rights or privileges in the Association or with respect to the Properties and shall not be subject to any further assessments imposed by the Association and payable after the date of such conveyance. Further, in the event that any part of a Lot is taken (or in lieu of and under threat of condemnation) by any authority having the power of condemnation or eminent domain, and if the Owner of such Lot responsibility for the maintenance and repair of such, as the case may be, elects to restore the remainder of the Lot, such Owner making such election shall restore such remainder of such Lot as nearly as practicable to the same condition it was in prior to such taking or conveyance and in accordance with all applicable standards, restrictions, and provisions of this Declaration, including Article 9 hereof, and all applicable zoning, subdivision, building and other governmental regulations. All such work or restoration shall be commenced promptly following such taking or conveyance and shall be carried through diligently to conclusion within a reasonable time.

IN WITNESS WHEREOF, the undersigned Declarant has executed this Declaration this day of April, 2005.

DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC,

a Georgia limited liability company

Riehard McWhorter, Manager

STATE OF GEORGIA NORTH CAROLINA COUNTY OF BUNCOMBE

I, Susan Stray Horn Barbour, a Notary Public of Burbour Burbour, a Notary Public of Burbour Burbour Burbour Burbour And Barbour And Barbour Burbour Bu

WITNESS my hand and notarial seal at office, this Bunday of April, 2005.

Susan Strayhorn Barbour, Notary Public

SUSAN STRAYHORN BARBOUR

Print Name here

NOTARY SEAL:

My Commission expires: 12-10-06

EXHIBIT "A"

THE SETTINGS OF BLACK MOUNTAIN

BEING all of that piece, phase or tract of land lying and being in Buncombe County, NC, being shown and described on a plat entitled "Phase 1- The Settings of Black Mountain", which survey is dated April 15, 2005 and was prepared by Eric S. McAbee, PLS, which plat is recorded in the office of the Register of Deeds for Buncombe County, North Carolina in Plat Book 96, at Pages 162 through 164 (hereinafter the "plat"). For a more detailed description of courses, distances, metes and bounds of lots, easements and common areas shown within said Phase One, reference is had to the above referenced plat which is incorporated herein by reference.

TOGETHER WITH AND SUBJECT TO all easements, restrictions and rights of ways of record and as specifically shown on the above referenced recorded plat.

BEING a portion of that property conveyed to The Settings of Black Mountain, LLC, A Georgia limited liability company by deed recorded in Record Book 3947, at Page 557 of the Buncombe County, North Carolina Registry.

EXHIBIT "B"

CONSENT OF MORTGAGEE

THE SETTINGS OF BLACK MOUNTAIN

Branch Banking and Trust Company, the Beneficiary under that certain Deed of Trust from The Settings of Black Mountain, LLC to BB&T Collateral Service Corporation, Trustee, recorded in Deed of Trust Book 3947, at Page 573 of the Buncombe County, NC Public Registry ("Deed of Trust"), conveying the property described therein, hereby: (a) consents to the recordation of this Declaration and the imposition of the provisions of the North Carolina Planned Community Act to the real property described in the Deed of Trust (the "Property"); and (b) subordinates the lien and operation of the Deed of Trust to this Declaration and the provisions contained herein, except that the Deed of Trust is not subordinated to any liens or assessments created by or arising from the Declaration. In the event of foreclosure of the Deed of Trust, or the transfer of any portion of the Property in lieu of foreclosure, Beneficiary and Trustee agree that the purchaser at any such foreclosure or the transferee under any such deed in lieu of foreclosure shall take title to the Property together with and subject to all of the terms of this Declaration. The execution of this Consent of Mortgagee by the Beneficiary shall not be deemed or construed to have the effect of creating between the Beneficiary and Declarant the relationship of partnership or of joint venture, nor shall it be deemed to impose upon the Beneficiary any of the liabilities, duties or obligations of the Declarant under these Restrictions nor does Beneficiary consent to the subordination of the Deed of Trust to any liens or assessments created by or arising from the Declaration. Beneficiary executes this Consent of Mortgagee solely for the purposes set forth above. The Trustee also joins in and executes this Consent of Mortgagee at the request of Beneficiary, and for the purposes set forth above, and in the ordinary course of business.

TRUSTEE:	BENEFICIARY:
BB&I Collateral Service Corporation,	Branch Banking and Trust Company,
a North Carolin Fanking Corporation	A North Carolina Banking Corporation
By: Rresident	By: Calcol S. President

STATE OF NORTH CAROLINA	
COUNTY OF Runconte	
I, A Notary Public of said State and County, do hereby Lauid Z. Hauss, personally appeared by that is <u>Vice</u> , President of BB&T Collateral Service Banking Corporation, Trustee, and that by authority duly the foregoing instrument was signed in its name by its <u>Vi</u>	efore me this day and acknowledged e Corporation, A North Carolina given and as the act of the corporation,
Witness my hand and official stamp or seal, this 13	
Notar()Public	(Official Seal)
My Commission Expires: 11-6-2005	TO BUILDING
STATE OF NORTH CAROLINA COUNTY OF Bunconte	

I, A Notary Public of said State and County, do hereby certify that

Michael 5. Bohan, personally appeared before me this day and acknowledged that

is <u>Ure</u>, President of Branch Banking and Trust Company, a North Carolina Banking

Corporation, Beneficiary, and that by authority duly given and as the act of the corporation, the

foregoing instrument was signed in its name by its <u>Ure</u> President.

Witness my hand and official stamp or seal, this 13 day of April, 2005

Notate Public

(Official Seal)

My Commission Expires: 11-6-2005

H

Du

Doc ID: 018091710004 Type: CRP
Recorded: 12/15/2006 at 02:58:10 PM
Fee Amt: \$23.00 Page 1 of 4
Workflow# 2084377
Buncombe County. Nc
Otto W. DeBruhl Redister of Deeds
BK 4333 PG 317-320

Return after recording to: Prepared by: Susan S. Barbour, Esq. Box 31 McGuire, Wood & Bissette, PA 48 Patton Avenue

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this 12 day of December, 2006 by and between Richard W. McWhorter and wife, Sandra Dugas McWhorter ("Owner") and The Settings of Black Mountain, LLC, A Georgia Limited Liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant is the developer of The Settings of Black Mountain, a subdivision located in Black Mountain, Buncombe County, North Carolina; and

WHEREAS, Owner is the owner of Lot 158 of The Settings of Black Mountain as more particularly shown and described by deed recorded in Deed Book 4151, at Page 374 and as shown on amended plat recorded in Plat Book 123, at Page 76 of the Buncombe County, NC Registry (hereinafter "Lot 158"); and

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration");

WHEREAS, Lot 158 was previously submitted to the provisions of the Declaration to become a part of the Community of The Settings of Black Mountain by the filing of deed to Owner recorded in Deed Book 4151, at Page 374 of the Buncombe County, NC Registry; and

. . . .

WHEREAS, pursuant to Article 15.2 of the Declaration, Declarant has the unilateral option, right and privilege to amend the Declaration for any purpose by filing in the public records, an amendment or Supplemental Declaration; and

WHEREAS, the Class "B" membership as set forth in Article 3.3 (b) of the Declaration has not been terminated; and

WHEREAS, any such amendment shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein; and

WHEREAS, Declarant wishes to exercise its unilateral option, right and privilege to amend the Declaration; and

WHEREAS, pursuant to Article 10.12 of the Declaration no Lot may be subdivided or its boundary lines changed after a subdivision plat depicting the Lot has been approved and filed in the Public Records without Declarant's prior written consent; and

WHEREAS. Owner has requested that Lot 158, containing 10 acres more or less, be approved to be subdivided to create no more than two Lots and Declarant has agreed to officially approve said request.

NOW THEREFORE, in consideration of \$10.00 and other valuable considerations the receipt and sufficiency of which is hereby acknowledged, the Declarant hereby amends the Declaration in the following:

- 1. Declarant hereby declares that **as to Lot 158 only**, Article 10.12 of the Declaration is hereby amended in its entirety as follows: Lot 158 may be further subdivided to create no more than two Lots. No subdivision plat of Lot 158 may be finalized and recorded without the Declarant's prior written approval which approval shall not be unreasonably withheld. Any subdivision and subsequent conveyance of a portion of Lot 158 must be pursuant to reference to an approved and recorded subdivision plat. From and after the filing of this Amendment in the Buncombe County, North Carolina Records, the term "Lot" as used therein shall refer to and include the subdivided portions of Lot 158 upon the recording of a subdivision plat as set forth herein.
- 2. Except as specifically modified hereinabove, the Declaration shall remain in full force and effect and shall apply to Lot 158 or any subsequent subdivision thereof. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding upon and inure to the benefit of the Owner of Lot 158 and any subsequent owner of any subdivided portion of Lot 158 which subdivided portion shall be considered a "Lot" pursuant to the terms of the Declaration and to all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant and Owner have caused this instrument to be signed and sealed as of the date and year first above written.

DECLARANT:
THE SETTINGS OF BLACK MOUNTAIN, LLC
a Georgia limited liability company
By:
Richard W. McWhorter, Manager
OWNER:
(SEAL)
Richard W. McWhorter
The Dung I
nunara Myas W (SEAL)
Sandra Dugas McWhorter
STATE OF GEORGIA
COUNTY OF Cobb
I, a Notary Public, of the aforesaid state and county, certify that RICHARD W.
McWHORTER personally appeared before me this day and being personally known to me or
having produced satisfactory evidence of his identity in the form of a drivers license
acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC,, A
Georgia limited liability company and further acknowledges the voluntary and due execution of
the foregoing instrument on behalf of and as the official act of the limited liability company.
the foregoing modulient on benefit of and as the official act of the inflited flatinty company.
WITNESS my hand and notary seal at office, this /2 day of December, 2006.

____ My Commission expires:_

(NOTARY SEAL)

STATE OF GEO	ORGIA	
COUNTY OF	Lubb	

I, a Notary Public, of the aforesaid state and county, certify that RICHARD W. McWHORTER and SANDRA DUGAS McWHORTER personally appeared before me this day and being personally known by me or having produced satisfactory evidence of his and her identity in the form of drivers licenses acknowledged the voluntary and due execution of the foregoing instrument for the purposes intended therein.

WITNESS my han	d and notary seal	at office, this	12 day	of December,	2006.
----------------	-------------------	-----------------	--------	--------------	-------

My Commission expires: 4/5/08

Notary Public

Chris Holland

Print Name

(NOTARY SEAL)

1

Doc ID: 017239139095 Type: CRP Recorded: 03/17/2006 at 12:00:12 PM Fee Amt: \$26.00 Page 1 of 5 Workflow# 2012299 Buncombe County, NC Otto W. DeBruhl Register of Deeds Bk 4193 Pc1908-1912

Return after recording to: Prepared by: Susan S. Barbour, Esq. Box 31 McGuire, Wood & Bissette, PA 48 Patton Avenue

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this <u>Ib</u> day of March, 2006 by Richmarc Black Mountain, LLC, A Georgia Limited Liability company ("Owner") and approved by The Settings of Black Mountain, LLC, A Georgia Limited Liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant is the developer of The Settings of Black Mountain, a subdivision located in Black Mountain, Buncombe County, North Carolina; and

WHEREAS, Owner is the owner of a tract located adjacent to The Settings of Black Mountain as more particularly shown and described by plat recorded in Plat Book 102, at Page 62 of the Buncombe County, NC Registry (hereinafter "Phase 2A Property"); and

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration");

WHEREAS, Owner desires to submit the Phase 2A Property to the provisions of the Declaration to become a part of the Community of The Settings of Black Mountain; and

WHEREAS, pursuant to Article 2 and Article 7 of the Declaration, Declarant has the unilateral option, right and privilege to subject to the provisions of the Declaration and the

jurisdiction of the Association, all or any portion of the Additional Property, by filing in the public records, an amendment or Supplemental Declaration annexing all or any portion of such property;

WHEREAS, pursuant to Article 7 of the Declaration, such amendment to the Declaration or Supplemental Declaration shall not require the vote of the Members; and

WHEREAS, any such annexation shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein; and

WHEREAS, such amendment or Supplemental Declaration shall specify such specific use restrictions or other covenants, conditions and restrictions to be applicable to the annexed property as Declarant may, in its own discretion, determine; and

WHEREAS, Declarant wishes to exercise its option to submit to the provisions of the Declaration and the jurisdiction of the Association the Phase 2A Property more particularly described on Exhibit "A" attached hereto and by this reference made a part hereof ("Submitted Property"), and to thereby cause this Submitted Property to become part of the Community known as The Settings of Black Mountain; and

WHEREAS, the Submitted Property also known as Phase 2A and as shown on Plat recorded in Plat Book 102, at Page 63 of the Buncombe County, NC Registry, is to be made subject to the Declaration.

NOW THEREFORE, in consideration of the premises and of the benefits to be derived by the Declarant and accruing to the Property and Owners within The Settings of Black Mountain, the Declarant hereby amends the Declaration in the following respects for the purpose of submitting the Submitted Property to the Declaration and jurisdiction of the Association as follows:

- 1. Declarant hereby declares that the Submitted Property is hereby subjected to all terms, provisions, covenants, restrictions, easements and conditions of the Declaration. The Submitted Property shall, from and after the date of recordation of this Amendment, be owned, held, transferred, sold, conveyed, used, occupied and mortgaged or otherwise encumbered subject to the provisions of the Declaration, and every grantee of every interest in the Submitted Property, by acceptance of a deed or other conveyance of such interest, whether or not such deed or other conveyance is expressly made subject to the provisions of the Declaration, and whether or not such grantee shall consent thereto in writing, shall take such interest subject to the provisions of the Declaration and shall be deemed to have consented to the same. From and after the filing of this Amendment in the Buncombe County, North Carolina Records, the "Community", as such term is defined by and described in the Declaration, shall include the Submitted Property, and the term "Lot" as used therein shall refer to and include the Lots located on the Submitted Property.
- 2. Except as modified hereinabove, the Declaration shall remain in full force and effect and shall, from and after the effective date hereof, apply to the Submitted Property. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding

upon and inure to the benefit of all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant has caused this instrument to be signed and scaled as of the date and year first above written.

SUBMITTED BY:

OWNER:

RICHMARC BLACK MOUNTAIN, LLC,

a Georgia limited liability/company

By:

Richard W. McWhorter, Manager

APPROVED AND ACCEPTED BY:

DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC, a Georgia limited hiability company

By: Richard W. McWhorter, Manager

STATE OF NORTH CAROLINA COUNTY OF BUNCOMBE

I, a Notary Public, of Buncombe County, North Carolina, certify that RICHARD W. McWHORTER personally appeared before me this day and acknowledged that he is Manager of RICHMARC BLACK MOUNTAIN, LLC., A Georgia limited liability company and further acknowledges the due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.

WITNESS my hand and notary seal at office, this 16 day of March, 2006.

Juan Shayhour Barban My Commission expires: 12-10-06

Notary Public SUSAN STRAY HORN & MRBOUR

Print Name

NOTARY SEAL)

STATE OF NORTH CAROLINA COUNTY OF BUNCOMBE

I, a Notary Public, of Buncombe County, North Carolina, certify that RICHARD W. McWHORTER personally appeared before me this day and acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC,, A Georgia limited liability company and further acknowledges the due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.

WITNESS my hand and notary seal at office, this 16 day of March, 2006.

Alpan Straighan Balos My Commission expires: 12-10-06
Notary Public

Olary Public SUSAW STRAY HORN BAKBOUR

Print Name

(NOTARY SEAL)

EXHIBIT "A"

SUBMITTED PROPERTY

INCLUDING AND BEING all of that property conveyed to Richmarc Black Mountain, LLC, A Georgia limited liability company as shown and described by deeds recorded July 12, 2005 in Deed Book 4072, at Page 1711 and Deed Book 4072, at Page 1714 of the Buncombe County, NC Registry AND ALSO INCLUDING ANY SUBSEQUENTLY CONVEYED PROPERTY to Richmarc Black Mountain, LLC, A Georgia limited liability company.

The above referenced property shall be subdivided and shall be known as Phase 2A of The Settings of Black Mountain Subdivision and this limited power of attorney shall be considered in full force and effect for Phase 2A or any subsequent phases of this property which shall be known as The Settings of Black Mountain Subdivision as the same may be shown on subsequently recorded plats and subdivision documentation.



Doc ID: 020351980004 Type: CRP Recorded: 09/14/2007 at 11:26:35 AM Fee Amt: \$23.00 Page 1 of 4 Workflow# 2397257 Buncombe County, NC Otto W. DeBruhl Register of Deeds

BK 4463 PG 412-415

Return after recording to: Prepared by: Susan S. Barbour, Esq. Box 31 McGuire, Wood & Bissette, PA 48 Patton Avenue

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this 11th day of September, 2007 by The Settings of Black Mountain, LLC, A Georgia Limited Liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant is the developer of The Settings of Black Mountain, a subdivision located in Black Mountain, Buncombe County, North Carolina; and

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration");

WHEREAS, Owner desires to submit the Phase 3A Property to the provisions of the Declaration to become a part of the Community of The Settings of Black Mountain; and

WHEREAS, pursuant to Article 2 and 7 of the Declaration, Declarant has the unilateral option, right and privilege to subject to the provisions of the Declaration and the jurisdiction of the Association, all or any portion of the Additional Property, by filing in the public records, an amendment or Supplemental Declaration annexing all or any portion of such property;

WHEREAS, pursuant to Article 7 of the Declaration, such amendment to the Declaration or

Supplemental Declaration shall not require the vote of the Members; and

WHEREAS, any such annexation shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein; and

WHEREAS, such amendment or Supplemental Declaration shall specify such specific use restrictions or other covenants, conditions and restrictions to be applicable to the annexed property as Declarant may, in its own discretion, determine; and

WHEREAS, Declarant wishes to exercise its option to submit to the provisions of the Declaration and the jurisdiction of the Association the Phase 3A Property more particularly described on Exhibit "A" attached hereto and by this reference made a part hereof ("Submitted Property"), and to thereby cause this Submitted Property to become part of the Community known as The Settings of Black Mountain; and

NOW THEREFORE, in consideration of the premises and of the benefits to be derived by the Declarant and accruing to the Property and Owners within The Settings of Black Mountain, the Declarant hereby amends the Declaration in the following respects for the purpose of submitting the Submitted Property to the Declaration and jurisdiction of the Association as follows:

- 1. Declarant hereby declares that the Submitted Property is hereby subjected to all terms, provisions, covenants, restrictions, easements and conditions of the Declaration. The Submitted Property shall, from and after the date of recordation of this Amendment, be owned, held, transferred, sold, conveyed, used, occupied and mortgaged or otherwise encumbered subject to the provisions of the Declaration, and every grantee of every interest in the Submitted Property, by acceptance of a deed or other conveyance of such interest, whether or not such deed or other conveyance is expressly made subject to the provisions of the Declaration, and whether or not such grantee shall consent thereto in writing, shall take such interest subject to the provisions of the Declaration and shall be deemed to have consented to the same. From and after the filing of this Amendment in the Buncombe County, North Carolina Records, the "Community", as such term is defined by and described in the Declaration, shall include the Submitted Property, and the term "Lot" as used therein shall refer to and include the Lots located on the Submitted Property.
- 2. Except as modified hereinabove, the Declaration shall remain in full force and effect and shall, from and after the effective date hereof, apply to the Submitted Property. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding upon and inure to the benefit of all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant has caused this instrument to be signed and sealed as of the date and year first above written.

SUBMITTED BY DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC,

a Georgia limited liability company

By:

Richard W. McWhorter, Manager

STATE OF GEORGIA COUNTY OF Forseth

I, a Notary Public, of Forset County, State of GEORGIA, certify that RICHARD W. McWHORTER personally appeared before me this day and being personally known by me and who acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC,, A Georgia limited liability company and further acknowledges the due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.

WITNESS my hand and notary seal at office, this // day of September, 2007.

Notary Public

DEDAT IC. VE

Print Name

(NOTARY SEAL)

My Commission expires: 6-5-2010

4

EXHIBIT "A"

SUBMITTED PROPERTY

INCLUDING AND BEING all of that property shown as Phase 3A of The Settings of Black Mountain and being more particularly shown and described on plats recorded in Plat Book 114, at Page 3 and Plat Book 114, at Page 4 the Buncombe County, NC Registry, reference to said plats being made for a more particular description of said Phase 3A of The Settings of Black Mountain.

TOGETHER WITH AND SUBJECT TO all easements, restrictions and rights of ways of record.

ND: 4835-6058-0097, v. 1

Doc ID: 020195890003 Type: CRP Recorded: 07/25/2007 at 03:03:44 PM Fee Amt: \$20.00 Page 1 of 3 Workflow# 2385683 Buncombe County. NC Otto W. DeBruhl Register of Deeds

BK 4440 PG 417-419

Return after recording to: Prepared by: Susan S. Barbour, Esq. Box 31 McGuire, Wood & Bissette, PA 48 Patton Avenue

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS **FOR** THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this 23rd day of February, 2007 by The Settings of Black Mountain, LLC, A Georgia Limited Liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 recorded in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration"); and

WHEREAS, pursuant to request by the North Carolina Division of Water Quality (hereinafter "DWQ") lots within The Settings of Black Mountain which are adjacent to or include a stream or wetland area are to be specifically identified and all lot owners are to be provided with specific notification of wetland and buffer regulations; and

WHEREAS, pursuant to Article 15.2 of the Declaration, Declarant has the unilateral option, right and privilege to amend the Declaration for any purpose by filing in the public records, an amendment or Supplemental Declaration; and

WHEREAS, the Class "B" membership as set forth in Article 3.3 (b) of the Declaration has not been terminated; and

WHEREAS, any such amendment shall be effective upon the filing for record of any such

amendment or Supplemental Declaration, unless otherwise provided therein; and

WHEREAS, Declarant wishes to exercise its unilateral option, right and privilege to amend the Declaration; and

NOW THEREFORE, in consideration of \$10.00 and other valuable considerations the receipt and sufficiency of which is hereby acknowledged, the Declarant hereby amends the Declaration by amending Section 10.16 as follows:

- (a) The following lots within Phases One and Two of the Settings of Black Mountain as shown on the recorded plats thereof in the Buncombe County, NC Registry are adjacent to or include a stream or wetland: Lots 1, 5, 6, 7, 14, 15, 16, 17, 18, 25, 26, 27, 28, 35, 36, 37, 39, 45, 46, 47, 48, 49, 50, 95, 96, 97, 107, 110, 117, 118, 119, 120, 127, 134, 136, 137, 139, 145, 146, 147, 148, 149, 150, 151, 152, 153, 154, 171, 172, 173, 174, 177, 210, 242, 243, 244, 249, and 252. Declarant is committed to insuring that all streams and wetlands within the Settings of Black Mountain are protected and generally left in their natural state and the above designated lots in Phases One and Two are subject to a stream buffer ten feet (10') in width from the top of the stream bank (bankfull) to be delineated by a surveyor or, if no top of bank can be delineated by a surveyor, then in the alternative, the buffer shall be measured from the edge of the stream (ordinary high water) and shall be twenty five feet (25') in width.
- (b) All common areas as located and shown on all plats for Phases One and Two of The Settings of Black Mountain shall be subject to a twenty five (25) foot wide stream buffer as measured from the edge of the stream.
- (c) Excepting those lots as noted in subsection (d) below, all lots and common areas located and shown on all plats for Phase Three (3) of The Setting of Black Mountain shall be subject to a twenty five (25) foot wide stream buffer as measured from the edge of the stream.
- (d) Lots 86, 89, 90, 94, 120 and 122 of Phase Three (3) of The Settings of Black Mountain shall be subject to a stream buffer ten feet (10') in width as measured from the edge of stream and only limited clearing for the placement of the residential home site shall be allowed within that portion of the lot located outside of the ten foot buffer but within twenty five feet (25') of the edge of stream.
- 2. Except as specifically modified hereinabove, the Declaration shall remain in full force and effect. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding upon and inure to the benefit of all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant has caused this instrument to be signed and sealed as of the date and year first above written.

DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC

a Georgia limited liability company

Richard W. McWhorter, Manager

STATE OF GEORGIA
COUNTY OF (DIVIDING)

I, a Notary Public, of the aforesaid state and county, certify that RICHARD W. McWHORTER personally appeared before me this day and being personally known to me or having produced satisfactory evidence of his identity in the form of a drivers license acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC,, A Georgia limited liability company and further acknowledges the voluntary and due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.

WITNESS my hand and notary seal at office, this 25 day of February, 2007.

Notary Public

Print Name

(NOTARY SEALM)

My Commission expires: 4.21.2

ND: 4828-0432-7937, v. 1



Doc ID: 022927190003 Type: CRP Recorded: 07/08/2010 at 11:04:15 AM Fee Amt: \$20.00 Page 1 of 3 Workflow# 0000033636-0001

Buncombe County, NC Otto W. DeBruhl Register of Deeds

вк 4798 рс 11-13

Return after recording to: Worley& Scott, A. Box 99 Prepared by: Susan S. Barbour, Esq. Box 31 McGuire, Wood & Bissette, PA 48 Patton Avenue

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this day of July 2010 by The Settings of Black Mountain, LLC, A Georgia Limited Liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant is the developer of The Settings of Black Mountain, a subdivision located in Black Mountain, Buncombe County, North Carolina; and

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration");

WHEREAS, pursuant to Article 15.2 of the Declaration, Declarant has the unilateral option, right and privilege to amend the Declaration for any purpose by filing in the public records, an amendment or Supplemental Declaration; and

WHEREAS, the Class "B" membership as set forth in Article 3.3 (b) of the Declaration has not been terminated; and

WHEREAS, any such amendment shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein; and

2

WHEREAS, Declarant wishes to exercise its unilateral option, right and privilege to amend the Declaration; and

WHEREAS, pursuant to Article 2.7 of the Declaration, Declarant has revised and amended that Phase 1, Subdivision Plat recorded in Plat Book 96, at Page 163 (hereinafter "Original Phase 1 Plat") and that Phase 3, Subdivision Plat recorded in Plat Book 108, at page 171 (hereinafter "Original Phase 3 Plat") by revised plat recorded in Plat Book 151, at Page 96 (hereinafter "Revised Lot 127 Plat") of the Buncombe County, NC Registry for the express purpose of combining Lot 127 and Common Area # 23 as shown on the Original Phase 1 Plat and Original Phase 3 Plat and creating one lot to be known as Revised Lot 127, containing 0.95 acres, as shown on the Revised Lot 127 Plat (hereinafter "Revised Lot 127"); and

NOW THEREFORE, in consideration of \$10.00 and other valuable considerations the receipt and sufficiency of which is hereby acknowledged, the Declarant hereby amends the Declaration as follows:

- 1. Revised Lot 127 is one Lot pursuant to the terms of the Declaration effective as of June 24, 2010 being the date of recording of that plat recorded in Plat Book 151, at Page 96.
 - 2. Revised Lot 127 shall only be subject to a single lot assessment.
- 3. Except as modified hereinabove, the Declaration shall remain in full force and effect and shall, from and after the effective date hereof, apply to the Submitted Property. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding upon and inure to the benefit of all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant has caused this instrument to be signed and sealed as of the date and year first above written.

SUBMITTED BY DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC,

a Georgia limited liability company

Rw

Richard W. McWhorter, Manager

STATE OF GEORGIA
COUNTY OF (16b)
I, a Notary Public, of County, State of GEORGIA, certify that RICHARD W. McWHORTER personally appeared before me this day and being personally known by me and who acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC,, A Georgia limited liability company and further acknowledges the due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.
, WITNESS my hand and notary seal at office, this 6^{-4} day of July 2010,
My Commission expires: 8/13/2013
Notary Public
Vaushy H. Mrgar
Print Name (NOTARY SEAL)
VAUGHN ALEXANDER MORGAN
NOTARY PILIC
FULTON COUNTY, GEORGIA
WIT COMM. EXPIRES
8/13/2013

Prepared By and Return to: Craig Justice, Esq. Van Winkle Law Firm 11 N. Market Street Asheville, NC 28801

SUPPLEMENT TO THE DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR THE SETTINGS OF BLACK MOUNTAIN

This Supplement to the Declaration of Covenants, Conditions and Restrictions for The Settings of Black Mountain is made this 28th day of February 2011 by The Settings of Black Mountain, LLC, a Georgia limited liability company ("Declarant").

WITNESSETH:

WHEREAS, Declarant is the developer of The Settings of Black Mountain, a subdivision located in Black Mountain, Buncombe County, North Carolina; and

WHEREAS, Declarant, as the developer of The Settings of Black Mountain, recorded that certain Declaration of Covenants, Conditions, and Restrictions for The Settings of Black Mountain, on April 22, 2005 in Deed Book 3993, Page 34, of the Buncombe County, NC Registry (as may be amended from time to time, the "Declaration");

WHEREAS, pursuant to Article 15.2 of the Declaration, Declarant has the unilateral option, right and privilege to amend the Declaration for any purpose by filing in the public records, an amendment or Supplemental Declaration; and

WHEREAS, the Class "B" membership as set forth in Article 3.3 (b) of the Declaration has not been terminated; and

WHEREAS, any such amendment shall be effective upon the filing for record of any such amendment or Supplemental Declaration, unless otherwise provided therein; and

WHEREAS, Declarant wishes to exercise its unilateral option, right and privilege to amend the Declaration; and

NOW THEREFORE, in consideration of \$10.00 and other valuable considerations the receipt and sufficiency of which is hereby acknowledged, the Declarant hereby amends the Declaration as follows:

- 1. Section 4.4 of the Declaration is hereby deleted in its entirety and the following new Section 4.4 is substituted in lieu thereof:
 - 4.4 <u>Personal Property and Real Property for Common Use</u>. The Association through the action of its Board may acquire, hold, encumber and dispose of tangible and intangible personal property and improved or unimproved real estate and leasehold and other interests in personal property and real estate. The general authority in the foregoing sentence includes, without limitation, the following:
 - (a) The Declarant and its designees, with the Declarant's prior written consent, may convey to the Association improved or unimproved real estate, or interests in real estate, located within the property described in Exhibit A, personal property and leasehold and other property interests. Such property shall be accepted by the Association and thereafter shall be maintained by the Association at its expense for the benefit of its Members. Declarant shall not be required to make any improvements or repairs whatsoever to property to be conveyed and accepted pursuant to this Section 4.4.
 - (b) The Association may purchase and/or receive an assignment of any note or notes or interest in any note or notes pertaining to the Community and may foreclose same and may otherwise purchase or acquire any Lots or interests in any Lots within Community or other real estate for the purposes that include the financing of enhancements to the Community. The Board of the Association may elect to levy a special assessment binding on all Lot owners related to acquisition of any such note or notes or interests therein and/or purchase or other acquisition of any Lot or Lots or interests therein or other real property consistent with the procedure in Section 8.4 hereof.
- 2. Section 4.7(a) of the Declaration is hereby deleted in its entirety and the following new Section 4.7(a) is substituted in lieu thereof:
 - (a) The Board may impose sanctions (including suspension of the right to vote for failure to pay timely any assessment) for violation of the Governing Documents after compliance with the notice and hearing procedures set forth in Section 4.7(g) below. In the event that any occupant, guest or invitee of a Lot violates the Governing Documents, the Board or any committee established by the Board, with the Board's approval, may sanction such occupant, guest or invitee and/or the Owner of the Lot that the violator is occupying or visiting. Sanctions may include the actions enumerated below. In every instance in which the Board may act, any committee established and approved by the Board, may act in the Board's stead.

- (i) The Board may impose reasonable monetary fines which shall constitute a lien upon the Lot of the violator. A fine not exceeding One Hundred Fifty Dollars (\$150.00) may be imposed for any one violation. Without further hearing a fine may be imposed for each day of a continuing violation commencing on the day the Board's decision is made to impose the fine until the violation is cured. Collection of fines may be enforced against an Owner as if such charges were a Common Expense owed by the Owner involved. In the event that any occupant, guest or invitee of a Lot violates the Governing Documents and a fine is imposed, the fine shall first be assessed against the occupant; provided however, if the fine is not paid by the occupant within the time period set by the Board, the Owner shall pay the fine upon notice from the Board.
- (ii) The Board may suspend an Owner's right to vote and such right to suspend may be exercised if the Owner is more than ninety (90) days delinquent in paying any assessment or other charge owed to the Association.
- 3. Section 4.7(g) of the Declaration is hereby deleted in its entirety and the following new Section 4.7(g) is substituted in lieu thereof:
 - (g) The Board shall not impose a fine, suspend the right to vote or suspend the right to use the Common Area (provided, however, a late charge or interest penalty for failure to pay any assessment pursuant to Article 8 of this Declaration shall not be considered a fine for purposes of this paragraph), unless and until the Association has sent or delivered written notice to the violator as provided in subsection (i) below. Any such fine or fines and/or suspension may be effective or commence upon the sending of such notice or such later date as may be set forth in such notice, notwithstanding the violator's right to request a hearing before the Board to challenge such fine and/or suspension under subsection (ii) below.
 - (i) <u>Notice</u>. If any provision of the Declaration or Bylaws or any rule or regulation of the Association is violated, the Board shall send the violator written notice identifying the violation and fine(s) and/or suspension being imposed and advising the violator of the right to request a hearing before the Board to contest the violation or fine(s) and/or suspension or to request reconsideration of the fine(s) and/or suspension. In the event of a continuing violation, each day the violation continues or occurs again constitutes a separate offense, and fines may be imposed on a per diem basis without further notice to the violator.
 - (ii) <u>Hearing</u>. If a written request for hearing is received from the violator within ten (10) days of the date of the violation notice provided above, then the Board shall schedule and hold in executive session a hearing affording the violator a reasonable opportunity to be heard. The minutes of the meeting shall contain a written statement of the results of the hearing. The Board may establish rules of conduct for such hearing, which may include limits on time and on the number of participants who may be present at one time. If a violator does not respond within such ten-day period, the violator will be deemed to have waived the opportunity to have such hearing and to have consented to the imposition of such fine and/or suspension.

- 4. Section 8.1(d) of the Declaration is hereby deleted in its entirety and the following new Section 8.1(d) is substituted in lieu thereof:
 - All such assessments, together with late charges, simple interest at the rate of eighteen percent (18%) per annum, costs and reasonable attorney's fees actually incurred to enforce or collect such assessments, shall be a charge on the Lot and shall, after remaining unpaid for a period of thirty (30) days or longer and upon the filing of a claim of lien in the office of the Buncombe County, North Carolina Superior Court Clerk, be a continuing lien upon the Lot against which each assessment is made regardless of conveyance thereof. The Association's lien shall be prior and superior to all other liens except the lien for real estate taxes and other governmental assessments and charges against the Lot and liens and encumbrances (specifically including, but not limited to, a mortgage or deed of trust on the Lot) recorded before the docketing of the claim of lien in the office of the Buncombe County, North Carolina Superior Court Clerk. Each such assessment, together with late fees, interest, costs, and reasonable attorney's fees actually incurred, shall also be the personal obligation of the person who was the owner of such Lot at the time the assessment came due, and his or her grantee shall be jointly and severally liable for such portion thereof as may be due and payable at the time of conveyance; provided, however, any First Mortgagee or holder of a secondary purchase money Mortgage covering the Lot, provided that neither the grantee nor any successor grantee on the secondary purchase money Mortgage is the seller of the Lot who obtains title to a Lot subject to this Declaration pursuant to the remedies provided in such Mortgage or Foreclosure of the Mortgage, will not be liable for such Lot's unpaid assessments which accrued prior to the acquisition of title to such Lot by the Mortgagee. Assessments shall be paid in such manner and on such dates as may be fixed by the Board of Directors, which may include, without limitation, (i) acceleration of any assessment for Owners who are delinquent in the payment of their assessments; (ii) scheduling of assessment payments into periodic payments and other flexible payment plans, (iii) reduction (for any period) of the simple interest rate of eighteen percent (18%); adoption of amnesty plans, and (iv) such other payment or collection mechanisms as, in the discretion of the Board, are rationally related to the achievement of the needs of the Association and the equitable treatment of the Owners. The assessments shall be paid annually in advance, unless otherwise provided by the Board. The Association shall, upon request, furnish to any Owner liable for any type of Assessment a written statement signed by the Association officer setting forth whether the Assessment has been paid. This statement shall be conclusive evidence of payment. The Association may require the advance payment of a reasonable processing fee for the issuance of such a statement.
- 5. Section 8.2 of the Declaration is hereby amended by adding the following sentence to the end of the section:

Notwithstanding anything to the contrary contained herein, in the event that Declarant's Mortgagee acquires Declarant's interest with respect to any Lot pursuant to Foreclosure of a Mortgage encumbering such Lot, then such Mortgagee shall have the same rights as Declarant with respect to the payment of assessments as set forth in this Section 8.2.

6. Except as modified hereinabove, the Declaration shall remain in full force and effect. The terms hereof shall be supplemental to the terms of the aforesaid Declaration and in the event of any inconsistency, the terms of the Declaration shall control. This Amendment shall be effective upon its filing for record in the Public Records of Buncombe County, North Carolina and shall be binding upon and inure to the benefit of all Owners of Lots in the Community and their respective heirs, executors, administrators, legal representatives, successors and assigns.

IN WITNESS WHEREOF, Declarant has caused this instrument to be signed and sealed as of the date and year first above written.

SUBMITTED BY DECLARANT:

THE SETTINGS OF BLACK MOUNTAIN, LLC, a Georgia limited liability company

By:

Richard W. McWhorter, Co-Manager

STATE OF GEORGIA COUNTY OF COBB

I, a Notary Public, of ________ County, State of GEORGIA, certify that RICHARD W. McWHORTER personally appeared before me this day and being personally known by me and who acknowledged that he is Manager of THE SETTINGS OF BLACK MOUNTAIN, LLC., A Georgia limited liability company and further acknowledges the due execution of the foregoing instrument on behalf of and as the official act of the limited liability company.

WITNESS my hand and notary seal at office, this 28th day of February, 2011.

Notary Public

Deena K. Snipes

Print Name

My Commission expire

DEC